The Liverpool Baby Breathing Study

The Cross-sectional and Longitudinal Assessment of Respiratory Symptoms and the Impact on Quality of Life in the First 22 Months of Life

A Birth Cohort Study

Thesis submitted in accordance with the requirements of the University of Liverpool for the degree of Masters of Philosophy by Joshua Ian Stead

July 2017

Errata

The following amendments have been made to this thesis in line with the recommendations made by examiners following Viva Voice examination on the 10th August 2017.

Section 1.1 Introduction to the Liverpool Baby Breathing Study- Addition of study aims as outlined by the study protocol and addition of objectives that were completed by the author. This helps the reader understand the expectations of the study as a whole and within this thesis and will help clarify if these aims were achieved later in the thesis. **(Page 18)**

Section 1.2.1 Respiratory Disease and Symptoms - Table 1 was created from three separate tables. (Page 21)

Section 1.5.2 The Liverpool Respiratory Symptom Questionnaire - Figures 1-4 were added. These figures show the Liverpool Respiratory Symptom Questionnaire. This has been moved from the appendix to make it more clear to the reader the questionnaire that has been used in the LBBS. (Pages 32-35)

Section 1.6 Birth Cohort Studies- Previously this section contained a large amount of extraneous detail of the design and findings of previous birth cohort studies and other respiratory studies. This information has been condensed into a table and two summaries (section 1.6.1 and 1.6.2) discussing recruitment and attrition in previous cohort studies. The in depth analysis of the cohort studies has been moved to the appendix. (*Pages 30-40*)

Section 2.1 Section 1. LBBS Study Design and Development Summary - Previously this section contained a detailed description of work performed by previous MPhil students, and results of the Public Patient Involvement studies that they performed. This detailed description and the result was moved to the appendix and has been replaced with a short summary of this work. (*Pages 45-47*)

Section 2.1.5 Modification of the Emailing Timeline to Reduce Attrition in the LBBS- This section was added. It describes how and why the author modified the questionnaire emailing timeline in the LBBS. This was not previously described and is an important part of the work carried out by the author in developing the study. (Pages 46-47)

Section 2.2.2 Data Entry and Coding - This section was added. It provided a description of how the data was entered and coded for the use in the statistical packages. This was a significant proportion of work performed by the author that was not previously described in the thesis. (*Pages 47-48*)

Section 2.2.8 Correction for Multiple Testing Analysis - A discussion on correction for multiple testing during analysis of the LBBS data was added. This discussion identifies why multiple testing is often performed, the benefits and disadvantages of using correction for multiple testing and why it was not used during analysis of the LBBS. (Page 51)

2.2.9- Justification and Development of the Methods of Longitudinal Analysis in the LBBS- This section was added to the thesis to provide a more detailed description of how the longitudinal analysis methods used in the LBBS were developed. Prior to this modification this description was lacking detail. It now discusses in much detail how and why the method of multi-level mixed effects modelling was chosen and the advantages and disadvantages of its use. This section also

describes the exactly process that was used to develop the model and how the final results were obtained. *(Pages 52-53)*

Section 2.2.11 Presentation of the Analysis of the LBBS in this Thesis - This section was added. It provides a description of how the results of the LBBS will be presented in Chapter 3. Due to the large amount of results presented in the thesis it was difficult for the reader to follow previously. The addition of this section makes it easier for the reader to understand how and why the results will be presented in this thesis. This will make the thesis easier to read and follow as a whole. (Pages 52-54)

Section 3 Chapter 3- At the beginning of each subsection a description of what the results that follow show and how they relate to the study aims. This will add a narrative thread to the presentation of the results and help the reader follow the analysis. **(Pages 55-122)**

Section 3.5 Cross Sectional Analysis- Tables describing other health conditions were modified so that the types of underlying condition were in alphabetical order. Diseases within the condition type were rearranged so that they were in order of frequency in the cohort. *(Pages 69-117)*

Section 3.5 Cross Sectional Analysis- Tables describing the median, interquartile range, mean and mode score of each risk factor/exposure group was moved from the appendix to the main body of the text where the significant differences in LRSQ scores are described. This makes the presentation of the results of the cross sectional analysis more consistent throughout the 4 time points that are presented in this thesis. This will make the analysis easier to follow for the reader. The box and whisker diagrams of significant results only remains the same. (Pages 69-117)

Section 4.4 The Clinical Relevance of the LRSQ Score – This section was added. It describes what would be considered a clinically significant or parentally important difference in LRSQ score was adding. Although this has not been previously described the addition of this section gives the thesis more clinical relevance. (Page 129)

In addition to this, spelling, grammatical, and syntax errors have been corrected. Missing graph axis labels have been added.

Abstract

The Liverpool Baby Breathing Study- The cross-sectional and longitudinal assessment of respiratory symptoms and the impact on quality of life in the first 22 months of life by Joshua Stead.

Introduction. The Liverpool Baby Breathing Study (LBBS) is a longitudinal birth cohort study that investigates natural history of respiratory disease in preschool children in Merseyside. Merseyside has a high prevalence of risk factors for preschool respiratory disease. Preschool children in Liverpool have high rates of hospitalisation for asthma and bronchiolitis. Respiratory disease impacts on the quality of life (QoL) of infants and their families. The LBBS also assesses QoL impact on infants and their families, which makes the LBBS a novel study.

Study Aims. The aims of this thesis are: 1. To create a profile of the infants and mothers enrolled in the LBBS and compare this to the eligible population. 2. To describe respiratory symptoms in the cohort using the Liverpool Respiratory Symptom Questionnaire (LRSQ) from birth until 22 months of age. 3. To determine the differences in LRSQ scores in infants exposed to various risk factors. 4. To determine the change in the LRSQ score over time and the variability in this change with exposure. 5. To validate the LRSQ in a longitudinal study.

Methods. The LBBS uses the Liverpool Respiratory Symptom Questionnaire (LRSQ). Recruitment for the study was performed in the Liverpool Women's Hospital (LWH). Eligible births were those infants who lived in Liverpool postcodes L1-38, spoke sufficient English and were cared for by their parents. Questionnaires were sent to participants by post or through an automated online emailing system. Recruitment rates and response rates were described using descriptive statistics. Sensitivity analysis was performed, comparing the drop out population's exposure variables to the remaining cohort. The profile of the cohort was compared to the profile of eligible births in the LWH by χ^2 analysis. Cross-sectional analysis compared those exposed to risk factors using χ^2 , Mann-Whitney U, Kruskal-Wallis analysis and Fisher's Exact Test. The cross-sectional analysis of the LBBS was weighted to ensure the index of multiple deprivations (IMD) decile distribution of the cohort was comparable to the eligible births in the LWH. The longitudinal analysis of the LBBS was done using multi-level mixed effects models. The internal consistency of the LRSQ was assessed using Cronbach alpha coefficients.

Results. 694 of the interested mothers (5.57% of all eligible births) consented to the study and completed the initial questionnaire. Attrition in the study was 44.0%, 60.4% and 63.0% at 10, 16 and 22 months. Mothers participating and retained in the LBBS were older, less deprived, more highly educated, breastfed more and smoked less in pregnancy than mothers of all eligible births. Infants were more likely to be of white or mixed ethnicity and born preterm than the eligible births. Cross-sectional analysis showed nursery attendance, being male, decreasing maternal age, decreasing maternal education, preterm birth, low birth weight, a family history of atopy and presence of other household children increased respiratory symptoms. Longitudinally analysis found nursery attendance, being male, preterm birth, low birth weight, a family history of atopy and other household children increased respiratory symptoms over time. Breastfeeding, increasing gestational age and sharing a bedroom decreased respiratory symptoms over the first 22 months of life. Respiratory symptoms had a major impact on the QoL of both the infant and their families. There were unexpected and mixed results regarding the effect of smoke exposure on respiratory health in the LBBS. The LRSQ had acceptable to good internal consistency.

Conclusion. The LBBS deployed the LRSQ using contemporary technology, and has been accessible to mothers in Liverpool across all deciles between 2012 through 2017. LBBS has validated use of the LRSQ in a longitudinal birth cohort. Breastfeeding had the greatest protective effect with a positive duration dependent dose-effect on respiratory symptoms and QoL. Nursery school attendance had the greatest deleterious effect on respiratory symptoms and QoL.

Contents

Errata1
Abstract
Contents
List of Tables9
List of Figures11
List of Abbreviations14
Acknowledgements15
Collaboration16
Foreword17
Chapter 1-Introduction18
1.1-Introduction to the Liverpool Baby Breathing Study18
1.2-Respiratory Disease in Preschool Children20
1.2.1-Respiratory Disease and Symptoms20
1.2.2-The Burden of Paediatric Respiratory Disease21
1.3-Risk Factors for Paediatric Respiratory Disease22
1.3.1-Air Pollution23
1.3.2-Breastfeeding23
1.3.3-Smoke Exposure23
1.3.4-Prematurity24
1.3.5-Low Birth Weight24
1.3.6-Household Status24
1.3.7-FH of Atopy25
1.3.8-Other Risk Factors and Exposures25
1.4-Measures of Respiratory Disease in Preschool Children26
1.5- Respiratory Symptom Questionnaires27
1.5.1-Paediatric Respiratory Symptoms Questionnaires27
1.5.2-The Liverpool Respiratory Symptom Questionnaire

1.6- Birth Cohort Studies	37
1.6.1- Summary of Recruitment in Birth Cohort Studies	38
1.6.2- Summary of Attrition in Birth Cohort Studies	39
1.7-Merseyside Health Profile and Respiratory Studies	40
1.7.1-Merseyside Health Profile	40
1.7.2-Merseyside Respiratory Studies	40
1.8-Conclusion	41
Chapter 2 -Methods	43
2.1-Section 1. LBBS Study Design and Development Summary	44
2.1.1-Aims	44
2.1.2-Protocol Development	44
2.1.3-Questionnaire and Study Material Development	44
2.1.4-Recruitment	44
2.1.5- Questionnaire Distribution	45
2.1.6- Batch Testing	46
2.2-Section 2. Methods Used in the Statistical Analysis of the LBBS	46
2.2.1-Aims	46
2.2.2- Data Entry and Coding	46
2.2.3- Statistical Packages Used During Analysis	47
2.2.4-Recruitment	47
2.2.5-Response Rates and Attrition	47
2.2.6-Profiles of Mothers in the LBBS	48
2.2.7-Profile of Infants in the LBBS	48
2.2.8-Cross-sectional Analysis	48
2.2.9-Longitudinal Analysis	51
2.2.10-Questionnaire Validation	52
2.2.11-Presentation of the Analysis of the LBBS in this Thesis	52
	55

	55
3.2-Drop Out	57
3.2.1-Summary	57
3.2.2-Recognition of Drop Out and Bias	58
3.2.3-Choice of Analysis to Mitigate for Drop Out Bias	61
3.2.4-Justification of Weighting Data – The Effect on Variables in the Cohort	61
3.3-Profiles of Mothers and Families in the Liverpool Baby Breathing Study	64
3.3.1-Deprivation	64
3.3.2-Maternal Age	64
3.3.3-Maternal Education	65
3.3.4-Smoking	65
3.3.5-Breastfeeding	66
3.3.6-Family History of Atopy	67
3.4-Profile of Infants in the Liverpool Baby Breathing Study	67
3.4.1-Sex	67
3.4.2-Multiple Births	67
3.4.3-Ethnicity	68
3.4.3-Ethnicity	68 68
3.4.3-Ethnicity3.4.4-Gestation3.4.5-Birth Weight	68 68 69
 3.4.3-Ethnicity 3.4.4-Gestation 3.4.5-Birth Weight 3.5-Cross-sectional Analysis 	68 68 69 69
 3.4.3-Ethnicity	68 68 69 69 70
 3.4.3-Ethnicity	68 69 69 70 83
 3.4.3-Ethnicity	68 69 70 83 96
 3.4.3-Ethnicity	68 69 70 83 96 107
 3.4.3-Ethnicity	68 69 70 83 96 107 118
 3.4.3-Ethnicity	68 69 70 70 96 107 118 118
 3.4.3-Ethnicity	68 69 70 70

	3.6.4-Symptoms with Colds	119
	3.6.5-Symptoms without Colds	119
	3.6.6-Symptoms on Increased Activity	120
	3.6.7-Other Respiratory Symptoms	120
	3.6.8-Childs QoL	120
	3.6.9-Family QoL	121
	3.7-Questionnaire Validation	122
Cł	napter 4- Discussion	123
	4.1-Recruitment and Response Rates	123
	4.2-Profiles of Mothers in the Liverpool Baby Breathing Study	127
	4.3-Profile of Infants in the Liverpool Baby Breathing Study	128
	4.4- The Clinical Relevance of the LRSQ Score	129
	4.5-Cross-sectional Analysis	130
	4.5.1-Sex	130
	4.5.2-Multiple Birth	130
	4.5.3-Ethnicity	131
	4.5.4-Maternal Age	131
	4.5.5-Maternal Education	132
	4.5.6-Gestation	134
	4.5.7-Birth weight	134
	4.5.8-Breastfeeding	135
	4.5.9-Smoking Exposure	136
	4.5.10-Family History of Atopy	137
	4.5.11-Nursery Attendance	138
	4.5.12-Sharing a Bedroom	139
	4.5.13-Household Children	140
	4.6-Longitudinal Analysis	141
	4.7-Questionnaire Validity	142

4.8 The Use of Contemporary Technology in the LBBS	143	
4.9-Strengths of the LBBS	144	
4.10-Weaknesses of the LBBS	145	
4.11-Future Recommendations	146	
Chapter 5-Conclusions from the LBBS	148	
Appendix	150	
References		

List of Tables

Table 1: The phenotypes of wheeze as described in literature 21
Table 2: Birth cohort studies identified by literature review
Table 3: Phases of the LBBS
Table 4: Recruitment rates in the LBBS
Table 5: Attrition in the LBBS 57
Table 6: Previous questionnaire total LRSQ score comparing the drop out population to the
remaining LBBS cohort
Table 7: The effect of weighting of the cohort by IMD decile on smoking exposures in the
LBBS
Table 8: Ethnicity in the LBBS in comparison to LWH and Liverpool Births
Table 9: Underlying medical conditions in the LBBS cohort at 4 months71
Table 10: Comparison of LRSQ scores between different exposure/risk factor groups in the
LBBS at 4 months
Table 11: Summary statistics of domain scores at 4 months 75
Table 12: Underlying medical conditions in the LBBS at 10 months 84
Table 13: Comparison of LRSQ scores between different exposure/risk factor groups in the
LBBS at 10 months
Table 14: Summary statistics of domain scores at 10 months 88
Table 15: Underlying medical conditions in the LBBS at 16 months 96
Table 16: Comparison of LRSQ scores between different exposure/risk factor groups in the
LBBS at 16 months
Table 17: Summary statistics of domain scores at 16 months 100
Table 18: Underlying medical conditions in the LBBS at 16 months 108
Table 19: Comparison of LRSQ scores between different exposure/risk factor groups in the
LBBS at 22 months
Table 20: Summary statistics of domain scores at 22 months 111
Table 21: Multi-level mixed effects model for Total LRSQ scores 118
Table 22: Multi-level mixed effects model for domain 1 (day time symptoms) 118
Table 23:Multi-level mixed effects model for domain 2 (night-time symptoms)
Table 24:Multi-level mixed effects model for domain 3 (symptoms with colds) 119
Table 25:Multi-level mixed effects model for domain 4 (symptoms without colds)
Table 26: Multi-level mixed effects model for domain 5 (symptoms on increased activity)

Table 27: Multi-level mixed effects model for domain 6 (other respiratory symptoms) 120
Table 28:Multi-level mixed effects model for domain 7 (infant's quality of life- higher score
indicates worse quality of life)121
Table 29:Multi-level mixed effects model for domain 8 (mother's quality of life- higher score
indicates worse quality of life)
Table 30: Cronbach's alpha coefficients for the LRSQ domains in the LBBS cohort
Table 31: Comparison of the internal validity of the LRSQ in the LBBS compared to previous
studies
Table 32: Recruitment rates during recruitment pilot study
Table 33: Properties of AFC [®] and JotForm [®] 196

List of Figures

Figure 1	Page 1 of the Liverpool Respiratory Symptom Questionnaire	32
Figure 2	Page 2 of the Liverpool Respiratory Symptom Questionnaire	33
Figure 3	Page 3 of the Liverpool Respiratory Symptom Questionnaire	34
 Figure 4	Page 4 of the Liverpool Respiratory Symptom Questionnaire	35
Figure 5	Inclusion and Exclusion criteria in the LBBS	44
Figure 6	Risk factors and exposures assessed in the LBBS	49
Figure 7	Flow diagram of the recruitment in the LBBS	56
Figure 8	Flow diagram of attrition in the LBBS	58
Figure 9	Accumulative percentage drop out in the LBBS in each IMD decile at each questionnaire	59
Figure 10	Accumulative percentage drop out in the LBBS by mother's highest qualification attained	59
Figure 11	Percentage of the cohort with mothers in each age category at each questionnaire in the LBBS	60
Figure 12	Percentage of the cohort exposed to tobacco smoke at each questionnaire in the LBBS	60
Figure 13	Percentage of the cohort that were breastfed at each questionnaire in the LBBS	60
Figure 14	Percentage of the cohort breastfed in the LBBS before and after weighting by IMD decile	61
Figure 15	Percentage of the cohort in the LBBS exposed to maternal smoking in pregnancy before and after weighting by IMD decile	62
Figure 16	Percentage of the cohort in the LBBS exposed to ETS in pregnancy from other household smokers before and after weighting by IMD decile	62
Figure 17	Percentage of LBBS cohort exposed to any smoking in pregnancy before and after weighting by IMD decile	62
Figure 18	Percentage of the cohort in the LBBS exposed to household smoking postnatally before and after weighting by IMD decile	62
Figure 19	Percentage of the LBBS cohort in each IMD decile in comparison to LWH births and the Liverpool population	64
Figure 20	Percentage of mothers in the LBBS cohort in age group in comparison to LWH, Liverpool and England births	65
Figure 21	Percentage of the cohort by highest qualification attained in the LBBS in comparison to the population of Merseyside and England	65
Figure 22	Percentage of mothers that smoked during pregnancy in the LBBS in comparison to LWH, Liverpool and England births	65
Figure 23	The location of smoking of household smokers during pregnancy in the LBBS	66
Figure 24	Percentage of breastfed infants in the LBBS in comparison to the rates of breastfeeding in LWH, Liverpool and England	66
Figure 25	Duration of breastfeeding in breastfed infants in the LBBS	66
Figure 26	Combinations of atopic disease in first degree relatives in the LBBS	67
Figure 27	Proportion of males and females in the LBBS in comparison to the population of Liverpool, the North West and Great Britain	67
Figure 28	Gestational age of the LBBS cohort in comparison the LWH births	68
Figure 29	Birth weight of the LBBS cohort in comparison the LWH, North West and England births	69
Figure 30	The number of other household children in the LBBS households at 4 months of age.	70
Figure 31	Percentage of households with smokers at 4 months in the LBBS	70
Figure 32	Histogram of Total LRSQ scores at 4 months	72
Figure 33	Total LRSQ scores by infant sex	74
Figure 34	Total LRSQ Scores by maternal age group	74
Figure 35	Total LRSQ scores by FH of atopy	74

Figure 36	Total LRSQ scores by sharing a bedroom			
Figure 37	Respiratory symptom prevalence per 100 infants in the first 4 months of life			
Figure 38	Percentage of the cohort that attended their GP or hospital with respiratory symptoms in the first 4 months of life by maternal age group			
Figure 39	Percentage of the cohort that attended their GP or hospital with respiratory symptoms in the first 4 months of life by highest maternal qualification attained	80		
Figure 40	The impact of respiratory disease on the infant's QoL in the first 4 months in the LBBS (per 100 infants)	81		
Figure 41	The impact of respiratory disease on the infant's mother's QoL in the first 4 months in the LBBS (per 100 infants)	81		
Figure 42	Scatter graph of respiratory symptom scores and quality of life scores at 4 months	82		
Figure 43	Number of children in households at 10 months	83		
Figure 44	Household smoking at 10 months	83		
Figure 45	Histogram of Total LRSQ Scores at 10 Months	85		
Figure 46	Total LRSQ scores at 10 months by infant sex	87		
Figure 47	Total LRSQ score at 10 months by infant ethnicity	87		
Figure 48	Total LRSQ score at 10 months by preterm birth	87		
Figure 49	Total LRSQ at 10 months by nursery attendance	87		
Figure 50	Total LRSQ score at 10 months by maternal age	87		
Figure 51	Respiratory symptom prevalence at 10 months per 100 infants	91		
Figure 52	Percentage of the cohort that attended their GP or hospital with respiratory symptoms in the first 10 months of life by maternal age group	92		
Figure 53	Percentage of the cohort that attended their GP or hospital with respiratory symptoms in the first 10 months of life by highest maternal qualification attained	92		
Figure 54	The impact of respiratory disease on the infant's QoL at 10 months in the LBBS (per 100 infants)	93		
Figure 55	The impact of respiratory disease on the infant's mother's QoL at 10 months in the LBBS (per 100 infants)	93		
Figure 56	Scatter diagram showing respiratory symptom scores against QoL Scores at 10 months	95		
Figure 57	Household smoking in the LBBS at 16 months	96		
Figure 58	Histogram of the total LRSQ scores at 16 months	97		
Figure 59	Total LRSQ scores at 16 months by infant sex	99		
Figure 60	Total LRSQ score at 16 months by nursery attendance	99		
Figure 61	Total LRSQ scores at 16 months by ETS exposure in pregnancy	99		
Figure 62	Total LRSQ score at 16 months by IMD decile	99		
Figure 63	Respiratory symptom prevalence in the LBBS at 16 months	102		
Figure 64	The impact of respiratory disease on the infant's QoL 16 months in the LBBS (per 100 infants)	105		
Figure 65	The impact of respiratory disease on the infant's mother's QoL at 16 months in the LBBS (per 100 infants)	105		
Figure 66	Scatter diagram of respiratory symptom score and quality of life score at 16 months	106		
Figure 67	Household smoking in the LBBS cohort at 22 months	107		
Figure 68	Histogram of total LRSQ scores at 22 months	108		
Figure 69	Total LRSQ scores at 22 months by maternal education	110		
Figure 70	Total LRSQ score at 22 months by other household smoking in pregnancy	110		
Figure 71	Total LRSQ scores at 22 months by nursery attendance	110		
Figure 72	Respiratory symptom prevalence at in the three months prior to 22 months old in the LBBS	113		
	·			

Fi	gure 73	Percentage of the cohort that attended their GP or hospital with respiratory symptoms at 22 months of life by maternal age group	115
Fi	gure 74	Percentage of the cohort that attended their GP or hospital with respiratory symptoms at 22 months of life by highest maternal qualification attained	115
Fi	gure 75	The impact of respiratory disease on the infant's QoL at 22 months in the LBBS (per 100 infants)	116
Fi	gure 76	The impact of respiratory disease on the infant's mother's QoL at 22 months in the LBBS (per 100 infants)	116
Fi	gure 77	Scatter diagram of respiratory symptom score and quality of life score at 22 months	117

List of Abbreviations

A&E	Accident and Emergency	
AFC [®]	Adobe Forms Central ® The Aven Longitudinal Study of Parents and Children	
ALSPAC	The Avon Longitudinal Study of Parents and Children The Asthma Therapy Assessment Questionnaire	
ATAQ	The Asthma Therapy Assessment Questionnaire	
BiB	Born in Bradford	
BMI	Body Mass Index	
CAQ	Childhood Asthma Questionnaire	
CAT	The COPD Assessment Test	
CCQ	The Clinical COPD Questionnaire	
CF	Cystic Fibrosis	
COPD	Chronic Obstructive Pulmonary Disease	
DNA	Deoxyribonucleic acid	
ECRHS	The European Community Respiratory Health Survey	
EISL-WQ	The International Study of Wheezing Infants Questionnaire	
ELSPAC	The European Longitudinal Study of Pregnancy and Childhood	
ETS	Environmental Tobacco Smoke	
FEV1	Forced expiratory volume in one second	
FH	Family History	
FOT	Forced Oscillation Technique	
FVC	Forced vital capacity	
GI	Gastrointestinal	
GP	General Practitioner	
lg	Immunoglobulin	
IMD	Index of Multiple Deprivations	
ISSAC	The International Study of Asthma and Allergy in Childhood	
K-BILD	King's Brief Interstitial Lung Disease	
LBBS	Liverpool Baby Breathing Study	
LRSQ	Liverpool Respiratory Symptom Questionnaire	
LWH	Liverpool women's Hospital	
MASS	Manchester Astrima and Allergy Study	
MEDALL	The Mechanisms of development of Allergy	
MIRC		
NHS		
NICU	Neonatal Intensive Care Unit	
	Paediatric Astrinia Quality of Life Questionnaire	
	The Drevention and Incidence of Asthma and Mite Allermy	
	Public Patient Involvement	
	The Quality of Life for Recoiratory Illnoss Quastionnaira	
	Rechiratory Inductance Plothycmography	
	Respiratory Syncytial Virus	
RTI	Respiratory Tract Infection	
SoB	Shortness of Breath	
	Statistical Package for Social Sciences	
5F55 CCNI		
	Jeret for Poeniratory and Asthma Control in Vide	
W/VCAP	Wuthenshawe Community Asthma Broject	
VV I CAF	wythenshawe community Astima Ploject	

Acknowledgements

The help and support of many different people over the last year has made the production of this thesis possible.

I would like to thank my supervisor, Dr Calum Semple. The work in producing this thesis would not have been possible in any way without him. His time and support through the year has made the production of this piece of work possible. Without his guidance, expertise, teaching and contacts, the production of this thesis would have been impossible. For all that, I am incredibly thankful.

I would like to thank Professor Ben Shaw for providing data from the LWH eligible births, without which a large proportion of this work would not have been possible.

I continue my thanks to Dr Semple and Professor Shaw and would like to thank Professor Paul McNamara, Professor Kevin Sothern and the previous MPhil Students, Rosanna Pickles, Bethan Griffith, Sanjay Patel and Jennifer Barclay for the initiation and running of the study before my involvement began. Without this previous work the study would not have existed for me to be involved with, and it is unlikely I would have found a project which would have enjoyed working on as much as this study.

I am extremely grateful to Dr Steven Lane from the University of Liverpool who has developed the methods for the longitudinal analysis of the study and helped me throughout the year with the analysis of the study. Without his help, teaching and knowledge, the analysis of this study would not have been possible. I cannot thank him enough for his time, help and input into the study.

I extend my thanks to the other MPhil Students this year for helping me through the year with a range of things from administration to providing conversations about how we are all getting on. I would particularly like to thank Pearl Ampha who has been based at Alder Hey Children's Hospital with myself; she has always been there to ask quick questions to and provided me with support I have needed through this year.

Finally I would like to thank my family and friends for their support over the last year whether that be financially, emotionally or socially. Without them this year would have been incredibly tough and much more difficult than it was. I cannot thank them enough.

Collaboration

The following work was completed either previous to my involvement in the LBBS or during my involvement in the study by other investigators described below.

The initiation of the Liverpool Baby Breathing Study and protocol development was conducted by Miss Rosanna Pickles, Miss Bethan Griffith, Dr Calum Semple, Professor Paul McNamara, Professor Kevin Sothern, and Professor Ben Shaw.

Development and initiation of the distribution of questionnaires to the study participants was conducted by Miss Rosanna Pickles, Bethan Griffith, Sanjay Patel and Jennifer Barclay.

Recruitment to the LBBS Study was conducted by Miss Rosanna Pickles, Miss Bethan Griffith, Mr Sanjay Patel and Miss Jennifer Barclay.

The collection and production of the demographic data of all the eligible births in the Liverpool Women's Hospital during the recruitment period was performed by Professor Ben Shaw in the LWH.

The development and validation of the LRSQ was conducted by Dr Ruth Trinick, Professor Kevin Southern, Professor Paul McNamara, Dr Colin Powell, Dr Arturo Solis and Professor Ben Shaw.

The development of the methods for the longitudinal analysis of the study was developed by Dr Steven Lane from the University of Liverpool.

Foreword

This thesis is comprised of five chapters.

Chapter one provides an introduction to the LBBS. It discusses background information respiratory disease in the preschool population. Recruitment and attrition in previous birth cohort studies are discussed. Measures of paediatric lung function are discussed along with analysis of previously used respiratory symptoms questionnaires and the in depth analysis of the Liverpool Respiratory Symptom Questionnaire. The health profile in Liverpool is discussed providing reasons why Liverpool is an ideal city to conduct a study of this nature.

Chapter two provided the details of the methods of the LBBS. Section one of this chapter is an outline of previous work that had been conducted; this describes the questionnaire development, initiation of the study and recruitment. Section two provides a detailed description of work performed by the author. It describes the methods of analysis of recruitment rates, dropout rates and the questionnaire responses.

Chapter three describes the results of the LBBS. This chapter is split into seven sections. Section one is the results of the analysis of recruitment and response rates in the LBBS. Section two describes attrition in the LBBS. This section provides a description of how the study accounted for the confounding effects created by the discrepancies between the drop out population and the population remaining in the study. Section three and four describe the profiles of the mothers and infants in the LBBS. In these sections there is a comparison of the LBBS profiles to the population of Liverpool and the eligible births in the LWH. Section five describes the results of the cross sessional analysis of each of the four questionnaires at four different time points in the first 22 months of life. Section six provides the results of the longitudinal analysis of the LBBS over the first 22 months of life. Section seven provides the results of the validation of the LRSQ when used in longitudinal studies.

Chapter four provides a discussion on the findings from the LBBS and how the current study's links to previous findings. The strengths and weaknesses of the LBBS and recommendations for future work are also discussed in chapter four.

Chapter five is a summary of the conclusions from the LBBS.

The Liverpool Baby Breathing Study

Chapter 1-Introduction

This chapter provides an introduction to the LBBS. It has a description of why the Liverpool Baby Breathing Study (LBBS) began, and why it is novel. It summarises the work done by previous students, prior to the author's involvement, and the work completed by the author leading to the production of this thesis. An overview of paediatric respiratory disease is given along with a discussion on the burden of respiratory disease in the paediatric population. Common risk factors previously identified for paediatric respiratory disease are discussed. Methods used to measure respiratory disease and lung function in the paediatric age group and the problems they possess are described. The recruitment methods and methods used to reduce attrition in previous birth cohort studies are described. Previously used respiratory symptom questionnaires are discussed, followed by an in depth description of the Liverpool Respiratory Symptom Questionnaire (LRSQ) that is used in the LBBS. The health profile of the Liverpool population and reasons why Liverpool is an ideal city to conduct a study of this nature is discussed.

1.1-Introduction to the Liverpool Baby Breathing Study

The Liverpool Baby Breathing Study (LBBS) was formerly known as 'The Liverpool Respiratory Birth Cohort Study'. It is a longitudinal birth cohort study that assesses the respiratory symptoms of preschool children from Liverpool. It assesses the impact of these symptoms on both the infant's and their family's quality of life (QoL). The aims of the study as outlined in the study protocol are;

- To create a profile of the infants and mothers enrolled in the LBBS and compare this to the population of Liverpool and those born in the LWH.
- To describe respiratory symptoms of preschool children in Liverpool using the LRSQ from birth until 5 years of age by biannual assessment.
- To determine the differences in LRSQ scores in the different populations at each questionnaire time point.
- To determine the change in the LRSQ scores over time and the variability in this change between different population groups in the LBBS.
- To validate the LRSQ in a longitudinal study.

The LBBS assesses infants biannually from four months to five years of age by means of a parent completed questionnaire; the Liverpool Respiratory Symptom Questionnaire (LRSQ). The LRSQ was developed by Powel et al. at Alder Hey Children's Hospital and Liverpool Women's Hospital (LWH). The LRSQ assesses a number of common respiratory symptoms, including cough, wheeze and shortness of breath (SoB), and the effect they have on the infant's and their family's QoL (1). In the LBBS the LRSQ has been used alongside a questionnaire assessing demographics and exposures developed by Rosanna Pickles and Bethan Griffith (2, 3). This questionnaire explored common risk factors associated with respiratory disease in preschool children such as deprivation (4), tobacco smoke exposure prenatally (5) and postnatally (6), breastfeeding (7) and prematurity (8). These questionnaires, along with assessment of healthcare service attendance allowed the study of the prevalence, the natural history and the impact of respiratory disease in preschool children from Liverpool. This study is now in its 4th year, with the whole cohort having completed the 5th questionnaire at 28 months of life. For the purpose of this thesis the first four questionnaires are discussed.

Liverpool has a high prevalence of risk factors for paediatric respiratory disease in its population such as deprivation (9), smoking (both in pregnancy (10) and postnatally (11)) and the low levels of breastfeeding (12). Liverpool has high rates of respiratory disease in both the adult and the paediatric population (13, 14); there are high levels of bronchiolitis and asthma hospitalisations in Liverpool compared to other areas of England (15). The LWH provides the mainstay of maternity care in Liverpool with around 8500 births per year. This has provided the research team with an ideal and valuable centre for recruitment and access to the majority of births in Liverpool.

The LRSQ has previously been validated for use in the preschool and infant population (1). It has been shown to distinguish well between those with Cystic Fibrosis (CF) from those without, and those who were unwell from those who were stable in both preschool children and in those 6-12 years old (16). It has successfully been used to examine the relationship between neurodisability and respiratory symptoms (17) and is being increasingly used in studies of respiratory disease in the preschool age group. This is the first time it has been used in a longitudinal study and on such a large population. LRSQ has allowed the researchers in the current study to explore the prevalence and patterns of respiratory symptoms and how these change over time in the preschool population of Liverpool.

My involvement in the LBBS began in September 2016 after previous work by Rosanna Pickles, Bethan Griffith, Sanjay Patel, and Jennifer Barclay under the supervision of Dr Calum Semple. At this point, the study protocol and supplementary documents had been finalised and ethical approval of the study had been granted in May 2012 by the East Midlands Research Ethics Committee, part of the NHS Research Authority (Reference: 12/EM/1904). Recruitment had finished, having occurred between October 2012 and November 2014 and data collection was into its fourth year. All the Participants had completed four questionnaires, and the fifth questionnaire was due to be fully completed in May 2017. My role within the LBBS research team was to analyse the data from the initial four questionnaires and continue with maintenance of the study. To achieve this, and to produce this thesis, the following objectives were completed:

- Perform a literature review on previous birth cohort studies and asses their design, recruitment strategies and ways in which attrition was reduced.
- Produce a systematic method of analysis of the LBBS data both (cross sectional and longitudinal) by having discussions with experts at the University of Liverpool.
- Modify the study to reduce attrition by inviting participants back into the study if they had not intended to leave.
- Code the raw data so that it could be analysed by the statistics packages used in the LBBS.
- Perform the analysis of the LBBS data, discuss the findings in line with previous findings, and investigate unexpected results.

1.2-Respiratory Disease in Preschool Children

1.2.1-Respiratory Disease and Symptoms

Cough is the most common problem managed by General Practitioners (GP's) and is most common in the preschool age group. Respiratory tract infection (RTI) and asthma are the leading causes of cough in the preschool age group (18). Two thirds of children will attend their GP with a RTI and three quarters of these will have a cough. Seventy five per cent of coughs will show improvement by one week however one in eight experience complications within two weeks of presentation (19). Very few children with recurrent cough go on to develop asthma with wheeze at four years (18). One in three children develop wheezing associated with a RTI in first three years of life but 60% of these will stop wheezing by six years (20). There are many different phenotypes of wheeze described in literature, most commonly distinguished by their pattern (21), timing of onset (21) and the patient's atopic status(22). These are described in the tables below.

Pattern	Episodic (viral) Wheeze	Wheezing during discrete time periods associated with viral cold. No wheeze in between episodes
(20)	Multiple Trigger Wheeze	Discrete exacerbations with wheeze in between.
	Transient	Symptoms commence before three years. No symptoms by or after six years
Timing (20)	Persistent	Symptoms commence before three years and continue until and after six years
	Late-Onset	Symptoms start after three years
Atopy	Atopic	Sensitized to one aeroallergen by age of six years, reaction to this allergen causes wheeze
(21)	Non-Atopic	Wheeze with viral infections during years two and three of life

Table 1: The phenotypes of wheeze as described in literature

Those with transient early wheeze can be distinguished from other types of wheeze by a deficit in lung function which is evident shortly after birth, before any RTI (20). At birth, persistent wheezers have similar lung function to non-wheezers but had nearly 50% better lung function than transient wheezers. At six years however, persistent wheezers had significantly reduced lung function compared to other phenotypes of wheeze. Persistent wheezers have more frequent symptoms and more frequently have eczema, rhinitis and a family history (FH) of maternal asthma (20).

1.2.2-The Burden of Paediatric Respiratory Disease

In the United Kingdom (UK) across all age groups one in five deaths are due to respiratory disease, which is higher than ischemic heart disease. Only countries with poor health care systems and low funding having higher rates of respiratory disease in Europe. In 2006 respiratory disease was reported to cost the NHS around £6 billion and account for around one million hospital admissions (23).

Respiratory disease is the most common reported long term illness in children and babies with around one in five children and babies having a diagnosis of asthma in 2006 (23). Asthma alone costs the National Health Service (NHS) of around £672 million per year which accounts for between one per cent and two per cent of the NHS healthcare budget (23). In 2009 the estimated rate of asthma in the UK was 136.6/10,000 person years, and the 18-year prevalence was 22.9%(24). The Born in Bradford (BiB) cohort described the rate of wheezing disorders to be 40.3/1000 person years in the first five years of life and identified that one of five children will have been diagnosed with a wheezing disorder or rhinitis by five years of age (25). Respiratory Syncytial Virus (RSV) will effect one in five infants in first year of life; between two and three per cent of these will require hospital admission (26). In the UK RSV accounts for around 450,000 GP consultations, 29,000 hospitalizations and 83 deaths per year with the highest rates in those less than six months old (27). CF has an incidence of around one in 3,500 with a mortality rate between two and five per cent at two years of age, 10 to 30 per cent at five years and 35 per cent at 15 years. Children with CF will experience on average three exacerbations per year (28). The World Health Organization (WHO) reported that the biggest single cause of death globally in the first five years of life was acute respiratory illness such as pneumonia (29).

Respiratory disease impacts on the QoL of both the patients and their family. QoL was reported to be lower in infants aged between two months and five years of age with wheeze (30) and was lower in both children and their caregivers in diagnosed and undiagnosed asthma patients (31). Both patients diagnosed with CF and their families reported a lower QoL (16). A worse QoL in the infant and their caregivers was reported in those with severe asthma (compared to mind/moderate asthma) and those with poor control of their asthma (compared to those with good control) (32). It has been shown that QoL of children aged one to 14 years was improved with increased control of their asthma (33).

These figures show the burden of respiratory disease on both the population of the UK and the NHS and why investigation into the natural history of respiratory disease in preschool children is important.

1.3-Risk Factors for Paediatric Respiratory Disease

Many risk factors for paediatric respiratory disease have been described in the literature. However the significance of these risk factors and the natural history of respiratory disease is disputed in the preschool population. The commonly reported risk factors for paediatric respiratory disease are discussed below and less reported risk factors are discussed later in this thesis.

1.3.1-Air Pollution

Coal Dust pollution was shown to be associated with increased respiratory symptoms in primary school children in both Merseyside (34) and elsewhere in the UK. In Nottingham it was shown that living within 90 meters of a road and therefore increased exposure to traffic air pollutants was associated with an increased prevalence of wheeze in children (35). In children already diagnosed with asthma, increased levels of air pollution was associated with an increase in symptoms and the use of bronchodilators (36). It was shown that hospital admissions due to respiratory symptoms and nitric oxide production in children's lungs were higher at times of higher air pollution levels; this may be due to inflammation of the airway epithelium (37, 38).

1.3.2-Breastfeeding

The evidence for the relationship of breastfeeding and wheeze over the ages of five is conflicting (39), however there is strong evidence supporting the benefit of breastfeeding on respiratory health before five years of age. Respiratory symptoms in the first six months of life decrease as breastfeeding time increased. Likewise a shorter breastfeeding time and none-exclusive breastfeeding was associated with increased risk of asthma symptoms (7). Breastfeeding for one month reduced the incidence of lower respiratory tract infection (LRTI) in the first six months of life (40). Breastfeeding for six months or more was associated with lower levels of wheezing, SoB, dry cough, and phlegm production during the first four years of life. The strongest association was in the first and second year of life (41). Breastfeeding may also reduce the negative effect of maternal smoking and premature birth on respiratory symptoms (7).

1.3.3-Smoke Exposure

There is a large evidence base supporting the hypothesis that smoking during pregnancy and postnatal environmental tobacco smoke (ETS) exposure is associated with a higher prevalence of respiratory symptoms in children. The effects of maternal and paternal smoking prenatally and postnatally is likely to be dependent on the individual and their environment. It may cause an adverse effect on the immune system, the structure and the function of the developing lung (42).

Maternal Smoking in Pregnancy

Maternal smoking in pregnancy has been shown to increase the incidence of wheeze in early childhood (43). Smoking in pregnancy, but not after, increased recurrent wheeze and the

diagnosis of asthma in the first 2 years of life (5). Meta-analysis of eight European cohort studies has shown that maternal smoking in pregnancy increased the risk of both wheeze and asthma between four and six years of age regardless of ETS after birth. Maternal smoking in pregnancy had a dose response relationship with the number of wheezing episodes, persistent wheezing and asthma diagnosis at 15 years of age (44).

Postnatal Smoking

Infants with a mother who smoked were shown to be three times more likely to have a LRTI in the first three years of life and had a younger age of first infection than those who did not have a mother who smoked. There was a dose response relationship shown between the number of cigarettes the mother smoked and the incidence of LRTI (45). Similarly in a Polish cohort study, three year olds who were exposed to ETS had more RTIs which required increased admissions to hospital and antibiotic treatment (6). Admissions to hospital with a LRTI in the first five years of life was higher in those infants with a smoking mother (46). ETS was associated with an increase incidence of wheeze in the first (47) and second year of life (5). In the first two years of life ETS was associated with new onset asthmatic and allergic symptoms and negatively associated with remission of symptoms between three and six years of age (48).

1.3.4-Prematurity

Preterm birth was associated with an increase prevalence of early remitting wheeze and persistent wheeze over the first seven years of life (49). In a Swiss cohort study, wheeze, inhalation therapy and hospitalisation in the first year of life were all associated with preterm birth (50). Babies born at less than 28 weeks are at risk of bronchopulmonary dysplasia (BPD) but there is mixed evidence on the difference in frequency of respiratory symptoms between premature babies with and without BPD (51).

1.3.5-Low Birth Weight

A meta-analysis study has shown that a low birth weight significantly increases the risk of developing asthma in childhood (52). A birthweight less than 1.5kg was a risk factor for diagnosis of asthma and hospitalisation due to asthma between birth and four years of age (53). A low birthweight was associated with reduced lung function at six weeks of age (54) and increased incidence of wheeze in the first three years of life (55).

1.3.6-Household Status

Poor housing increases respiratory disease in the first two years of life and a damp home environment increase cough and phlegm production at 10 years of age (56, 57). Household overcrowding may increase the risk of LRTI in children (58) however having siblings at home was associated with decreased risk of asthma diagnosis between 4.5 and 14 years of life (59-61). Bed sharing at 24 months was shown to be a risk factor for wheeze between two and six years of age (62). Increased household humidity was associated with an increased prevalence of wheeze (63), however higher dust endotoxin levels were shown to be protective of asthma (64). Higher household income was associated with increased respiratory health, while children from poorer families had an increased risk of LRTI and asthma (56, 65, 66). Excessively clean environments were associated with increased wheezing in childhood (67). Many of these findings are in line with the hygiene hypothesis of asthma (68, 69).

The Hygiene Hypothesis of Allergy

It was suggested by Strachan in 1989 that infection in early childhood, due to the presences of older siblings may reduce the development of allergic illness (69). Rook developed this theory further by suggesting that the proper development of the immune system depended on microbes that were present during evolution. The absence of these microbes may lead to the development of allergic disease (68).

1.3.7-FH of Atopy

Having a mother with a history of atopy is associated with infant asthma between three and five years of age (70). Maternal asthma was a risk factor for wheeze in girls of two years old and above and for boys of five years old and above. Paternal asthma was a risk factor for boys between two and five years (71). Children with mothers with asthma are more likely to have a diagnosis of asthma (72).

1.3.8-Other Risk Factors and Exposures

Wheeze during childhood is more common in boys than in girls (25). There is disagreement on the role of nursery attendance and its association with asthma and development of wheeze. Nursery attendance has been shown to be associated with an increased diagnosis of asthma between three and five years of age (70). In a different study however, it has been shown to reduce wheeze in the first five years of life (73). A further study showed nursery attendance to increase the symptoms of asthma during the first four years of life and be protective there after (74). A western diet with high levels of trans fatty acids may increase the risk of asthma, were as a Mediterranean diet with high fish, fruit and vegetables may be protective(75, 76). Sensitization to dogs, cats and dust mites were all associated with an increased risk of asthma (77, 78) and exposure to other furry pets was associated with increased early remitting and persistent wheeze(79).

1.4-Measures of Respiratory Disease in Preschool Children

There are many tests of lung function. The choice of test in preschool children is dependent on the intended outcome measure, the feasibility of the test, and cooperation of the child, and is often determined by their age. In 2007 the American Thoracic Society produced guidelines on the use of lung function tests in preschool children. Despite guidelines only 82.3% of children aged between four and five years can produced one acceptable spirometry result and only 54% can produce repeatable results (80).

Peak Expiratory flow rate is a simple and cheap assessment of airway obstruction which has produced acceptable measurements in preschool children (81). However, it is unreliable and has poor reproducibility so is only used as a screening tool (82).

A pneumotachometer and facemask can be used to measure tidal breathing, however it has not been tested for variability and the ability to detect disease in preschool children (80).

Respiratory Inductance Plethysmography (RIP) is performed by using volume signals monitored by bands placed around the chest and abdominal wall. In children between three and six years of age, RIP has successfully assessed changes in lung function and measured airflow obstruction (24, 83) however there is no data in infants younger than this. That being said, RIP is more successful than spirometry and the pneumotachometer system in very young children (84).

The Forced Oscillation Technique (FOT) is performed by exerting external pressure to the respiratory system and requires no breathing manoeuvres so little cooperation from the patient is required (85). FOT has been shown to be sensitive in assessing bronchodilation in children aged two to 18 years old and bronchial hyper-responsiveness in between five and 18 years old (86, 87).

The Interrupter Technique is the measure of the pressures produced during normal breathing after a sudden airflow interruption at the mouth (80). It is a feasible and repeatable measure of lung function and has been successfully used in those over three years for short term assessment of intervention response (80, 82).

The Multiple Breath Wash Out technique is as assessment of functional respiratory capacity and ventilation distribution which requires only passive breathing (80). Minimal cooperation is required and it has previously identified 73% of those with abnormal lung functions in CF patients between two and five years of age (88). Bronchial responsiveness is assessed by the administration of histamine or methacholine to the lungs to cause airway narrowing (82). This has a high sensitivity and positive predictive value in asthma, and is useful to determine the effectiveness of interventions however carries the same problems as spirometry when used in preschool children (80).

Nitric Oxide is produced in the lungs by eosinophils during airway inflammation in allergic asthmatic patients (89). Nitric oxide measurements have successfully been used in a study in school children (37) however not in preschool children.

In conclusion pulmonary function testing in those younger than five years old is difficult due to the reduced ability to voluntarily perform the manoeuvres required and if testing is essential it is often performed whilst the child is asleep. Although children between two and five years of age, are usually more cooperative than those below two years old, they still have short attention and are unlikely to perform the test as well as children over five years (80, 82). There is no routine or simple measurement of function suitable for regular use in preschool children, especially in those under two years of age. Because of this, respiratory symptom questionnaires are often used to assess respiratory symptoms in preschool children. Respiratory symptoms questionnaires are discussed in section 1.5.

1.5- Respiratory Symptom Questionnaires

There are several adult, self-reporting respiratory questionnaires that have been validated for various uses in the adult population (see appendix 5). Although some adult questionnaires may have informed the design of paediatric questionnaires, they are often very different. The majority of questionnaires assessing the paediatric age group are completed by the subject's parents. Several questionnaires have been developed and validated for use in the paediatric population and specifically the preschool population. These are discussed below.

1.5.1-Paediatric Respiratory Symptoms Questionnaires The Asthma Control Questionnaire for Children (ACQ)

The ACQ was developed to produce a reliable, valid and responsive questionnaire for clinical use, to assess asthma control in children aged six to 17 years old. It was developed without the need of a daily record of symptoms, medication use and airway calibre (90). It contains seven items assessing the previous seven days on a seven point scale; five symptom questions, a question assessing medication use and an FEV1 measurement (90). It has been validated for use in children aged six to 10 years of age when administered by an interviewer (91). Cut off points have been determined dividing patients into 'well controlled' and 'inadequately controlled' asthma (92).

The Asthma Therapy Assessment Questionnaire (ATAQ)

The ATAQ was developed to assist clinicians identify children and adolescents with asthma who are at risk of adverse outcomes. The questionnaire contains 20 items assessing symptom control, behaviours, attitude barriers, self-efficacy barriers and communication gaps. This questionnaire addresses issues that are not addressed by other questionnaires such as the ability to take medications. It is useful in identifying those in need of further management or medical attention in five to 17 year olds (93).

Breathmobile Case Identification Survey

The Breathmobile Case Identification Survey is a one-page survey that was developed as part of the Breathmobile programme. The Breathmobile programme aimed to detect cases of asthma in a large population aged less than 18 years of age (mean age 8.9) in reaction to a large increase in the incidence and prevalence of asthma (94). It contained seven questions to be completed by caregivers and was shown to have a sensitivity of 86.5% and a specificity of 83.6 per cent. It had a misclassification rate of 14.3 per cent. An inexpensive incentive to complete the questionnaire increased return rates from 35.3 to 65.0 per cent (95). It has been validated in a low socioeconomic, urban population to identify uncontrolled asthma with moderate to severe disease activity (94).

The Childhood Asthma Control Test

The Childhood Asthma Control Test was developed due to a lack of measure of asthma control in children under 12 years of age. Its development has provided a tool that provides the clinician with a quick overall picture of asthma control and aids communication between clinicians, caregivers and the child (96). It is a seven-item questionnaire with four child completed and three caregiver completed items. It is reliable and accurate in assessing asthma control in four to 11 year olds (96). A score of 19 or less indicates inadequate control (96) with a score of 12 showing the poorest control and increased risk of poor outcome (97).

Childhood Asthma Questionnaire (CAQ)

There are three versions of the CAQ; A, B and C which were each developed for a different age range (four to seven years, eight to 11 years and 12 to 16 years respectively). The questionnaires contain 14, 22 and 41 items respectively (98). These questionnaires assess the QoL of the respondent due to their diagnosis of asthma using pictorial scales of either increasing size bars or cartoon faces. The younger responders completed the questionnaires with the help of their parents (98, 99). The questionnaires had good test-test repeatability

and were shown to be good estimates of disease stability. Internal consistency of the questionnaires varied but was higher when used in older children (98).

Chronic Cough QoL Questionnaires

The Cough Specific QoL Questionnaire was originally developed in Brisbane due to the lack of a cough specific QoL measure in the paediatric population. It assesses parental feelings and worries towards their child's chronic chough (100). It is a 27-item questionnaire assessing physical, social and psychological domains of QoL. It is a reliable and valid outcome measure and is sensitive to change over time when assessing children under 14 years (100). Later, a 16-item child completed questionnaire was developed. That has been shown to be valid and reliable in seven to 17 year olds and is sensitive to change over time (101).

The International Study of Wheezing Infants Questionnaire (EISL-WQ)

The International Study of Wheezing Infants Questionnaire (EISL-WQ) contains 45 questions assessing demographics and risk factors for wheeze, respiratory symptoms, physician diagnosis of asthma and medication use. It was designed to be completed by parents (102). It has been validated in Spanish and Portuguese and shows agreement with clinical examination in infants aged 12 to 15 months old (103, 104). A shortened version containing 21 questions has been validated in Portuguese showing good reliability, reproducibility and internal consistency for children aged 12 to 36 months of age (102).

ISAAC Questionnaire

The ISAAC questionnaire was developed by experts using a consensus process at a workshop using previously validated questions. It contained 21 items assessing prevalence and severity of wheeze, eczema and rhinitis (105). The questionnaire was self-reported in 13 and 14 year olds, and parental reported in six to seven year olds (105). It has been successfully used in an international study (ISAAC) (105). The questionnaire has been produced in a variety of languages (105, 106) and an international video version of the questionnaire has also been validated (107).

The Leicester Respiratory Cohorts Questionnaire

The Leicester Respiratory Cohorts Questionnaire contains 51 questions assessing three domains; symptoms, environment and home factors, and family and social status (108, 109). The symptom domain assesses the nature of the respiratory history and specifically assesses wheeze, cough and doctor diagnosed asthma. If wheeze is reported an additional 10 questions assess the phenotype of the wheeze (109). It has been shown to have excellent repeatability (110).

The Mechanisms of Development of Allergy (MeDALL) Core Questionnaire

The MeDALL Core Questionnaire was developed to harmonize questionnaires that asses asthma and allergy used in birth cohort studies to allow comparison between studies (111). It is now in use in 11 birth cohort studies in eight different languages. It was developed by members of eight European birth cohorts to asses 14 to 18 year olds and their parents. It consists of eight sections with a total of 66 core questions with an additional 19 optional questions (111). There has been no formal validation of this questionnaire and it may not be suitable to younger populations.

The Paediatric Cough Questionnaire

The Paediatric Cough Questionnaire is a measure of the QoL of a child with a chronic cough completed by the caregiver. It contains five questions assessing frequency; sleep disturbance (of parent and child), severity, and 'bothersomeness' to the child on a 6-point Likert scale. It showed good test-test repeatability, internal consistency and validity (112).

Paediatric Asthma Quality of Life Questionnaire (PAQLQ)

The PAQLQ was designed to examine the burden of asthma experienced by children with the disease. It aimed to assess areas of function and physical and emotional wellbeing. It consisted of 23 items split into three domains; activity limitation, symptoms, and emotional function. The unique aspect of this is that the questionnaire is individualised; at the first time of completing the questionnaire the participants asked to identify three activities they enjoy doing. Questions are assessed on a 7-point Linkert scale (113). The PAQLQ was able to detect change in health status and differentiate from stable patients aged seven to 17 years. It was reproducible in stable patients (113). A pictorial version has been developed, however this had only been validated in those over five years of age (114).

Test for Respiratory and Asthma Control in Kids (TRACK)

The TRACK questionnaire is a five item questionnaire completed by parents which assesses symptoms, their frequency and their effect on activity, waking at night and medication use (115). It has been validated and is a reliable, easy to administer questionnaire. Cut off scores identifying those with respiratory control problems and a minimally important difference score have been determined (115, 116). The questionnaire is responsive to changes in symptoms over time (117).

Conclusion

There are many paediatric respiratory questionnaires some of which have been validated for use in the paediatric population. However the majority of these are not validated in those less than five years of age. The majority of questionnaires are disease specific or assess QoL alone. The LRSQ questionnaire used in the LBBS is the only questionnaire that is validated in the preschool population, which assesses a range of respiratory symptoms under various conditions and combines the assessment of the QoL of the child and their caregiver with a respiratory questionnaire.

1.5.2-The Liverpool Respiratory Symptom Questionnaire Development of the LRSQ

The Liverpool LRSQ was designed by respiratory consultants at the Liverpool Women's Hospital and Alder Hey Children's Hospital. It was designed using common reoccurring features and concepts from standard questionnaires already in use in older populations. It assesses wheeze, cough and breathlessness during both the day and night, and the impact of these symptoms on both the child's and family's quality of life over the previous three months. It is the assessment of the impact of these symptoms which makes the questionnaire unique. A 5-point Linkert scale is used ranging from no symptoms to symptoms every day (1). The original questionnaire contained 32 questions. A cohort of 242 babies was used to assess practicality, response rate, reliability, validity and criterion in comparison to expert opinion. (1).

Response rates to the questionnaire sent via post in an earlier study were 64% for the first questionnaire and 56% for the repeat questionnaire. Questionnaires completed in the outpatients department produced a response rate of 100%. Overall this led to 114 pairs of questionnaires for further analysis (1). The LRSQ showed moderate short term repeatability (Kappa score of 0.4 or more) in all bar one question which showed fair repeatability, and good internal consistency in all domains (1). Factor analysis showed four questions were responsible for a large proportion of the variance in responses so these were removed from the original questionnaire which was then rearranged into eight domains (table 2) (1). The questionnaire showed agreement with respiratory expert opinion. Those with asthma scored significantly higher in all domains. ROC analysis showed that all domains had a good sensitivity for identifying asthma patients, however some domains showed unacceptable specificity (1).

Overall the original development of the LRSQ has produced a practical, acceptable respiratory symptom questionnaire for the use in under five year olds, containing 28 questions over eight domains which produced good response rates and was valid, with good internal consistency (1). The questionnaire is shown in figures 1-4.

Copyright Liverpool Womens Hospital

Name of Child

Study Number

1. This first question refers to at any time in your child's life:

Has your child ever had wheezing (whistling noise	Yes	No	
coming from the chest) at any time in the past?			

2. The next questions are specifically aimed at the last three months:

A) During the day (when awake) in the <u>last three months</u> :							
i) My child h	i) My child has had wheezing (whistling noise coming from the chest):						
Every day	most days	some days	a few days	not at all			
ii) My child ł	has had a cough:						
Every day	most days	some days	a few days	not at all			
iii) My child	has had a rattly	chest:					
Every day	most days	some days	a few days	not at all			
$\mathrm{iv})$ My child	has been short o	of breath:					
Every day	most days	some days	a few days	not at all			
1							

B) During the	B) During the night (when asleep) in the <u>last three months</u> :						
i) My child ha	i) My child has had wheezing (whistling noise coming from the chest):						
Every night	most nights	some nights	a few nights	not at all			
ii) My child ha	s had a cough:						
Every night	most nights	some nights	a few nights	not at all			
iii) My child ha	iii) My child has had a rattly chest:						
Every night	most nights	some nights	a few nights	not at all			
iv) My child ha	iv) My child has been short of breath:						
Every night	most nights	some nights	a few nights	not at all			
v) My child has snored :							
Every night	most nights	some nights	a few nights	not at all			

Figure 1: Page 1 of the Liverpool Respiratory Symptom Questionnaire

C) How	C) How many colds has your child had in the last three months:					
None	one	two	three	more the three	an alway. has a cold	5
If the an section I	swer to tl):	he above q	uestion is	' <i>none</i> ' con	tinue to ques	tions in
When m	y child h	as had a C	OLD in th	e <u>last thre</u>	e months:	
i) My cł	hild has ha	d wheezing	g (whistling	g noise cor	ning from the	chest):
Every co	ld m	ost colds	some c	olds	a few colds	not at all with colds
ii) My cł	nild has ha	id a cough:				
Every co	ld m	ost colds	some c	olds	a few colds	not at all with colds
iii) My c	hild has h	ad a rattly o	chest:			
Every co	ld m	ost colds	some c	colds	a few colds	not at all with colds
iv) My c	hild has b	een short of	f breath:			
Every co	ld m	ost colds	some c	colds	a few colds	not at all with colds

D) When my child does NOT have a COLD, in the <u>last three months</u> :							
i) My child has had wheezing (whistling noise coming from the chest):							
Every day	most days	some days	a few days	not at all			
ii) My child ha	ii) My child has had a cough:						
Every day	most days	some days	a few days	not at all			
iii) My child ha	as had a rattly ch	iest:					
Every day	most days	some days	a few days	not at all			
iv) My child has been short of breath:							
Every day	most days	some days	a few days	not at all			

Figure 2: Page 2 of the Liverpool Respiratory Symptom Questionnaire

E) When my child has been MORE ACTIVE (e.g. crawling, walking or when excited) in the <u>last three months</u> :							
	i) My child has had wheezing (whistling noise coming from the chest):						
	Every day	most days	a few days	not at all			
	ii) My child has coughed:						
	Every day	most days	some days	a few days	not at all		
iii) My child has had a rattly chest:							
	Every day	most days	some days	a few days	not at all		
iv) My child has been short of breath:							
	Every day	most days	some days	a few days	not at all		

F) These next three questions are about other problems your child may have had. Over the <u>last three months</u> :							
i) My child has had noisy breathing that does not seem to come from the chest :							
Every day	most days	some days	a few days	not at all			
ii) My child l	ii) My child has had fast breathing:						
Every day	most days	some days	a few days	not at all			
iii) My child has had noisy breathing that appears to come from the throat or back of the throat:							
Every day	most days	some days	a few days	not at all			

Figure 3: Page 3 of the Liverpool Respiratory Symptom Questionnaire
G) The next four questions are on how your child's chest symptoms actually affect HIM or HER over the last three months: i) My child's chest symptoms have affected my child's feeding or eating: Every day not at all most days some days a few days ii) My child's chest symptoms have woken up my child: Every night some nights a few nights not at all most nights iii) My child's chest symptoms have reduced my child's activity: Every day most days some days a few days not at all iv) My child's chest symptoms have made my child unusually tired: Every day most days some days a few days not at all

H) The next four questions are on how your child's chest symptoms actually affect YOU and YOUR family's life the last three months: i) My child's chest symptoms have limited my activities: Every day most days some days a few days not at all ii) My child's chest symptoms have resulted in adjustments being made to our family life: Every day most days some days a few days not at all iii) My child's chest symptoms have disturbed our sleep:

some nights

some days

a few nights

a few days

not at all

not at all

iv) I have been worried about my child's chest symptoms:

most nights

most days

Every night

Every day

Figure 4: Page 4 of the Liverpool Respiratory Symptom Questionnaire

Previous Use of the LRSQ

Since its development the LRSQ has been further validated and has become a popular questionnaire in assessing respiratory symptoms in under five year olds.

The LRSQ can differentiate between those with CF and those without and between active and stable CF disease in both preschool infants and six to 12 year olds. It correlates well with other measures of respiratory disease. In the same study, analysis of the questionnaires showed all eight domains had good internal consistency in the preschool aged group. Removal of questions about snoring improved the internal consistency further (16).

The LRSQ has been increasingly used in studies that are independent of the original research team that are described below. The ADEM study used the LRSQ in the development of a new asthma diagnostic technique (118). The LRSQ was used as an outcome measure in premature infants without chronic lung disease to measure of lung function (119). Finally it was used in a study in preschool infants with wheeze to assess if the response to corticosteroids can be predicted by inflammatory response measures or by measuring the change in symptoms and lung function (120).

The LRSQ has been used in the 'Prevent Passive Smoke Exposure study' which assessed the benefit of motivational interviewing in preventing passive smoke exposure in those with a high asthma risk (121). It was used as a disease measure in a study assessing the impact of primary prevention programs on risk taking behaviour for allergic disease (122). A cross-sectional study showing preterm birth was associated with increased wheeze used the LRSQ to assess the respiratory symptoms in under five year olds (25, 46). It has also been used successfully to show that RSV associated wheeze was more likely to produce allergic sensitisation and lead to increase symptoms and steroid use compared to RSV Bronchiolitis (123).

Five studies investigating maternal and birth risk factors have used the LRSQ (124-128). Two studies assessing impact of maternal asthma control in pregnancy on preschool respiratory symptoms used the LRSQ (124, 125). They found that firstly, optimised control of maternal asthma may reduce the risk of infant bronchiolitis and possibly subsequent asthma in childhood (125). Secondly that exposure to maternal asthma may alter the infant's peripheral blood DNA methylation profile which is a possible risk factor for childhood asthma development (124) The LRSQ has been used to investigate the maternal acetate levels in pregnancy and showed that a higher levels and a high fibre diet may be associated with reduced preschool airway disease (127). Similarly the effects of amniotic protease on

childhood respiratory symptoms was assessed using the LRSQ but there was no association found (128). Lastly it was used to investigate the management of very early preterm premature rupture of membranes with amnion transfusion, but again no significant benefit was found with regards infant respiratory symptoms (126).

The LRSQ was used to assess the respiratory outcomes in children with neurodisability. It successfully concluded that children with neurodisability have higher respiratory symptom scores (17, 129). In infants who have undergone lobectomy due to congenital thoracic malformations the LRSQ was able to differentiate those who have ever had asthma compared to those who have not (130).

The LRSQ has been used in a variety of studies alongside the ISAAC questionnaire as a popular measure of baseline atopy. This combination of questionnaires has been used in studies assessing the use of skin prick testing and nitric oxide production measurement in the investigation of peanut allergies (131, 132). Likewise it has been used in a study to investigate the link between environmental smoke exposure, interleukin 4 and 13 gene polymorphisms and risk of infant wheeze. It was found those exposed to ETS with the interleukin 4 gene polymorphisms conferred at a tenfold increased risk of wheeze (133).

1.6- Birth Cohort Studies

Birth cohort studies have been commonly used to investigate risk factors for paediatric respiratory disease and to map the natural history of respiratory disease in children. A literature review of Birth Cohort Studies was performed to investigate their findings and study designs. The search terms used to identify cohort studies performed in both Britain and internationally, which follow both national and regional populations are shown in appendix 1. An in depth analysis of their main findings and their study design was performed and is shown in appendix 2; only those cohort studies with findings relevant to preschool respiratory disease and its risk factors, or those with novel and interesting study designs are discussed. During this literature review four large respiratory studies that did not follow a 'birth cohort study' design were identified. A detailed analysis of their study design and key findings is also included in appendix 3. The remaining cohort studies found in the literature review but that are not discussed in detail are shown in appendix 4. Table 2 shows a summary of the study design of the relevant cohort studies. A summary of recruitment strategies and methods to reduce attrition in previous cohort studies is given below.

Table 2: Birth cohort s	tudies identified b	by literature r	eview
-------------------------	---------------------	-----------------	-------

Cohort	Year	Cohort Size	Recruitment Rate	Attrition
1946 British Cohort Study (134)	1946	13,600	82%	12.6% (4 years) 19.9% (15 years)
1958 British Cohort Study (135)	1958	17,400	99%	0.7% (7 years)
1970 British Cohort Study (136)	1970	16,600	96%	21.1% (5 years)
The Millennium Cohort Study (137)	2000	18,800	88%	17.2% (3 years) 17.8 (5 years)
Aberdeen Children of the 1950's Cohort Study (138)	1950	12,000	85%	36% (50 years)
Avon Children of the 90's Study (139)	1991	14,541	95%	2.6% (2 years) 15.6% (5 years)
Born in Bradford (49)	2007	13,776	64%	25% (1 year) 30% (2 years)
Isle of Man Cohort Study (140)	1992	1,314	95%	21.2% (15 years)
The Newcastle Thousand Family Study (141)	1947	1,146	84%	26.1% (5 years)
The Southampton Women's Cohort (142)	1998	12,583	75%	5% (1 year) 7% (2 years)
Manchester Asthma and Allergy Study (143)	1995	1,500	41%	6.9% (1 year) 8.3% (3 years)
The Prevention and Incidence of Asthma and Mite Allergy (144)	1996	4,000	50%	8% (8 years)
The Tucson Children's Respiratory Study (45)	1980	1,200	78%	13% (5 years)

1.6.1- Summary of Recruitment in Birth Cohort Studies

Recruitment strategies in previous birth cohort studies have led to recruitment rates between 41% and 99% of the eligible population (135, 143); regional cohort studies report recruitment rates between 95% and 41% (139, 143). Recruitment has been from GP practices (142), prenatal clinics (144), during postnatal care (134-136) or using electronic records after

birth (137, 138). Recruitment has been performed by either midwives, other members of the study team (136, 141, 143) or by participants self-registering (142). The length of recruitment has varied from one week in some studies (134-136) to over a year in others (72, 74, 75); the use of all year round recruitment has reduced the seasonal variability in characteristics of birth and diseases. The use of advertisement before the study in the Isle of Man cohort study lead to very high participation rates (140). The Avon Children of the 90's study and the Isle of Man Cohort study were the two regional cohort studies with the most successful recruitment strategies. They both recruited for over a year using midwives to recruit mothers during their prenatal care (139, 140).

The profile of the infants and their families recruited into the cohort studies were often not representative of the population eligible for recruitment particularly in smaller regional birth cohort studies. For example in the Avon Children of the 90's study (ALSPAC), mothers were more likely to be white, own a home and car, be married, have a higher socioeconomic status and live in an overcrowded home that the population eligible for recruitment (139). The BiB cohort contained older, more educated mothers (49) and the mothers in the Prevention and Incidence of Asthma and Mite Allergy study (PIAMA) were more highly educated and more likely to breastfeed than the population of the Netherlands (144). Likewise infants of the minor ethnicities were under represented in the 1958 British Birth cohort (135).

1.6.2- Summary of Attrition in Birth Cohort Studies

It is clear from previous cohort studies that attrition is an unavoidable phenomenon in cohort studies due to a variety of causes. These include death, withdrawal of consent, emigration and loss of contact. The 1958 British cohort had the lowest attrition rate; it was still able to trace 99.3% of its participants at 7 years (135). Of the smaller regional cohorts the PIAMA study had the lowest attrition rates with only 8% of the cohort being lost over the first 8 years of the study (144). The BiB 1000 sub-study had the highest rates of attrition with 30% of the cohort being lost over the first 2 years (49).

Measures have been taken to increase response rates and reduce attrition in previous cohort studies. Methods used are regular contact with participants through the use of newsletters, birthday cards (49), phone calls (142) and study websites (49). School registers and links to medical or government records have been used to reduce the proportion of participants who are lost due to a change in their contact details (134-136). It is clear that studies that have more resources (such as access to school/medical/government records and a team designated to improving follow up) had reduced attrition rates in comparison to smaller studies with fewer resources.

1.7-Merseyside Health Profile and Respiratory Studies

1.7.1-Merseyside Health Profile

In 2015 Liverpool was ranked the 4th most deprived city in the UK with 8.7% of its population being in the most deprived 1% of the UK population (145). Over 32% of children in Liverpool live in low income families (13). Life expectancy is lower than the national average for both males and females and this is most pronounced in more deprived areas (13). Liverpool has high levels of preschool respiratory disease compared to the average in the UK. The Chief Medical Officer's report of 2012 stated that Liverpool has a higher rate of hospital admissions due to bronchiolitis and the duration of stay in hospital is longer than the average in England (15). The hospital admission rate due to LRTI's is similar to that of the average in England however the length of stay in hospital is longer than the average in England (15). The prevalence of asthma in Liverpool is increasing and the number of emergency admissions due to asthma in both children and adults is higher (304.7 compared to 216.1 per 10000 patients), and the average duration of stay in hospital is longer again indicating inadequate control (14, 146). The rate of death due to respiratory conditions at all ages is higher in Liverpool than the average in England (14). Liverpool has high rates of smoking compared to the average in England and smoking related deaths are more common, accounting for 928 deaths a year (13).

Children in Liverpool tend to be less educated than the average children in England, and they are more likely not to be in education at 16-18 years old and more likely to be living in poverty at 16 years of age (146). Liverpool has higher rates of child Accident and Emergency (A&E) attendances compared to England, higher levels of children with decayed teeth and a higher rate of admissions due to dental caries (146). In school year 6, 23% of children are classified as obese, which is higher than the national average (13).

In 2015 mothers in Liverpool were more likely to be teenagers, less likely to initiated breastfeeding, less likely to vaccinate their children and are more likely to be smoking at the time of delivery (16.1% compared to 10.6%) than the average mother in England (13).

All this together makes the Liverpool population a good cohort for respiratory epidemiological studies due to the higher prevalence of risk factors for respiratory disease that they express.

1.7.2-Merseyside Respiratory Studies

In 2006, 61.4% of children in Liverpool lived in household with at least one smoker, compared to 53% in the rest of the UK. This exposure to tobacco smoke was shown to give the children

a higher average saliva cotinine concentration than the average in England (1.6+/-0.4 ng/ml compared to 0.5ng/ml) (147). Risk factors for an increased ETS exposure was maternal (rather than paternal) smoking, age less than seven years of age, being male and a lower socioeconomic status (147).

Two studies investigated the effect of dust air pollution on respiratory symptoms in school children in Liverpool. Liverpool was an ideal place for this study due to the Bootle dock yard situated within a large residential area (36, 148). Studies found that exposure to the coal dust increased risk of respiratory symptoms and absence from school. Those expose to coal dust were also more likely to be low birth weight and their parents were more likely to be unemployed and smoke (34). It was shown that living within two kilometres of the docks made the children twice as likely have excess cough, breathlessness, be absent form school and have a doctor diagnosis of asthma compared to those over two kilometres from the docks (148).

Two cross-sectional studies distributed questionnaires based on Clifford et al. (149) across 15 schools on Merseyside to investigate the effect of parental asthma on premature birth and subsequent respiratory disease (150, 151). It was found that maternal smoking and maternal asthma were risk factors for preterm birth which was associated with respiratory disease in school children (150).

Between 1991 and 1998 maternal asthma and hospital admissions due to asthma had nearly doubled in Liverpool. Wheezing had increased in school children but the diagnosis of asthma (defined by the triad of symptoms of cough, SoB and wheeze) had remained stable. Higher deprivation, male sex, allergy, obesity and parental asthma were all shown to be associated with higher risk of asthma (151).

A study, that assessed lung infection in adults from Liverpool, showed that opportunistic mycobacterial infections accounted for 3.9% of pulmonary infections. This was higher than a similar study in Wales at a similar time (2.9%). The high smoking rates in Liverpool may have accounted for this difference (152).

1.8-Conclusion

The LBBS is a longitudinal birth cohort study now running into its 6th year. It uses the LRSQ, a popular, validated measure of respiratory disease in preschool children, to assess the respiratory symptoms of the cohort biannually from four months to five years. The study aims to map the natural history of respiratory disease in Merseyside. Birth cohort studies have been used to assess respiratory disease for many years and have produced many novel

findings with regards paediatric respiratory disease. Common risk factors for respiratory disease discussed in the literature are parental smoking pre and postnatally, air pollution, premature birth, low birth weight and deprivation. Breastfeeding has been shown to be protective of respiratory disease in early life. Merseyside is an ideal area for a respiratory cohort study due to its high prevalence of risk factors of respiratory disease in children, including high maternal smoking, low breastfeeding rates and high air pollution. Liverpool has a high rate of hospitalisation of asthma and bronchiolitis compared to the population of the UK. A questionnaire was chosen to assess respiratory disease rather than formal measures of lung function as they are unreliable in the preschool population. The LRSQ has been chosen as a measure of respiratory disease in the current study due to its previous development and validation on the preschool population of Liverpool. It also incorporates assessment of the QoL of infants and their families into the questionnaires. This is an important factor to consider when assessing respiratory disease as shown by previous studies. Attrition is an inevitable phenomenon in cohort studies for a variety of reasons including death, emigration, loss of interest and loss of contact. A good relationship with the cohort is imperative for reducing attrition; this has previously been done in some studies with the use of birthday cards, newsletters and social media. Well supported studies with team members dedicated to reducing attrition by contacting participants, and studies with access to school or hospital records have the lowest attrition rates.

Chapter 2 - Methods

There are two distinct phases in the methodology of the LBBS; the methods of the study design and the methods of analysis of the data of the study. The study design and development were conducted by previous investigators and published in their MPhil theses (2, 3). A short summary of this is presented in section one of this chapter. The second phase of work was the analysis of the LBBS conducted by the author. The methods used to analysis the data are described in detail in section two of this chapter.

The LBBS is 'a prospective, longitudinal birth cohort study' that uses the LRSQ 'to conduct biannual assessment of the respiratory symptoms in preschool children born in Liverpool from birth until the age of five years'. The design of the LBBS and questionnaire development began in 2012 and has undergone five phases outlined in table 3.

Phase		Investigator (s)	Dates
1	Protocol Completion and Ethical Approval	Miss Rosanna Pickles Miss Bethan Griffith Dr Calum Semple Prof Paul McNamara Prof Kevin Southern Prof Ben Shaw	May 2012
2	Questionnaire Development Redevelopment	Miss Rosanna Pickles Miss Bethan Griffith Mr Sanjay Patel	September 2012- January 2013 June 2015
3	Recruitment	Miss Rosanna Pickles Miss Bethan Griffith Mr Sanjay Patel Miss Jennifer Barclay	January 2013 - November 2014
4	Distribution	Miss Jennifer Barclay	September 2014 - August 2015
5	Analysis	Mr Joshua Stead	September 2016 – June 2017

Table 3: Phases of the LBBS

2.1-Section 1. LBBS Study Design and Development Summary

2.1.1-Aims

The aims of this section of the study were to:

- Develop the study protocol and obtain ethical approval for the study.
- Develop study materials and the questionnaire to be used in the study using expert knowledge, Public Patient Involvement (PPI) sessions and pilot studies.
- To determine and initiate a method of creating an automated online system for the distribution and completion of questionnaires.

2.1.2-Protocol Development

Protocol development finished and ethical approval was granted in May 2012 (Research Ethics Committee Ref: 12/EM/1904). The study protocol was developed by Miss Rosanna Pickles, Dr MG (Calum) Semple, Prof Paul S McNamara, Prof Kevin W Southern and Professor NJ (Ben) Shaw. The study protocol is shown in appendix 6.

2.1.3-Questionnaire and Study Material Development

Rossana Pickles and Bethan Griffith developed the questionnaires through the use of PPI sessions based at the LWH. This work has been described previously by MPhil students in their own work (2, 3). A summary of this is given in appendix 7. The final design of the questionnaire and other materials used in the LBBS are shown in appendix 10- 13.

2.1.4-Recruitment

The recruitment phase was carried out by Rossana Pickles, Bethan Griffith, Sanjay Patel and Jennifer Barclay and occurred between January 2013 and November 2014. Eligible Infants were recruited after birth from the LWH. Inclusion and exclusion criteria are shown in figure 5. Initially if interested in the study, parents (usually mothers)

would complete a postcard with

INCLUSION

- Born at LWH
- Parents lived in postcodes L1-L38
- Spoke sufficient English to consent to the study and complete questionnaire

EXCLUSION

• Infants in local authority care

Figure 5: Inclusion and Exclusion criteria in the LBBS

• Infants with futile prognosis as identified by clinical team

contact details and leave it with the research team after their discharge from hospital. Later parents would complete an app-based form on a tablet offered by members of the research team. The research team or automated email process would then contact the participants when their child reached four months of age with the consent form and questionnaire one via post or email respectively. A four-week pilot study of recruitment of infants into the LBBS was performed by Rosanna Pickles and Bethan Griffith. The results of this pilot study are shown in appendix 8. Discussions within the research team took into account the results of the pilot study and it was decided that four days a week (Monday, Tuesday, Thursday and Friday) would be a more suitable strategy. This was trialled for one week producing response rates of 69% and was therefore deemed effective (2, 3).

2.1.5- Questionnaire Distribution

Questionnaire distribution began in May 2013, is ongoing, and due to finish in October 2019. A consent form and questionnaire one was sent to the cohort when the infant was four months of age. Infants are assessed every six months until 10 questionnaires have been completed at four years and 10 months of age. The cohort has currently received four questionnaires. The PPI sessions showed that the questionnaire would be most acceptable to the public if available online. Therefore the development of an online questionnaire that was automatically sent to participants was required; this work was done by Bethan Griffith and Roseanna Pickles and is shown in appendix 9. It was decided that questionnaires would be completed using Adobe FormsCentral[®] (AFC[®]) (Adobe Systems Inc. California, USA). The email platform that would be used to send the questionnaire link to the participants was MailChimp[®] (The Rocket Science Group, Atlanta, USA). In June 2015 AFC[®] was suspended to allow development as a future premium product with charges. This forced the study team to change to use JotForm[®] (JotForm Inc. San Francisco, USA) for questionnaire completion and response storage.

Modification of the Emailing Timeline to Reduce Attrition in the LBBS

During the early analysis of the data it was apparent that a significant proportion of the drop out in the LBBS was due to participants not being invited back into the study if they did not complete the previous questionnaire. This was because the study protocol intended to analysis the longitudinal data of the LBBS using multivariate regressions that do not allow for missing data. The use of multi-level mixed effect models allowed participants to miss questionnaires and still be included in the analysis. Therefore it was decided to send participants the next questionnaire regardless of them completing the previous questionnaire providing they had not intentionally left the study. This was done by modifying the automation instructions in MailChimp[®] so that sending the next questionnaire to a participant was no longer dependant on opening the previous questionnaire. This modification was completed by the author. Participants could still intentionally leave the study by unsubscribing from the emailing list themselves and by emailing the study team directly.

2.1.6- Batch Testing

Due to ethical approval requirements in 2012, the study needed a method of identifying deceased infants throughout the study so as to avoid contact with bereaved parents that could cause distress. This process was performed by the IT department at Alder Hey Children's Hospital through batch testing the list of infants in the study on a monthly basis. Transfer of information between institutions for batch testing was done using encrypted files.

2.2-Section 2. Methods Used in the Statistical Analysis of the LBBS

2.2.1-Aims

The aims of this phase of the study were to;

- To create a profile of the infants and mothers enrolled in the LBBS and make a comparison with the population of Liverpool and those born in the LWH.
- To describe respiratory symptoms of preschool children using the LRSQ from birth until 22 months of age.
- To determine the differences in LRSQ scores in the different populations at each questionnaire time point.
- To determine the change in the LRSQ scores over time and the variability in this change between different population groups in the LBBS.
- To validate the LRSQ in a longitudinal study.

These aims were achieved by analysing each of the following individually; 1. The response rates and drop out from the LBBS, 2. The profile of the infants their mothers and families, 3. The LBBS data at each time point individually (cross-sectional), 4. The LBBS data from all time points together (longitudinal) and 5. the internal validity of the questionnaire.

2.2.2- Data Entry and Coding

Questionnaire responses were downloaded from JotForm (and previously AFC) to Microsoft Excel (Microsoft Excel, 2016, Washington, United States). At this point data was in a noncoded form with responses matching that of the questionnaire format. Data was coded using the find and replace function to give each variable response a numerical number which could be used in the statistical packages used for data analysis in the LBBS. The codes used in the LBBS are shown in appendix 15. The table in the appendix shows the variable code and the response code used in the statistical packages. Birth weights that were given in 'pounds and ounces' were converted to grams for the purpose of this analysis. Postcodes were converted to IMD decile as described in section 2.2.6 of this thesis. Descriptions of underlying health conditions were converted to health conditions were possible by the author. Where this was not possible the description given by the mother was given in the results section of this thesis.

2.2.3- Statistical Packages Used During Analysis

Social Sciences (SPSS) version 22 (IBM Corp. Released 2013. IBM SPSS Statistics for Windows, Version 22.0. Armonk, NY: IBM Corp.) and STATA version 14 (StataCorp. 2015. Stata Statistical Software: Release 14. College Station, TX: StataCorp LP.) were used during the analysis of the study. SPSS was used during analysis of recruitment and response rates, the infant and mother profiles in the LBBS and the cross-sectional analysis of the LBBS. STATA was used for the longitudinal analysis of the LBBS and the validation of the LRSQ.

2.2.4-Recruitment

Recruitment and response rates were determined from denominator data given by Professor Ben Shaw from LWH for all the eligible births in the LWH during the recruitment period. Numerator data were the number of mothers considered to be interested in the study who returned a post card to the study team, signed up to the study via the quick response code or via the tablet used by the research team. Only those mothers who completed the consent form and responded to questionnaire one when their infant was four months of age were labelled as enrolled in the study. Descriptive statistics were used to describe the recruitment and response rates in the LBBS.

2.2.5-Response Rates and Attrition

Response rates to each questionnaire were described by descriptive statistics. Those who intentionally removed themselves from the mailing list or emailed the recruitment team to be removed from the study were described as having left the study. MailChimp[®] gave statistics on emails that had 'bounced' indicating an incorrect email and these participants were subsequently described as email bounces. Babies that had died since mothers declared their interest to the study or were enrolled were identified by batch tracing using records at Alder Hey Children's Hospital on a monthly basis. Participants that did not respond to the questionnaire were declared lost to follow up.

The demographics and exposures of the drop out population and remaining population at each questionnaire was described using summary statistics to evaluate the change in the cohort characteristics as attrition occurred. This was used to account for the uneven drop out from exposure groups throughout the study.

2.2.6-Profiles of Mothers in the LBBS

The profile of the mothers in the LBBS will be described by descriptive and summary statistics. Where appropriate the profile of mothers in the LBBS will be compared to the mothers of all the eligible births from the LWH during the recruitment period, and that of the population of Liverpool and England using χ^2 analysis. A significant result was indicated by p≤0.05. The profile of the mothers of the eligible births in LWH was obtained from Professor Ben Shaw at the LWH. Data regarding maternal age group and education was obtained from the Office of National Statistics(153). Deprivation data was obtained from the UK Government Official Statistics (154). The postcodes of all the eligible births from the LWH and the postcodes of the mothers enrolled in the LBBS were converted to IMD decile using The Department of Communities and Local Governments Postcode Converter (155). Participants in IMD decile onehad the highest level of deprivation. Cigarette smoking and breastfeeding rates in Liverpool and England were obtained from the NHS Statistical Release for a similar time period of recruitment (12, 156).

2.2.7-Profile of Infants in the LBBS

The profile of the infants enrolled in the LBBS will be described by descriptive statistics. Where appropriate the profile of the infants in the LBBS will be compared to that of all the eligible births from the LWH during the recruitment period, and that of the population of Liverpool and England using χ^2 analysis. A significant result was indicated by p<0.05. The profile of the eligible infants born in the LWH was obtained from Professor Ben Shaw at the LWH. Data regarding sex and ethnicity was obtained from the Official Labour Market National Statistics (157). Data regarding multiple births in England was obtained from the Office of National Statistics (153). Data regarding gestational age and birth weight was obtained from the NHS Digital Monthly Maternity Services Statistics (158).

2.2.8-Cross-sectional Analysis

Cross-sectional analysis in the LBBS study was split into five sections and was carried out using data from each of the four questionnaires individually; 1. Risk factor exposure analysis, 2. LRSQ total and domain score analysis, 3. Symptom prevalence analysis, 4 .Health care attendance analysis and 5. Analysis of the impact on the Child's and their Family's QoL. Demographics, Risk Factors and Exposures Used During Cross-sectional Analysis The factors used to group the cohort during analysis are shown in figure 6. Due to drop out in the study, and the particularly low levels of non-white ethnicities the decision was made to analyse ethnicity using two groups; white and other ethnicities.

Sex (Male/Female)
Birth number (Single/Multiple)
Infant Ethnicity (White/Other)
IMD decile (1-10)
Maternal age at birth (<20/ 20-24/ 25-29/ 30-34/ 35-39/ 40-44/ ≥45)
Highest maternal qualification attained, (Degree or Higher/ Higher Education/ A-Level/ GCSE/ Other/ None)
Preterm birth (Preterm/Term)
Low birth weight (Low Birth Weight/ Normal Birth Weight)
Breastfeeding (Breastfed/ Not Breastfed)
Breastfeeding time up to 4 months (Not Breastfed/ <1 Month/ ≥1- <4 Months/ ≤4 months)
FH of atopy (Yes/ No)
Other household children (Yes/ No)
Number of other household children (1-4, 5+)
Sharing a Bedroom (Yes/ No)
Sharing a Bedroom with Who (Parents/ another Child/ Other)

Figure 6: Risk factors and exposures assessed in the LBBS

Exposure Analysis

Descriptive statistics of the infant's exposures to risk factors in the previous three months were presented. Exposures include nursery attendance, other household children, sharing a bedroom and ETS exposure. Households with smoking were compared to the prevalence of smoking in Liverpool in 2012 and 2015 (11, 157) using χ^2 analysis; a significant result was indicated by p≤0.05. A description of the co-morbid health conditions in the LBBS cohort was given.

LRSQ Scores

Descriptive statistics of the total and domain LRSQ scores were presented. The Mann-Whitney U test and Kruskal-Wallis H Test were used to compare questionnaire scores in exposed groups and non-exposed groups; a significant result was indicated by $p \le 0.05$. The Mann-Whitney U test was used to compare scores of variables with only two groups whereas

the Kruskal-Wallis H test was used to compare variables when there was more than two groups. Post hoc testing using Dunn's Test was performed to identify differences between pairs of groups when the Kruskal-Wallis test indicated significant variance between groups. During comparison of exposed and non-exposed groups within the cohort, the cohort was weighted so that it represented the deprivation of the eligible births in the LWH using factor weighting. The weighting factors used are shown in the appendix.

Correction for Multiple Testing Analysis

Post hoc analysis of 'maternal age', 'maternal education', 'IMD decile' and 'breastfeeding time' caused multiple testing. Multiple testing increases the chance of finding a significant result by chance (159). Because of this, the significance value cut off is usually adjusted to account for this. One commonly used method is the Bonferroni adjustment. Bonferonni adjustment states that a significant p value = p/n where n is the number of tests that have been performed (159). In the analysis of this thesis is was decided not to correct the analysis for multiple testing. The reason for this is that for the purpose of this thesis the LBBS is interested in trends in scores with changing variables (for example with increasing maternal age or increasing education or increasing breastfeeding time). Bonferroni adjustment is often over conservative (159), and therefore these trends may be missed if used. Likewise finding a significant result when using Bonferrioni adjustment is proportional to the number of tests done (159); in the LBBS the extremes of maternal age are lost and therefore less tests are done. This changes the adjusted significant p vale which may mean changes in the difference in score that is significant. This being said, any results produced in this thesis where multiple testing has occurred must be approached with this in mind.

Symptom Prevalence and Health Care Attendance

Symptom prevalence and health care service attendance in exposed and non-exposed groups was compared by χ^2 analysis and Fisher's Exact Test where appropriate. Fisher's Exact Test was indicated where expected values were deemed to be less than 10. During comparison of exposed and no exposed groups within the cohort, the cohort was weighted so that it represented the deprivation of the eligible births in the LWH using factor weighting. The weighting factors used are shown in the appendix.

The Impact on the Child's and their Family's Quality of Life

Domains seven and eight in the LRSQ are quality of life domains for the infant and their families respectively. Analysis of these domains was identical to other domains in the LRSQ

as described in the 'LRSQ scores' section above. The prevalence of the individual impacts on the QoL of both the infant and these were shown graphically at each time point.

2.2.9-Longitudinal Analysis

Longitudinal analysis of the change in LRSQ score over time and domain scores over time was analysed using multilevel mixed-effects modelling and compared exposed groups to nonexposed groups. The final multi-Level model used in the analysis of the LBBS therefore consisted of three levels; 1.IMD decile at birth, 2.highest maternal qualification obtained and 3.questionnaire number. Exposures and risk factors deemed to significantly effect LRSQ scores over time if $p \le 0.1$.

Justification and Development of the Methods of Longitudinal Analysis in the LBBS

The LBBS required a method of longitudinal analysis that allowed for the reduction of confounding variables and could assess multiple variable simultaneously. This could be done by either multivariate regression analysis or using multi-level mixed effects models. The LBBS needed a method of analysis that could allow for a large amount of missing data due to many participants completing some questionnaires and not others. Multi-level mixed effects models coped with missing data more effectively than the multivariate regression method of analysis. For this reasons multi-level mixed effects models were chosen as the method for longitudinal analysis in the LBBS. The multi-level model originally consisted of only two levels; 1.IMD decile at birth and 2.questionnaire number which allowed for both the under representation of the more deprived population of Liverpool and participants who had not completed every questionnaire.

To develop the longitudinal models, risk factors of respiratory disease were firstly added to the model individually. Individual exposures and risk factors deemed to change LRSQ scores over time significantly if p \leq 0.1. Once risk factors and exposures were deemed to have a significant impact on LRSQ scores over time they were added into the model one at a time, beginning with the exposure or risk factor with the smallest p value. Risk factors and exposers with a p value that remained \leq 0.1 once multiple variables had been added to the model remained in the final model. If p values of variables became larger than 0.1 as more variables were added to the model, they were removed from the model. If the addition of a variable caused multiple variables' p value become greater than 0.1 the variable with the largest p value was removed first.

During the longitudinal analysis of the LBBS data it became apparent that the category 'Highest maternal qualification' was causing a confounding effect of the results of the LBBS. It consistently removed many risk factors and exposures from the models when it was added to the model. For this reason it was decided by the research team to add 'highest attained maternal qualification' as a level in the multi-level mixed effects model. The final model used in the LBBS therefore consisted of three levels; 1.IMD decile at birth, 2.highest maternal qualification obtained and 3.questionnaire number. The longitudinal analysis was then performed again using the methods previously described.

2.2.10-Questionnaire Validation

The internal validity of the domains within the LRSQ was tested using Cronbach alpha test to validate the LRSQ in a longitudinal study. An α score of >0.7 was deemed to show acceptable internal constancy, α >0.8 was deemed to show good internal consistency and α >0.9 was excellent internal constancy. Below α =0.7 was deemed as unacceptable internal consistency (1).

2.2.11-Presentation of the Analysis of the LBBS in this Thesis

There is a large amount of data analysis from the LBBS to be presented in this thesis. The presentation of the results will be split into seven sections:

- 1. Analysis of recruitment,
- 2. Analysis of the attrition in the LBBS,
- 3. Analysis of the profile of the mothers in the LBBS,
- 4. Analysis of the profile of the infants in the LBBS
- 5. Results of the cross sectional analysis,
- 6. Results of the longitudinal analysis,
- 7. Results of the validation of the LRSQ.

Recruitment

The recruitment and response rates in the LBBS will be described using descriptive statistics. The reasons for participants not being recruited will also be discussed. This will also be shown pictorially in a flow diagram.

Attrition

Attrition in the LBBS will be described. A flow diagram will be used to describe the reasons for attrition. Following this analysis of the drop out population will be show. This will be described and shown graphically to show the difference between the drop out population and the remaining cohort. A table will be presented to show the difference in total LRSQ scores in the drop out population and the remaining population. This will show the impact of the difference in the characteristics of the drop out population and the remaining cohort on symptom scores. Following this there will be a description of the ways in which this uneven

attrition was accounted for during the analysis. This will be justified using graphs to show the reduction in the confounding effects of the uneven drop out after adjustment.

Profile of the Mothers in the LBBS

The characteristics of the mothers in the LBBS cohort will be presented. The characteristics will be described and compared to LWH births, or the population of England or Liverpool were appropriate. These comparisons will be presented graphically or in tables.

Profile of the Infants in the LBBS

The characteristics of the infants in the LBBS cohort will be presented. The characteristics will be described and compared to LWH births, or the population of England or Liverpool were appropriate. These comparisons will be presented graphically or in tables.

Cross Sectional Analysis

For the presentation of the cross sectional analysis each questionnaire will be presented individually in chronological order. For each questionnaire, firstly the summary statistics of the LRSQ scores will be presented alongside a histogram of the LRSQ scores to show the spread of scores in the cohort. Secondly the comparison of total LRSQ scores between infants exposed and not exposed to risk factors will be presented. A table that shows the comparison of LRSQ scores between those exposed and unexposed to risk factors will be presented; this table shows the median, the interguartile range, the mean and the mode score for each exposure/risk factor group. Results that show a statistically significant score between those exposed and not exposed to risk factors will be described in detail and a box and whisker diagram of these results will be shown. This will be followed by an analysis of each of the domain scores in the LBBS. A table will be presented with the minimum and maximum score, the median, mean and the standard deviation of each of the domain. This table shows which domains scored highest and lowest and allows the comparison of domains. Differences in domain scores between exposed and non-exposed groups will be described; only significant results will be described. Following this, the prevalence of specific respiratory symptoms in the previous 3 months will be presented. This will be presented graphically per 100 infants. A comparison of symptom prevalence in the previous 3 months in those exposed and not exposed to risk factors and exposure will be made. Only exposures with significant difference will be described. The use of health care services by infants in the previous 3 months will be presented. The difference in health care attendance n exposed and non-exposed groups will be described. Only significant results will be described. Health care attendance to infants born to mothers of different maternal ages and with different levels of education will be presented graphically. The effect of respiratory symptoms of QoL of the infants and their families will then be presented. A description of how the QoL of the infants and their families was affected by respiratory disease will be shown graphically. This will be followed by a description of differences in the effect of respiratory symptoms on QoL in those exposed and not exposed to risk factors and exposures. Finally a correlation between the QoL scores and respiratory symptom scores will be made.

Longitudinal Analysis

The longitudinal analysis of the LBBS will be presented as final multi-level mixed effects models. The coefficient, standard deviations and p values of risk factors and exposures that have a significant effect on the LRSQ scores over time will be presented. The total LRSQ score model will be presented first, followed by a model for each of the domains in the LBBS.

Questionnaire Validation

The Coronach alpha coefficients and a corresponding description of the domains internal consistency of each of the domains in the LBBS will be presented in a table.

Chapter 3- Results of the LBBS

This section presents the results of the LBBS. This presentation has 7 sections and will be presented as discussed in section 2.2.10. At the beginning of each section there will be a description of what the results that follow show and how this relates to the study aims.

3.1-Recruitment

This section outlines the recruitment rates in the LBBS. It shows the reasons for nonrecruitment. The reasoning behind these results and their significance is discussed in chapter four.

Recruitment led to 18.84% (n=2,346) of all the eligible births in the LWH (n=12,454) declaring their interest in taking part in the study. Recruitment analysis is shown in figure 7. Of all the births in the LWH between 23/01-2013 and 03/11/2014, 6.99% births occurred during breaks in recruitment. Of the births during active recruitment 11.94% were ineligible due to their postcode, 0.71% were ineligible due to speaking insufficient English, 0.52% were ineligible due to infants being taken directly into care of social services and 0.46% were not suitable to recruit as the infant had a futile prognosis. This led to 12,454 eligible births in the LWH during active recruitment. It was documented that 0.55% (n=68) of these actively declined to be a part of the study. 80.61% (n=10,040) of the eligible births could not be labelled either declined or interested in the study so therefore these births are assumed to have not shown interest in the study.

All interested families (n=2,346) were sent a consent form and questionnaire one when their child was four months of age by post or email according to their expressed preference. 18.84% (n=694) of the interested parents (684 Mothers and 10 Fathers), equating to 5.57% of all the eligible births, consented to the study and completed the initial questionnaire. See table 4. For the purpose of this thesis the parents of the infants will be referred to as mothers due to the negligible proportion of fathers.

Of the 2,364 infants whose mothers had declared their interest in the study 1.5% (n=36) did not consent to the study, 2.1% (n=49) gave a wrong postal address and could not be traced, 8.7% (n=204) gave a non-valid email and could not be traced and 0.2% (n=5) infants died in the first four months of life. 58.7% (n=1,358) of the infants whose mothers had declared their interest in the LBBS shortly after delivery were not enrolled four months later.



Figure 7: Flow diagram of the recruitment in the LBBS

Table 4:	Recruitment	rates	in	the	LBBS

Recruitment Stage	Eligible Births	Interested	Consented and Completed Q1
Response	12,454	2,346	694
% of Eligible LWH Births	100%	18.84%	5.57%

3.2-Drop Out

Attrition in the LBBS is described in this section. The cause of attrition in the study is described. The profile of the drop out population is compared to the remaining cohort. The difference in these profiles forms the basis for the choice of analysis methods in the LBBS. The justification of the analysis methods is also described in this chapter; these methods of analysis have allowed the LBBS to successfully complete the following aims of the current thesis;

- To describe respiratory symptoms of preschool children using the LRSQ from birth until 22 months of age.
- To determine the differences in LRSQ scores in the different populations at each questionnaire time point, using univariate and multivariate analysis.
- To determine the change in the LRSQ scores over time and the variability in this change between different population groups in the LBBS.

3.2.1-Summary

Of the 694 mothers that were enrolled and completed questionnaire 1, 54.0% (n=375) completed questionnaire 2, 39.6% (n=275) completed questionnaire three and 37.0% (n=257) completed questionnaire 4. The analysis of dropout in the LBBS is shown in table 5 and figure 8.

Questionnaire	Q1- 4 Months	Q2-10 Months	Q3-16 Months	Q4-22 Months
Response	694	375	275	257
% of Q1 Cohort	100%	54.0%	39.6%	37.0%
% Dropout from Previous Questionnaire		46.0%	26.7%	6.55%

Table 5: Attrition in the LBBS



Figure 8: Flow diagram of attrition in the LBBS

3.2.2-Recognition of Drop Out and Bias

During the early analysis of the cohort, it was noted that that drop out occurred more often in participants from more deprived postcodes. In the two most deprived deciles (IMD deciles one and two), drop out after four questionnaires was 73.3% and 65.0% respectively compared to 53.3% drop out and 60% drop out in the two least deprived deciles (IMD deciles nine and 10). This is shown in figure 9. Initial analysis showed that maternal education groups had unequal drop out. Mothers who were more highly educated dropped out less frequently than those who had a lower level education. This is shown in figure 10. Maternal age also had an uneven distribution of drop out as the study went on; older mothers were more likely to remain in the study. The mean age of the mothers in the cohort was 32 years at four months, 32 years and three months at 10 months, 32 years and 10 months at 16 months and 33 years and five months at 22 months. Figure 11 shows that younger mothers are more likely to drop out of the study than older mothers. This uneven drop out from the study produced a confounding effect within the study; infants who had been exposed to smoking pre and postnatally and had not been breastfed were more likely to drop out of the study; see figure 12 and 13. Table 10 shows the total LRSQ score for the cohort and the drop out population for the previous questionnaire. The drop out population scored higher in the previous LRSQ than the population that remain in the cohort. This clearly shows that the uneven drop out in the study population has created a confounding effect in the study and this drop out needed to be adjusted for in the analysis of the study.





Figure 9: Accumulative percentage drop out in the LBBS in each IMD decile at each questionnaire

Figure 10: Accumulative percentage drop out in the LBBS by mother's highest qualification attained



Figure 11: Percentage of the cohort with mothers in each age category at each questionnaire in the LBBS





Figure 12: Percentage of the cohort exposed to tobacco smoke at each questionnaire in the LBBS

Figure 13: Percentage of the cohort that were breastfed at each questionnaire in the LBBS

Table	6:	Previous	questionnaire	total	LRSQ	score	comparing	the	drop	out	population	to	the
remai	nin	g LBBS col	hort										

		Cohort	Cohort	Drop	Cohort	Drop	Cohort	Drop
		Q1	Q2	Out	Q3	Out	Q4	Out
TScore	Mean	9.9	9.4	10.5	9.1	10.5	9.7	10.1
T SCOLE	Median	5	5	6	5	6	5	5
	Mean		13.4	-	13.6	13.3	12.8	14.2
	Median		9	-	9	8.5	8	9
	Mean				15.3	-	14.7	17.1
	Median				10	-	9	13

3.2.3-Choice of Analysis to Mitigate for Drop Out Bias

After discussions with Dr Steve Lane at the University of Liverpool, it was decided that crosssectional analysis would be weighted using factor weighting to ensure the cohort at each time point was representative of the eligible births in the LWH. It was decided that the data should be weighted so that the IMD decile of the cohort was representative of that of all the eligible births in the LWH. Weighting factors for IMD decile at each questionnaire is shown in the appendix. A multilevel, mixed effects linear model would be used for the longitudinal analysis of the data. The model would have three levels: 1. IMD decile, 2. Highest maternal qualification and 3. Questionnaire number. This would account for the unequal drop out from each of the IMD Deciles and the maternal qualification groups. It also allowed for inclusion of data where one of a series of questionnaires for an individual in the cohort was missing.

3.2.4-Justification of Weighting Data – The Effect on Variables in the Cohort Breastfeeding in the LBBS

Over time the proportion of breastfed infants in the LBBS cohort increased due to the higher drop out of mothers who did not breastfeed. Weighting of the data reduced this change in the breastfeeding population somewhat. See figure 14. Although weighting reduced the drop out of the nonbreastfed infants the proportion of breastfed infants was still significantly higher in the LBBS when compared to all the eligible births at LWH at four months (χ^2 =73.3 p<0.05), 10 months (χ^2 =65.7 p<0.05), 16 months (χ^2 =82.8 p<0.05) and 22 months (χ^2 =66.8 p<0.05).



Figure 14: Percentage of the cohort breastfed in the LBBS before and after weighting by IMD decile

Smoking in the LBBS

Over time the proportion of the cohort that had been exposed to smoking in pregnancy and since birth reduced. Although weighting of the analysis reduced this decline somewhat, the proportion exposed to tobacco smoke during pregnancy and since birth was still lower than the eligible births at the LWH and the population of Liverpool. See figures 15-18 below.





Figure 15: Percentage of the cohort in the LBBS exposed to maternal smoking in pregnancy before and after weighting by IMD decile

Figure 16: Percentage of the cohort in the LBBS exposed to ETS in pregnancy from other household smokers before and after weighting by IMD decile



Figure 17: Percentage of LBBS cohort exposed to any smoking in pregnancy before and after weighting by IMD decile

Figure 18: Percentage of the cohort in the LBBS exposed to household smoking postnatally before and after weighting by IMD decile

The proportion of the cohort exposed to smoking, both during pregnancy and after birth are shown in table 11. Although weighting reduced the drop out of the infants exposed to tobacco smoke, the proportion of the cohort exposed to maternal smoking during pregnancy was still significantly smaller than the proportion of all the eligible births from the LWH at four months (χ^2 =21.5 p<0.05), 10 months (χ^2 =16.8 p<0.05), 16 months (χ^2 =16.0 p<0.05) and

22 months (χ^2 =13.8 p<0.05). Those infants exposed to ETS after birth was significantly smaller in the LBBS cohort than to population of Liverpool that smoked at four months (χ^2 =4.6 p<0.05) 16 months (χ^2 =12.3 p<0.05), and 22 months (χ^2 =11.7 p<0.05) when compared to 2012 data and was lower in the LBBS cohort at 16 months (χ^2 =9.5 p<0.05) and 22 months (χ^2 =9.1 p<0.05) when compared to 2015 data.

			Q1	%	Q2	%	Q3	%	Q4	%
	I PPS Cohort	No	606.0	90.5	345.0	92.0	257.0	93.5	241.0	93.8
Maternal	LBB3 CONOL	Yes	64.0	9.6	30.0	8.0	18.0	6.6	16.0	6.2
Smoke	Waightad	No	600.0	89.2	332.0	90.5	249.0	91.5	229.0	91.2
	weighted	Yes	73.0	10.8	35.0	9.5	23.0	8.5	22.0	8.8
	I DDC Cohort	No	537.0	80.2	307.0	81.9	238.0	86.6	216.0	84.1
	LBBS CONOL	Yes	133.0	19.9	68.0	18.1	37.0	13.5	41.0	16.0
Other Smoke		No	529.0	78.5	292.0	79.8	229.0	84.5	204.0	81.3
	weighted	Yes	145.0	21.5	74.0	20.2	42.0	15.5	47.0	18.7
	LDDS Cobort	No	513.0	76.6	292.0	77.9	227.0	82.5	210.0	81.7
Any Smoking	LBBS CONOL	Yes	157.0	23.4	83.0	22.1	48.0	17.5	47.0	18.3
In Pregnancy		No	501.0	74.4	275.0	75.1	214.0	79.0	195.0	77.7
	weighted	Yes	172.0	25.6	91.0	24.9	57.0	21.0	56.0	22.3
	IDDC Cohort	No	545.0	81.3	309.0	82.4	237.0	86.2	218.0	84.8
Llomo Smoko	LBBS COnort	Yes	125.0	18.7	66.0	17.6	38.0	13.8	39.0	15.2
	Waightad	No	534.0	79.3	228.0	77.8	222.0	85.1	206.0	85.1
	vveighted	Yes	139.0	20.7	65.0	22.2	39.0	14.9	36.0	14.9

Table 7: The effect of weighting of the cohort by IMD decile on smoking exposures in the LBBS

3.3-Profiles of Mothers and Families in the Liverpool Baby Breathing Study

This section describes the profiles of the mothers and families enrolled in the LBBS. It also compares the profile in the LBBS to the population of Liverpool and the mothers and families of the eligible births in the LWH. This description and comparison has allowed the following aim of this thesis to be partially achieved;

• To create a profile of the infants and mothers enrolled in the LBBS and make a comparison with the population of Liverpool and those born in the LWH.

3.3.1-Deprivation

Deprivation decile was calculated by using an online postcode converter (155). Participants in IMD decile one had the highest level of multiple deprivations. The distribution of IMD decile by postcode in the LBBS is shown below in comparisons to all the eligible births in the LWH and the population of Liverpool. The LBBS had a lower proportion of the cohort in IMD decile one (36.4%) compared to all the eligible births in the LWH (51.7%) and the population of Liverpool (45.0%) and had more in IMD decile 10 (1.6%) than the LWH eligible births (0.53%) and the Liverpool population (0.34%). See figure 19.



Figure 19: Percentage of the LBBS cohort in each IMD decile in comparison to LWH births and the Liverpool population

3.3.2-Maternal Age

The question regarding maternal age was added to the study after the pilot study, therefore only 567 mothers gave their age. The mean age of the mothers in the study was 31 years and 9 months. The youngest mother was 16 and he oldest 47 years of age giving a range of 31 years. Comparisons for the maternal ages in the LBBS were made to that of all the eligible births in the LWH and the births in Liverpool and England in the year of 2015. This showed

than the mothers in the LBBS are more likely to be older than those born in LWH, Liverpool and England. See figure 20.



Figure 20: Percentage of mothers in the LBBS cohort in age group in comparison to LWH, Liverpool and England births

3.3.3-Maternal Education

The cohort in the LBBS had 50.41% mothers with a degree or a post graduate degree, 20.0% of the cohort only attended higher education, 7.49% had A-Levels and 15.13% had GCSE's. 1.73% of the population had no qualifications. The remaining 5.48% had 'other' qualifications. When compared to the population of Merseyside and England, the LBBS cohort was more highly educated. The LBBS had higher a proportion of the people with a degree postgraduate degree or who or attended higher education and fewer people with A-levels, GCSE's, 'other' qualifications or no qualifications. See figure 21.

3.3.4-Smoking

Smoke exposure during pregnancy was explored through two questions. One assessing maternal smoking at any time



Figure 21: Percentage of the cohort by highest qualification attained in the LBBS in comparison to the population of Merseyside and England



Figure 22: Percentage of mothers that smoked during pregnancy in the LBBS in comparison to LWH, Liverpool and England births

during pregnancy, and one assessing maternal exposure to tobacco smoke (passive smoking) in the household during pregnancy. Of the 694 respondents 9.22% (n=64) of mothers smoked during pregnancy. This was compared to data from all eligible births from the LWH and the population of Liverpool and England. χ^2 analysis has shown these be significantly higher than the LBBS cohort $(\chi^2=33.8 \text{ p}<0.05 \text{ DoF}=1, \chi^2=28.9 \text{ p}<0.05$ DoF=1, χ^2 =4.2 p<0.05 DoF=1 respectively). See figure 22.

19.45% (n=13) of the cohort was exposed to other smoking in the household in pregnancy. The location of smoking by these other household smokers is shown in figure 23. Overall 22.91% (n=159) of the cohort was exposed to tobacco smoke of any kind in pregnancy.

3.3.5-Breastfeeding

In the LBBS 70.5% (n=489) of the cohort were breastfed. This was compared to the proportion of the eligible births from the LWH that were breastfed and all the births in Liverpool and England. The study cohort had a significantly higher proportion of breastfed infants than LWH births and all the births in Liverpool (χ^2 =127.7 p<0.05 DoF=1, χ^2 =78.2 p<0.05 DoF=1,). The proportion of breastfed infants in the LBBS was significantly lower than that of England (χ^2 =4.5 p<0.05 DoF=1). See figure 24. At four months 241 of the mothers were still breastfeeding, 98 had breastfed for greater than one month but



Figure 23: The location of smoking of household smokers during pregnancy in the LBBS



Figure 24: Percentage of breastfed infants in the LBBS in comparison to the rates of breastfeeding in LWH, Liverpool and England



Figure 25: Duration of breastfeeding in breastfed infants in the LBBS

were no longer breastfeeding at four months and 148 breastfed for less than one month. This is shown in figure 25.

3.3.6-Family History of Atopy

59.8% (n=415) of the cohort had a family history of atopy in at least one first degree sibling. 61.7% (n=256) of the cohort infants had a mother with atopy, 52.0% (n=216) had a father with atopy and 26.0% (n=108) had a brother or sister with atopy. Of the family members with atopy: 56.4% (n=234) had asthma, 77.8% (n=323) had eczema, and 64.1% (n=266) had hay fever. The overlap of atopic disease in the LBBS cohort is shown in figure 26.



Figure 26: Combinations of atopic disease in first degree relatives in the LBBS

3.4-Profile of Infants in the Liverpool Baby Breathing Study

This section describes the profiles of the infants enrolled in the LBBS. It also compares the profile in the LBBS to the population of Liverpool and the eligible births in the LWH. This description and comparison has allowed the following aim of this thesis to be completed;

• To create a profile of the infants and mothers enrolled in the LBBS and make a comparison with the population of Liverpool and those born in the LWH.

3.4.1-Sex

Of the 694 infants in the cohort, 49.7% (n=345) were male, and 50.3% (n=349) were female. This was not significantly different to all the births in Liverpool, the North West or Great Britain for a year in the recruitment period (p>0.05). See figure 27.

3.4.2-Multiple Births

The LBBS cohort contained 680 singletons and seven sets of twins. There were no triplets in the study. This was not



Figure 27: Proportion of males and females in the LBBS in comparison to the population of Liverpool, the North West and Great Britain

significantly different (p>0.05) to all the eligible births in the LWH and all the births in England.

3.4.3-Ethnicity

99.7% of the cohort gave information on their ethnicity. 90.6 of the cohort were white, 6.1% were mixed ethnicity, 0.7% were Asian, 1.4% were black and 0.3% were 'other' ethnicities. The cohort had a higher proportion of white and mixed ethnicity infants and lower proportions of Asian, black and other ethnicities when compared to the births at LWH and the population of Liverpool. See table 12.

Ethnicity	LBBS	LWH	Liverpool
White	90.63%	87.69%	88.91%
Mixed	6.05%	0.94%	2.52%
Asian	0.72%	5.29%	4.16%
Black	1.44%	3.59%	2.46%
Other	0.86%	2.49%	1.77%

Table 8: Ethnicity in the LBBS in comparison to LWH and Liverpool Births

3.4.4-Gestation

Gestational age of the infants enrolled in the LBBS and all the eligible births in the LWH are shown in figure 28. Preterm birth was defined as birth before 37 full weeks of pregnancy, indicated by the blue dotted line (160). 12.7% of the infants enrolled in the LBBS study were born preterm. This is significantly higher than the eligible births in the LWH (9.8%- χ^2 =6.00 p<0.05 DoF=1) and births in England (7.8%- χ^2 =22.6 p<0.05 DoF=1).



Figure 28: Gestational age of the LBBS cohort in comparison the LWH births

3.4.5-Birth Weight

The distribution of birth weights of babies in the LBBS, as well as the births in the LWH, the North West and England are shown in figure 29. Low birth weight was defined as birth under 2500 grams (161) and is indicated by the blue dotted line. 8.93% (n=62) of the births in the LBBS were low birth weight. This was not significantly different to that of the eligible births in the LWH (8.7% p>0.05), however was significantly higher than the births in England (6.5%- χ^2 =6.9 p<0.05 DoF=1).



Figure 29: Birth weight of the LBBS cohort in comparison the LWH, North West and England births

3.5-Cross-sectional Analysis

The cross sectional analysis of the LBBS data is split into 4 sections; one for each of the four time points that the infants were assessed for the current thesis. The presentation of each of the individual time points is described in section 2.2.10. The presentation of each other the time points individually in chronological order in the layout described earlier has allowed the following aims of this thesis to be achieved;

- To describe respiratory symptoms of preschool children using the LRSQ from birth until 22 months of age.
- To determine the differences in LRSQ scores in the different populations at each questionnaire time point.

3.5.1-Questionnaire 1-4 Months

Cohort Profile, Exposures and Risk Factors at 4 Months

Nursery Attendance, Other Household Children and Sharing a Bedroom

4.19% (n=28) of the cohort attended nursery in the first four months. 68.91% (n=416) of the cohort shared a bedroom. 97.41% (n=452) of these shared a bedroom with their parents, 4.4% (n=20) of which shared with their parents and another child. 1.94% (n=9) shared with another child but not their parents and 0.65% (n=3) shared their bedroom with another family member. 36.17% (n=242) of the cohort lived in a household with at least one other child. The number of other children in these households is shown in figure 30.



Figure 30: The number of other household children in the LBBS households at 4 months of age.

Household Smoking

18.66% (n=125) of the cohort was exposed to ETS; see figure 31. Thirteen infants were exposed to tobacco smoke inside the house, 120 stated they smoked outside, four smoked in the car and eight smoked elsewhere. The reported household smoking was significantly lower than the smoking rate for Liverpool in both 2012 (χ^2 =9.83 p<0.05 DF=1) and 2015

 $(\chi^2 = 6.27 \text{ p} = 0.036 \text{ DF} = 1).$



Figure 31: Percentage of households with smokers at 4 months in the LBBS

Underlying Medical Conditions

At four months of age 8.50% (n=59) of the cohort had at least one underlying medical condition. The underlying medical conditions in the cohort at four months are shown in table 13.
Type of Underlying Condition	Underlying Condition	Frequency
Cardiac (7)	Congenital Heart Disease	6
	Heart Murmur	1
	Reflux	8
Gastrointestinal (15)	Milk Intolerance	5
	Blood in the Stools	1
	Gastroenteritis	1
	'Bleed on the Brain'	1
Neurological (3)	Hydrocephalous	1
	Sagittal Craniosynostosis	1
Other (27)	Eczema	17
	Allergies	5
	Hip Dysplasia	2
	Cleft Pallet	1
	Hypospadias	1
	Trisomy 21	1
	Cystic Fibrosis	2
	Floppy Larynx	2
	Bronchitis	1
Respiratory (9)	Chronic Lung Disease of Prematurity	1
	Lung Collapse	1
	Oxygen Therapy (Reason not defined)	1
	Recurrent Bronchiolitis	1
Urinary (2)	Vesico-ureteric Reflux	1
	Duplex Kidney	1

Table 9: Underlying medical conditions in the LBBS cohort at 4 months

Total LRSQ Score

The total LRSQ score was calculated for each participant out a maximum of 132. The lowest total score was 0 and the highest score was 109. The mean score was 10.44, the median was 6 and the mode was 1. The distribution of the Total LRSQ scores at four months is shown in figure 32.



Figure 32: Histogram of Total LRSQ scores at 4 months

Total LRSQ Sores in those Exposed and Unexposed to Risk Factors and Exposures

Summary Statistics comparing demographics, exposures and risk factors are shown in table 10. Males had a significantly higher total respiratory symptom questionnaire score than females (z=-3.214 p=0.001); see figure 33. Infants born to mothers aged 25-29 years had significantly higher scores than children born to mothers in age groups older than this (p<0.05). Infants born to mothers less than 20 years of age had a higher score than those born to mothers aged over 45 years (p=0.046); see figure 34. Those with a family history of atopy had a significantly higher total score than those without (z=-2.322 p=0.02); see figure 35. Those who shared a bedroom had a lower score than those who did not share a bedroom (z=-2.361 p=0.018); see figure 36.

		25%	Median	75%	Mean	Mode
Information Court	Male	3	7	15	11.88	2
Infant's Sex*	Female	1	5	12	9.03	1
	Single Birth	2	6	14	10.45	1
Multiple Birth	Multiple Birth	1.71	5.42	13.21	9.96	0
	White	2	6	14	10.7	0
Ethnicity	Other	2	5	10.63	7.757	5
	<20	2	9.41	10	6.83	10
	20-24	3	6.76	13.38	6.76	0
-	25-29	3	6	16.49	13.37	5
Maternal Age*	30-34	1	4.64	10	7.42	1
-	35-39	1	4	12	7.8	1
-	40-44	1.82	4	13.38	7.39	1
-	>=45	1.03	1.89	-	1.66	2
	1	2	7	15	11.7	0
-	2	2	5	12.56	8.85	0
-	3	2	5	14	9.71	0
-	4	2	6	12	9.33	0
	5	2	5.09	12	11.13	1
IMD	6	1.92	5.5	14.08	8.2	6
	7	1.09	5	10.31	7.57	1
	8	1	4	10.85	6.76	0
	9	2	3.14	12.07	7.62	2
-	10	5.21	12.86	19.82	10.7	7
	Degree or Higher	2	5	11	8.16	2
	Higher Education	2	8	14.16	11.43	1
	A Level	2	7	28.1	16.43	0
Maternal Qualification	GCSE	2	6	16	11.23	1
	Other	2	7.62	15	12.92	0
	None	5	10.11	20	17.82	5
	No	2	5.68	14	10.4	1
Premature Birth	Yes	2	6.32	14	10.78	0
	No	2	6	14	10.58	1
Low Birth Weight	Yes	1	5.38	12.74	8.97	0
	No	2	5	14	10.42	1
Maternal Smoking In Pregnancy	Yes	3	7	13	10.6	4
Other Household Smoking In	No	2	5	14	10.22	1
Pregnancy	Yes	3	7	14	11.24	0
Any Smoke Exposure in	No	2	5	14	10.13	1
Pregnancy	Yes	3	7	14.06	11.36	3
Hone Could	No	2	6	14	10.19	1
Home Smoking	Yes	2	6	14.62	11.39	0
Dura at f	No	2	7.95	15	12.62	0
Breastfeeding	Yes	2	5	13	9.41	1
	Never	2	7.95	15	12.62	0
Durantifica d'E	<1 Month	2	6.26	14.42	10.99	1
Breastfeeding Time	1-4 Months	2	5	13.95	8.8	2
	>4 Months	2	5	12	8.51	1
	No	2	5	12.02	8.7	0
Family History of Atopy*	Yes	2	5	15	11.64	2
	No	2	6	14	10.43	1
Nursery Attendance	Yes	2.9	5	12.15	11.25	5
	No	2	5	13	9.86	5
Other Household Children	Yes	2	6	15.91	11.49	1
	No	3	7	15	12	5
Shares a Bedroom*	Yes	2	5	14	9.79	1

Table 10: Comparison of LRSQ scores between different exposure/risk factor groups in the LBBS at 4 months

*indicates significant result (p<0.05)





Figure 33: Total LRSQ scores by infant sex at 4 months





Figure 35: Total LRSQ scores by FH of atopy at 4 months



Figure 36: Total LRSQ scores by sharing a bedroom at 4 months

LRSQ Domain Sores in the First 4 Months

The summary statics of the domain scores in the first four months of life are shown in table 14. Symptoms with colds and night-time symptoms were the highest scoring domains. 'Symptoms on increased activity' was the lowest scoring domain. Domains seven and eight are QoL domains and are discussed elsewhere.

	Min	Median	Мах	Mean	SD
Domain 1	0	1	15	1.44	2.01
Day Time Symptoms					
Night-time Symptoms	0	2	19	2.14	2.37
Domain 3	0	1	21	2.33	3.17
Symptoms with Colds	_				
Domain 4	0	0	16	0.78	1.6
Symptoms without Colds					
Domain 5	0	0	16	0.48	1.43
Symptoms on increased Activity					
Domain 6	0	0	12	1.16	1.9
Other Respiratory Symptoms					
Domain 7	0	0	12	0.64	1 56
Childs Quality of Life	0	0	12	0.04	1.50
Domain 8 Parents Quality of Life	0	0	16	0.98	2.17

Table 11: Summary statistics of domain scores at 4 months

Sex

Males scored higher in 'daytime symptoms' (z=-2.112 p=0.035), 'night-time symptoms' (z=-2.923 p=0.003) and 'other respiratory symptoms' (z=-2.77 p=0.006) than females.

Maternal Age

Kruskal-Wallis testing revealed variance between different maternal age categories in the domains 'day time symptoms' (χ^2 =13.90 p=0.033 DoF=6), 'night-time symptoms' (χ^2 =14.71 p=0.022 DoF=6), and 'symptoms without colds' (χ^2 =13.97 p=0.030 DoF=6). Post hoc testing revealed those infants born to mothers aged 25-29 years of age scored higher for 'daytime symptoms' than those born to mothers aged 30-34 (p=0.002) and 35-39 years old (p=0.016). Those born to mother 25-29 year of age scored higher for 'night-time symptoms' and 'symptoms without colds' than those born to mothers aged 30-34 (p=0.002) and p=0.003 and p=0.005), 35-39 (p=0.009 and p=0.006) and those aged 40-44 (p=0.042 and p=0.030) years of age.

Gestation

Preterm babies scored higher than term babies for 'symptoms without colds' (z=-2.148 p=0.032).

Breastfeeding

Children who were breastfed scored lower for 'symptoms on increased activity' than children who were not breastfed (z=-2.198 p=0.028). 'Night-time symptoms' scores varied depending on the length of time that the child was breastfed for (χ^2 =8.735 p=0.033 DF=3). Post hoc analysis showed those who were still being breastfed at four months had lower 'night-time symptom' scores than: those who were never breastfed (p=0.016); those who were breastfed for only up to one month (p=0.03); and those who were breastfed for one to four months (p=0.025).

Family History of Atopy

Infants with a family history of atopy scored higher than those without a family history of atopy for 'symptoms without colds' (z=-2.122 p=0.034), 'symptoms on increased activity' (z=-2.193 p=0.033) and 'other respiratory symptoms' (z=-2.193 p=0.028).

Other Household Children

Infants who lived in households with other children scored higher for 'symptoms with colds' than children who lived in households with no other children (z=-2.079 p=0.038).

Sharing a Bedroom

Infants who shared a bedroom scored lower for 'symptoms with colds' (z=-2.971 p=0.003) than those children who slept in their own bedroom.

Respiratory Symptom Prevalence in First 4 months of Life

89.85% (n=602) of the cohort had respiratory symptoms in the first four months of life. Snoring was the most reported symptom affecting 62.09% (n=416) of the cohort. 61.94% (n=415) of the cohort had at least one cold in the first four months of life. The lowest reported respiratory symptom was SoB, reported by only 12.54% (n=84) of the cohort. The prevalence of the symptoms in the previous three months of life per 100 infants are shown in figure 37.



Figure 37: Respiratory symptom prevalence per 100 infants in the first 4 months of life

Prevalence of Symptoms Compared to Risk Factors and Exposures

Sex

Males had a higher prevalence of SoB (χ^2 =10.91 p<0.05 DF=1), snoring (χ^2 =9.98 p<0.05 DF=1), 'noisy Breathing from the throat' (χ^2 =6.75 p<0.05 DF=1) and 'noisy breathing not from the chest' (χ^2 =10.44 p<0.05 DF=1) than females.

Ethnicity

White infants had a higher prevalence of rattily chest (χ^2 =6.22 p<0.05 DF=1), snoring (χ^2 =6.63 p<0.05 DF=1) and raid breathing (χ^2 =5.68 p<0.05 DF=1) than other ethnicities.

Maternal Age

 χ^2 testing showed variation in the prevalence of wheeze (χ^2 =15.28 p<0.05 DF=6), rattily chest (χ^2 =19.18 p<0.05 DF=6), SoB (χ^2 =17.16 p<0.05 DF=6) and rapid breathing (χ^2 =15.11 p<0.05 DF=6) in infants born to mothers of different ages. Post hoc analysis revealed that the prevalence of wheeze was higher in infants born to mothers aged 25-29 than those born to

mothers aged 30-34 (χ^2 =9.76 p<0.05 DF=1) and 35-39 (χ^2 =10.58 p<0.05 DF=1). The prevalence of rattily chest was higher in infants born to mothers aged 25-29 than infants born to mothers aged 30-34 (χ^2 =14.70 p<0.05 DF=1), 35-39 (χ^2 =7.40 p<0.05 DF=1) and 40-45 (χ^2 =6.14 p<0.05 DF=1). The prevalence of SoB was higher in those infants born to mothers aged 25-29 than in mothers aged 30-34 (χ^2 =14.14 p<0.05 DF=1) and 35-39 (χ^2 =5.02 p<0.05 DF=1). The prevalence of rapid breathing was higher in infants born to mothers aged less than 20 compared to those born to mothers aged 20-24 (χ^2 =7.12 Fisher's Exact Test p=0.022 DF=1), 30-34 (χ^2 =8.16 Fisher's Exact Test p=0.015 DF=1), 35-39 (χ^2 =5.58 Fisher's Exact Test p=0.037 DF=1) and 40-44 (χ^2 =9.61 Fisher's Exact Test p=0.001 DF=1). Rapid breathing was also higher in infants born to mothers aged 30-34 (χ^2 =4.52 p<0.05 DF=1) and 40-44 (χ^2 =3.9 p<0.05 DF=1).

Maternal Qualification

Kruskal-Wallis testing showed there was variation in the prevalence of rattily chest (χ^2 =13.39 p<0.05 DF=5) and SoB (χ^2 =14.77 p<0.05 DF=5) in infants born to mothers with different levels of education. Post hoc analysis revealed that the prevalence of rattily chest was lower in those infants born to mothers with a degree or post graduate degree than those born to mothers who attended higher education (χ^2 =6.54 p<0.05 DF=1), had a-levels (χ^2 =5.42 p<0.05 DF=1) or GCSE's (χ^2 =5.88 p<0.05 DF=1). Those infants born to mothers with A-levels had a higher prevalence of SoB than those who were born to mothers with a degree or postgraduate degree (χ^2 =9.38 p<0.05 DF=1), those who attended higher education (χ^2 =12.16 p<0.05 DF=1) or those with GCSE's (χ^2 =5.63 p<0.05 DF=1).

Gestation

Infants born preterm had a lower prevalence of colds than those who were born term (χ^2 =4.46 p<0.05 DF=1) but had a higher prevalence of rapid breathing (χ^2 =5.26 p<0.05 DF=1).

Birth Weight

Infants born with a low birth weight had a higher incidence of colds than those who were born with a normal birth weight (χ^2 =5.39 p<0.05 DF=1).

Breastfeeding

The prevalence of wheeze (χ^2 =12.17 p<0.05 DF=1) and SoB (χ^2 =4.51 p<0.05 DF=1) were lower in those who were breastfed than those who were not breastfed.

Family History of Atopy

Infants with a family history of atopy had a higher prevalence of 'noisy breathing from the throat' (χ^2 =7.04 p<0.05 DF=1) and 'noisy breathing not from the chest' (χ^2 =5.38 p<0.05 DF=1) than those with no family history of atopy.

Nursery Attendance

Those who attended nursery reported a higher prevalence of snoring (χ^2 =6.63 p<0.05 DF=1) than those who did not attend nursery.

Other Household Children

Infants who lived in households with other household children reported a higher prevalence of colds than those who did not live with other children (χ^2 =4.24 p<0.05 DF=1), however they reported a lower prevalence of snoring than those who did not live with other children (χ^2 =5.50 p<0.05 DF=1).

Sharing a Bedroom

Infants who shared a bedroom had a lower prevalence of cough (χ^2 =4.99 p<0.05 DF=1), SoB (χ^2 =6.67p<0.05 DF=1) and colds (χ^2 =19.66 p<0.05 DF=1) than infants who slept alone.

Health Care Service Attendance due to Respiratory symptom

15.82% (n=106) of the cohort attended the GP in the first four months of life with respiratory symptoms. 6.27% (n=42) attended the hospital with respiratory symptoms. GP attendance was higher in those infants who were exposed to tobacco smoke in pregnancy (χ^2 =9.33 p<0.05 DF=1), attended nursery (χ^2 =5.54 p<0.05 DF=1), lived with other children (χ^2 =18.79 p<0.05 DF=1) and shared a bedroom (χ^2 =4.63 p<0.05 DF=1) but was reduced in those who were breastfed (χ^2 =10.60 p<0.05 DF=1). Hospital attendance was increased in males (χ^2 =6.69 p<0.05 DF=1), those born preterm (χ^2 =7.76 p<0.05 DF=1) and those who live with other children (χ^2 =30.13 p<0.05 DF=1). Infants born to mothers with increasing maternal age reduced both GP and hospital attendance whereas infants born to mothers with a lower level of education attended more frequently. This is shown in figure 38 and 39. There was no association between IMD decile and health care service use.



Figure 38: Percentage of the cohort that attended their GP or hospital with respiratory symptoms in the first 4 months of life by maternal age group



Figure 39: Percentage of the cohort that attended their GP or hospital with respiratory symptoms in the first 4 months of life by highest maternal qualification attained

The Impact of Respiratory Symptoms on QoL

The impact of respiratory symptoms on the QoL of the infant and their mother's is shown figures 40 and 41 respectively. The QoL of 22.8% (n=153) of the infants in the cohort was effected by their respiratory symptoms. The QoL of 29.7% (n=199) of the mothers were effected by their infants respiratory symptoms. Children who shared a bedroom were reported to have a better QoL than those who slept alone (z=-2.427 p=0.015) whereas children who lived in households with at least one other children had a worse QoL than those who did not live with other children (z=-2.055 p=0.04). Mothers of males reported a worse QoL due to their child's respiratory symptoms compared to mothers of females (z=-3.411 p=0.001). Mothers of infants with a family history of atopy reported a worse QoL compared to mothers of infants without a family history of atopy (z=-2.015 p=0.044) and mothers of children who lived in households with other children reported that they had a worse QoL than those who lived in households with other children reported that they had a worse QoL than those who lived in households with other children reported that they had a worse QoL than those who lived in a household with no other children (z=-2.156 p=0.031).



Figure 40: The impact of respiratory disease on the infant's QoL in the first 4 months in the LBBS (per 100 infants)



Figure 41: The impact of respiratory disease on the infant's mother's QoL in the first 4 months in the LBBS (per 100 infants)

Spearman's rank correlation analysis showed that higher Respiratory symptom scores were correlated with higher QoL domain scores indicating a worse QoL (r_s =0.674 p<0.001). This is shown in figure 42.



Figure 42: Scatter graph of respiratory symptom scores and quality of life scores at 4 months

3.5.2-Questionnaire 2- 10 Months Cohort Profile, Exposures and Risk Factors at 10 Months

Nursery Attendance, Other Household Children and Sharing a Bedroom

At 10 months 26.93% (n=101) of the cohort attended nursery. 43.47% (n=163) of the cohort lived with at least one other child. The number of other children in these households is shown in figure 43. 61.33% (n=230) of the cohort shared a bedroom. Of these 80.69% (n=117) shared with their parents and 18.62% (n=28) shared with other children without their parents. One infant shared a bedroom with another family member.



Figure 43: Number of children in households at 10 months

Household Smoking

17.6% (n=66) of the infants in the cohort were exposed to ETS in the previous three months; see figure 44. Of these smokers 4.5% (n=3) smoked inside, 95.5% (n=63) smoked outside, 3.0% (n=2) smoked in the car and 4.5% (n=3) smoked elsewhere. This reported rate of smoking was significantly lower than the smoking prevalence in Liverpool in 2012 (χ^2 =8.86 p<0.05 DF=1) and in

2015 (χ²=6.13 p<0.05 DF=1).





Underlying Medical Conditions

By 10 months 9.05% (n=34) of the cohort had at least one underlying medical condition. The underlying health conditions in the LBBS cohort at 10 months are shown in table 15.

Type of Underlying Condition	Underlying Condition	Frequency
Cardiac (3)	Congenital Heart Disease	3
	Imperforated Anus	1
Gastrointestinal (3)	Milk Intolerance	1
	Reflux	1
	Eczema	10
	Allergies	3
Other (18)	Clotting Disorders	2
	Hip Dysplasia	2
	Umbilical Hernia	1
	Asthma	2
	Bronchomalacia	1
Respiratory (6)	Chronic Lung Disease	1
	Constant Wheeze	1
	'Scaring of the lungs requiring oxygen	1
	therapy'	-
Urinary (2)	Vesico-ureteric Reflux	1
	Dilated Kidneys	1

Table 12: Underlying medical conditions in the LBBS at 10 months

Total LRSQ Score

At 10 months the lowest total score was 0 and the highest score was 114. The mean score was 13.50, the median was nine and the mode was 6. The distribution of the Totals LRSQ scores at 10 months is shown in figure 45.



Figure 45: Histogram of Total LRSQ Scores at 10 Months

Total LRSQ Score in Those Expose and Unexposed to Risk Factors and Exposures

Summary Statistics comparing demographics, confounding factors, exposures and risk factors are shown in table 13. Males had significantly higher LRSQ scores than females (z=-3.059 p=0.002); see figure 46. White babies had higher LRSQ scores than those in other ethnic groups (z=-2.771 p=0.006); see figure 47. Babies born preterm had higher LRSQ scores than those born term (z=-1.978 p=0.048); see figure 48. Those infants that had attended nursery had higher LRSQ scores than those who did not (z=-4.952 p<0.001); see figure 49. Infants born to mothers aged 20-29 years of age had higher LRSQ scores than those born to mothers aged 34-44 (p<0.05); see figure 50.

		25%	Median	75%	Mean	Mode
	Male	5	9.85	21	15.97	6
Infant's Sex*	Female	3	7	17	11.87	2
	Single Birth	4	8.86	20	14.14	0
Multiple Birth	Multiple Birth	2	2.52	16.84	9.19	2
	White	4	9	20	14.42	6
Ethnicity*	Other	2	4	12.79	7.89	2
	<20	1	11.52	23.34	11.59	1
	20-24	7	15.1	21.99	14.95	3
	25-29	4	10.28	28	18.07	4
Maternal Age*	30-34	4	9.27	19	12.94	0
	35-39	2	5.85	11.7	11.11	1
	40-44	2.46	3.63	11.95	7.06	3
	>=45	-	19	-	-	-
	1	3	8	21	14.91	0
	2	3	8	15.5	11.2	1
	3	2.04	5.17	14.15	11.58	2
	4	4.76	12	27.27	18.18	2
	5	4.08	10.17	14.37	11.6	6
IMD	6	2.5	9.49	12.62	11.36	0
	7	4.39	10.42	17.37	12.48	1
	8	4	11	18	11.33	4
	9	6.32	16.07	32.61	22.64	0
	10	4.43	19	-	18.67	4
	Degree or Higher	4	8	17.28	12.23	6
	Higher Education	2 46	7 37	16	11 23	2
	A Level	6	20	36	22.29	1
Maternal Qualification	GCSE	2 52	12	29.09	18.95	2
	Other	5	9.53	22.57	14 21	5
	None	2 12	5.33	20	9.71	3
	No	2.12	8	19	13 41	0
Premature Birth*	Vas	5	12.67	21	19.41	5
	No	2	2	10	14.01	0
Low Birth Weight	Voc	3	12.92	21	12.01	1
	No	2	0	21	13.23	
Maternal Smoking In Pregnancy	No	2	8	20	14.00	5
	fes	3	D	20.19	12.84	5
Other Household Smoking In Pregnancy	NO Yee	3	8	20	14.05	0
regliancy	Yes	4	10	18.85	13.52	5
Any Smoke Exposure in Pregnancy	NO	3	8	20	13.92	0
Tregnancy	Yes	4	10	19.52	14.03	5
Home Smoking	NO	4	8	20	14.13	0
	Yes	3	8	16.55	13.19	2
Breastfeeding	NO	4	11	22.05	15.98	5
	Yes	3	8	9	13.07	0
	Never	4	11	22.05	15.98	5
Breastfeeding Time	<1 Month	2.05	8	20	14.99	1
	1-4 Months	3.35	8.05	17.82	11.57	3
	>4 Months	3.25	8	18.76	12.46	0
Family History of Atopy	No	3	8.46	18	13	0
	Yes	4	8	20	14.67	6
Nursery Attendance*	No	3	7	16	11.85	2
	Yes	6	14	26	19.77	5
Other Household Children	No	4	9	18.49	12.39	6
	Yes	3	8	22	15.8	0
Shares a Bedroom	No	4	8	19	13.78	6
Shares a Bedroom	Yes	3	9	20	14.2	0

Table 13: Comparison of LRSQ scores between different exposure/risk factor groups in the LBBS at 10 months	

*indicates significant result (p<0.05)



Figure 46: Total LRSQ scores at 10 months by infant sex

Figure 47: Total LRSQ score at 10 months by infant ethnicity

Figure 48: Total LRSQ score at 10 months by preterm birth





Figure 49: Total LRSQ at 10 months by nursery attendance

Figure 50: Total LRSQ score at 10 months by maternal age

LRSQ Domain Scores in Months 4-10

The summary statics of the domain scores for months 4-10 are shown in the table 14. 'Symptoms with colds' and 'night-time symptoms' were the highest scoring domains. Symptoms on 'increased activity' was the lowest scoring domain. Domains seven and eight are QoL domains and are discussed elsewhere.

	Min	Median	Max	Mean	SD
Domain 1 Day Time Symptoms	0	1	15	1.9	2.3
Domain 2 Night-time Symptoms	0	2	18	2.5	2.6
Domain 3 Symptoms with Colds	0	3	20	4.1	3.6
Domain 4 Symptoms without Colds	0	1	12	0.7	1.5
Domain 5 Symptoms on Increased Activity	0	0	15	0.6	1.6
Domain 6 Other Respiratory Symptoms	0	0	9	0.7	1.5
Domain 7 Childs Quality of Life	0	0	13	1.4	2.2
Domain 8 Mothers Quality of Life	0	0	14	1.5	2.6

Table 14: Summary statistics of domain scores at 10 months

Sex

Males had significantly higher scores for 'day time symptoms' (z=-2.006 p=0.045), 'night-time symptoms' (z=-3.658 p<0.001), 'symptoms with colds' (z=-3.123 p=0.034) and 'symptoms on increased activity' (z=-3.466 p=0.001) when compared to females.

Ethnicity

White infants had higher scores for 'day time symptoms' (z=-2.108 p=0.002), 'night-time symptoms' (z=-2.137 p=0.033), 'symptoms without colds' (z=-2.275 p=0.023) and 'symptoms on increased activity' (z=-2.188 p=0.029) compared to infants in other ethnic groups.

Maternal Age

Kruskal-Wallis Testing revealed differences in scores for infants born to mothers of different maternal ages for 'day time symptoms' ($\chi^2 = 19.72 \text{ p} < 0.05 \text{ DF} = 6$), 'symptoms with colds' ($\chi^2 = 21.71 \text{ p} < 0.05 \text{ DF} = 6$), 'symptoms without colds' ($\chi^2 = 16.22 \text{ p} < 0.05 \text{ DF} = 6$), and 'symptoms on increased activity' ($\chi^2 = 13.31 \text{ p} < 0.05 \text{ DF} = 6$).

Post Hoc Testing Domain 1- Day Time Symptoms

Infants born to mothers aged 20-24, 25-29 and 30-34 had higher scores than infants born to mothers aged 35-39 (p=0.031, p=0.002, p=0.009) and 40-44 (p=0.006, p=0.002, p=0.005).

Post Hoc Testing Domain 3- Symptoms with Colds

Infants born to mothers aged 20-24 had higher scores than infants born to mothers aged 35-39 (p=0.02) and 40-44 (p=0.009). Infants born to mothers aged 25-29 had higher scores than infants born to mothers aged 30-34 (p=0.008), 35-39 (p=0.001) and 40-44 (p=0.001). Infants born to mothers aged 35-29 had higher scores than infants born to mothers aged 40-44 (p=0.049).

Post Hoc Testing Domain 4- Symptoms without Colds

Infants born to mothers aged less than 20, 25-29 and over 45 years had higher scores than infants born to mothers aged 35-39 (p=0.014, p=0.003, p=0.034) and 40-44 (p=0.027, p=0.04, p=0.038). Infants born to mothers aged 30-34 had higher scores than those born to mothers aged 35-39 (p=0.042).

Post Hoc Testing Domain 5- Symptoms on Increased Activity

Infants born to mothers aged 40-44 had lower scores than infants born to mother aged 20-24 (p=0.014), 25-29 (p=0.007), 30-34 (p=0.018), 35-39 (0.013) and over 45 years (p<0.001). Infants born to mothers over 45 years had a higher score than those born to mothers under 20 years (p=0.014).

Maternal Education

Kruskal-Wallis Testing revealed differences in domain scores for infants born to mothers with different levels of education for 'day time symptoms' ($\chi^2 = 11.96 \text{ p} < 0.05 \text{ DF} = 5$), 'symptoms without colds' ($\chi^2 = 32.35 \text{ p} < 0.05 \text{ DF} = 5$), and 'symptoms on increased activity' ($\chi^2 = 22.04 \text{ p} < 0.05 \text{ DF} = 6$).

Post Hoc Testing Domain 1- Day Time Symptoms

Infants born to mothers with a degree or a post graduate degree had lower scores than infants born to mothers with A-levels (p=0.027). Infants born to mothers who had attended higher education had lower scores than infants born to mothers with A-levels (p=0.009), GCSE's (p=0.048) and no qualifications (p=0.036).

Post Hoc Testing Domain 4- Symptoms without Colds

Infants born to mothers with A-levels had higher scores than infants born to mothers with a degree or post graduate degree (p<0.001) or who had attended higher education (p<0.001).

Infants born to mothers with GCSE's had higher scores than infants born to mothers or who had attended higher education (p=0.006) and those who had A-levels (p=0.015). Infants born to mothers with A-levels had higher scores than infants born to mothers or who had no qualifications (p=0.013) and those with 'other' qualifications (p=0.035).

Post Hoc Testing Domain 5- Symptoms on Increased Activity

Infants born to mothers with a degree or post graduate degree had lower scores than infants born to mothers with A-levels (p<0.001) and infants born to mothers who attended higher education had lower scores than infants born to mothers with A-levels (p<0.001) and GCSE's(p=0.006). Infants born to mothers with A-levels had lower scores than infants born to mothers with GCSE's (p=0.015) and those with no qualifications (p=0.013). Infants born to mothers with 'other' qualifications had lower scores than infants born to mothers with A-levels (p=0.035).

Smoking Exposure in Pregnancy

Those infants who were exposed to tobacco smoke in pregnancy whether that be maternal or other household smokers had lower scores 'without colds' than those who were not exposed (z=-2.228 p=0.026).

Breastfeeding

Those who were breastfed had lower scores for 'symptoms without colds' than those who were not breastfed (z=-2.56 p=0.011). Further testing revealed those who infants who were breastfed for four months or more had the lowest scores, and this was significantly lower when compared to those who had never been breastfed (p=0.004).

Family History of Atopy

Those infants with a family history of atopy scored significantly higher for 'symptoms with colds' than those who did not have a family history of atopy (z=-2.108 p=0.035).

Nursery Attendance

Those who attended nursery had higher scores for 'daytime symptoms' (z=-5.196 p<0.001), 'night-time symptoms' (z=-4.637 p<0.001), 'symptoms with colds' (z=-5.078 p<0.001), 'symptoms without colds' (z=-3.117 p=0.002) and 'other respiratory symptoms' (z=-2.52 p=0.012).

Respiratory Symptom Prevalence at 10 Months

93.1% (n=349) of the cohort had respiratory symptoms in the previous three months at 10 months of age. Colds were the highest reported respiratory symptom with 85.6% (n=321) of the cohort having at least one cold in the previous three months. The lowest reported respiratory symptom was rapid breathing reported by only 16.80% (n=63) of the cohort. The prevalence of the respiratory symptoms per 100 infants in the three months prior to 10 months of age are shown in figure 51.



Figure 51: Respiratory symptom prevalence at 10 months per 100 infants

Respiratory Symptom Prevalence in Those Exposed and Unexposed to Risk Factors

Males had a higher prevalence of cough (χ^2 =4.60 p<0.05 DF=1) and snoring (χ^2 =6.11 p<0.05 DF=1) than females. White infants had a higher prevalence of wheeze (χ^2 =4.81 p<0.05 DF=1) and rattily chest (χ^2 =5.77 p<0.05 DF=1) than infants of other ethnic groups. Preterm babies had a higher prevalence of rapid breathing than those who were born at term (χ^2 =6.70 p<0.05 DF=1). The prevalence of colds was higher in low birth weight babies than in babies born at a normal weight (χ^2 =5.64, Fishes Exact Test p=0.0128 DF=1). The prevalence of SoB (χ^2 =5.49 p<0.05 DF=1) and 'noisy breathing from the throat' (χ^2 =6.24 p<0.05 DF=1) was lower in babies that were breastfed. The prevalence of cold symptoms was higher in those infants with a family history of atopy when compared to those without a family history (χ^2 =8.05 p<0.05 DF=1). Infants who attended nursery had a higher prevalence of wheeze (χ^2

=6.19 p<0.05 DF=1), cough (χ^2 =4.75 p<0.05 DF=1), rattily chest (χ^2 =16.90 p<0.05 DF=1), colds (χ^2 =4.70 p<0.05 DF=1), and 'noisy breathing from the throat' (χ^2 =11.88 p<0.05 DF=1) than those who did not attend nursery. The prevalence of snoring was lower in those babies who shared a bedroom compared to those who slept alone (χ^2 =16.38 p<0.05 DF=1).

Health Care Service Attendance in Months 4-10

In months 4-10 29.07% (n=109) of the cohort attended the GP with respiratory symptoms. 8.80% (n=33) attended the hospital with respiratory symptoms. GP attendance was higher in those with a family history of atopy (χ^2 =14.09 p<0.05 DF=1), those who were born with a low birth weight (χ^2 =5.85 p<0.05 DF=1) and those who attended nursery (χ^2 =12.40 p<0.05 DF=1) but was reduced in those who were breastfed (χ^2 =5.83 p<0.05 DF=1). Hospital attendance was increased in infants with a family history of atopy (χ^2 =14.09 p<0.05 DF=1). Hospital attendance were born with a low birth weight (χ^2 =4.51 p<0.05 DF=1) but was reduced in those who were breastfed (χ^2 =9.52 p<0.05 DF=1). Increasing maternal age reduced both GP and hospital attendance; see figure 52. The relationship between maternal education and health care service attendance is shown in figure 53.







Figure 53: Percentage of the cohort that attended their GP or hospital with respiratory symptoms in months 4-10 of life by highest maternal qualification attained

The Impact of Respiratory Symptoms on QoL in Months 4-10

The impact of respiratory symptoms on the QoL of the infant and their families are summarised in figures 54 and 55 respectively. The QoL of 43.5% (n=163) of the infants in the cohort was effected by their respiratory symptoms. The QoL of 45.3% (n=163) of the cohorts mothers were effected by their infants respiratory symptoms.



Figure 54: The impact of respiratory disease on the infant's QoL at 10 months in the LBBS (per 100 infants)



Mothers reported that males had a worse QoL due to their respiratory symptoms than females (z=-3.411 p=0.001). White infants were reported to have a worse QoL when compare to infants of other ethnicity (z=-2.752 p=0.006). Kruskal-Wallis Testing revealed differences in QoL scores in infants born to mothers with different levels of education (χ^2 =17.41 p=0.004 DF=5). Post hoc testing revealed infants born to mothers with a degree or postgraduate degree and those who attended higher education had a better QoL due to respiratory symptoms than those born to mothers with A-Levels (p=0.001 and p=0.014) and GCSE's (p<0.001 and p=0.005). Infants born to mothers with A-Levels had a better QoL than those

infants born to mothers with no qualifications (p=0.018) or other qualifications (p=0.038). Those born to mothers with GCSE's had a worse QoL than those born to mothers no qualifications (p=0.047). Infants born preterm had a worse QoL than infants who were born term (z=-2.122 p=0.034). Infants who attended nursery had a worse QoL due to respiratory symptoms than those who did not attend nursery (z=-3.838 p<0.001) and those who lived with other children had a worse QoL than those who did not respiratory by mother children had a worse QoL than those who did not live in a household with other children (z=-2475 p=0.013).

Mothers of male infants reported a worse QoL due to their child's respiratory symptoms than mothers of females (z=-2.697 p=0.007). Likewise, mothers of white infants had a worse QoL due to their child's respiratory symptoms when compared to mothers of infants to other ethnic groups (z=-3.197 p=0.001). Kruskal-Wallis Testing revealed differences in QoL scores in infants born to mothers with different levels of education (χ^2 =19.447 p=0.002 DF=5). Pot hoc testing mothers with A-levels had a worse QoL due to their child's respiratory symptoms than mothers with a degree or post graduate degree (p=0.001), mothers who attended higher education (p<0.001) and mothers with 'other' qualifications (p=0.027). Mothers with a degree or post graduate degree (p=0.014) or who attended higher education (p=0.005). Mothers of infants born preterm had a worse QoL than mothers of infants who were born term (z=-2.054 p=0.012). Mothers of infants who attended nursery had a worse QoL due to their child's respiratory symptoms than mothers of infants born preterm had a worse qoL than mothers of infants who were born term (z=-2.054 p=0.012). Mothers of infants who attended nursery had a worse QoL due to their child's respiratory symptoms than mothers of infants who attended nursery (z=-3.488 p<0.001).

Spearman's rank correlation analysis showed that higher respiratory symptom scores were correlated with higher QoL domain scores indicating a worse QoL ($r_s = 0.744 \text{ p} < 0.001$). This is shown in figure 56.



Figure 56: Scatter diagram showing respiratory symptom scores against QoL Scores at 10 months

3.5.3-Questionnaire 3-16 Months

Cohort Profile, Exposures and Risk Factors at 16 Months

Nursery Attendance, Other Household Children and Sharing a Bedroom

At 16 months 51.6% (n=142) of the cohort attended nursery. 53.8% (n=148) of the cohort lived in a household with another child. Unfortunately a problem with the questionnaire meant that there was a large amount of missing data with regards the number of other household children. 33.1% (n=91) of the cohort shared a bedroom. 72.5% of these shared a bedroom with their parents, 28.6% (n=26) shared with another child and 2.2% (n=1) shared with another family member.

Household Smoking

13.82% (n=38) of the cohort were exposed to ETS in the three months before 16 months of age; see figure 57. 5.3% (n=2) of these smokers smoked inside, 84.2% (n=32) smoked outside, 2.6% (n=1) smoked in the car and 18.4% (n=7) smoked elsewhere. The proportion of the cohort exposed to smoke was significantly lower than the smoking rates in Liverpool in both 2012 (χ^2 =15.90 p<0.05 Figure 57: Household smoking in the LBBS DF=1) and 2015 (χ^2 =12.71 p<0.05 DF=1).



at 16 months

Underlying Medical Conditions

By 16 months old 7.64% (n=21) of the cohort had at least one underlying medical conditions. The underlying medical conditions in the cohort are shown in table 17.

Type of Underlying Condition	Underlying Condition	Frequency
Cardiac (3)	Congenital Heart Disease	3
Gastrointestinal (1)	'Not Described'	1
	Eczema	8
Other (16)	'Not Described'	4
	Allergies	1
	Congenital Talipes Equinovarus	1
	Hip Dysplasia	1
Respiratory (2)	Asthma	1
Respiratory (2)	Chronic Lung Disease	1
Urinary (1)	Vesico-ureteric Reflux	1

Table 15: Underlying medical conditions in the LBBS at 16 months

Total LRSQ Score

At 16 months the lowest total LRSQ score was 0 and the highest score was 92. The mean score was 15.31, the median was 10 and the mode was 6. The distribution of the Totals LRSQ scores at 16 months is shown in figure 58.



Figure 58: Histogram of the total LRSQ scores at 16 months

Total LRSQ Score in Those Expose and Unexposed to Risk Factors and Exposures

Summary Statistics comparing demographics, confounding factors, exposures and risk factors are shown in table 16. Male infants had significantly higher LRSQ scores than females (z=-3.009 p=0.003); see figure 59. Those who attended nursery had a higher LRSQ than those who did not attend nursery (z=-5.272 p<0.001); see figure 60. Paradoxically those who were exposed to ETS in pregnancy but whose mother who didn't smoke had a lower LRSQ score than those who were not exposed at all (z=-1.977 p=0.048); see figure 61. IMD decile of the cohort was established using their postcode. Decile one was the most deprived and decile 10 was the least deprived. Infants in IMD decile nine had a greater LRSQ score than those in deciles one, tow, three and four (p<0.05). Those in decile seven had a higher LRSQ score than those in deciles three and four (p<0.05). Those in decile five had a higher LRSQ sore than those in decile three (p<0.05); see figure 62.

		25%	Median	75%	Mean	Mode
	Male	6	12	24	19.4	6
Infant's Sex*	Female	3	9	19.14	12.07	3
	Single Birth	4	11	21	15.75	6
Multiple Birth	Multiple Birth	3.46	14.86	36.73	17.62	5
	White	4	10.16	21	15.84	6
Ethnicity	Other	6.15	12.5	25.04	15.02	16
	<20	1	1	-	1	1
	20-24	6	7.26	13	9.48	6
	25-29	3	13.26	26.88	21.31	1
Maternal Age	30-34	3	10.87	17	13.12	2
	35-39	3.88	10	21	13.54	5
	40-44	6	10.28	16	11.02	6
	>=45	-	15	-	-	-
	1	3	12	21.27	17.04	1
	2	4	9	16.72	11.33	6
	3	2	5.69	19.53	11.88	0
	4	3	5.95	17.15	9.71	3
	5	5.84	12	31.27	19.53	6
IMD*	6	6	13	24.64	19	1
	7	7 25	14	22.79	17	4
	8	4 39	7 55	15.17	9.58	6
	0	11 18	25.69	22.75	28.62	6
	10	1.88	18 36	55.75	11 25	0
	Degree or Higher	5	0.84	20	12.02	6
	Higher Education	3	10.1	18.41	16.05	1
		5	18.25	24.62	17.65	5
Maternal Qualification	ALEVEI	5	12	24.02	20.02	5
	Other	2.11	17.49	20.39	20.03	2
	Nana	2.11	7	52.03	23.41	2
	None	0.28	/	-	0.70	, ,
Premature Birth	NU	4	10	20.36	15.05	6
	Yes	Б	15	30.27	22.27	13
Low Birth Weight	NO	4	11	21	16	6
	Yes	5	10	18.4	12.48	5
Maternal Smoking In	NO	4	12	21	16.02	6
Freghancy	Yes	3.67	9.18	13	13.11	3
Other Household Smoking In	No	4	12	21.36	16.89	6
Pregnancy*	Yes	3.03	7.09	14.55	9.7	8
Any Smoke Exposure in	No	4	12	22	16.92	6
Pregnancy	Yes	3.58	8	15	11.52	3
Home Smoking *	No	5	12	21	16.61	6
	Yes	3	7.19	16	11.48	3
Breastfeeding	No	2.28	8	21	15.53	6
	Yes	5	11	21	15.86	6
	Never	2.28	8	21	15.53	6
Breastfeeding Time	<1 Month	5	12	24	19.55	13
	1-4 Months	3.81	12	22.14	16.5	0
	>4 Months	5	10	20	13.16	6
Family History of Atopy	No	3	10	19.63	14.99	6
	Yes	5	11	22	16.39	6
Nursery Attendance*	No	2.42	6	16	11.6	6
	Yes	7	15	25	20.26	6
Other Household Children	No	4	10	20	13.19	6
	Yes	4	12	24	18.47	1
Shares a Bedroom	No	4	10	21.2	15.5	6
Shares a Deuroom	Yes	4.87	12	20	16.33	1

 Table 16: Comparison of LRSQ scores between different exposure/risk factor groups in the LBBS at 16 months

*indicates significant result (p<0.05)



Figure 59: Total LRSQ scores at 16 months by infant sex

Figure 60: Total LRSQ score at 16 months by nursery attendance

Figure 61: Total LRSQ scores at 16 months by ETS exposure in pregnancy



Figure 62: Total LRSQ score at 16 months by IMD decile

LRSQ Domain Scores in Months 10-16

The summary statics of the domain scores for months 10-16 are shown in table 18. 'Symptoms with colds' and 'night-time symptoms' were the highest scoring domains. 'Symptoms on increased activity' and 'other respiratory symptoms' were the lowest scoring domains. Domains seven and eight are QoL domains and are discussed elsewhere.

	Min	Median	Max	Mean	SD
Domain 1	0	1	12	21	24
Day Time Symptoms	U	-		2.1	2
Domain 2	0	2	16	2.8	2.8
Night-time Symptoms	U	2	10	2.0	2.0
Domain 3	0	Л	10	17	2 2
Symptoms with Colds	0	4	15	4.7	5.0
Domain 4	0	0	12	٥Q	2 0
Symptoms without Colds	0	U	12	0.5	2.0
Domain 5	0	0	12	0.7	1 0
Symptoms on Increased Activity	0	0	12	0.7	1.5
Domain 6	0	0	10	0.7	16
Other Respiratory Symptoms	0	0	10	0.7	1.0
Domain 7	0	0	10	1 5	2.2
Childs Quality of Life	0	0	12	1.5	2.3
Domain 8	0	0	12	1 0	20
Mothers Quality of Life	0	0	15	1.0	2.0

Table 17: Summary statistics of domain scores at 16 months

Sex

Males had significantly higher scores for 'daytime symptoms' (z=-2.774 p=0.006), 'night-time symptoms' (z=-2.1148 p=0.032), 'symptoms with colds' (z=-3.225 p=0.001) and 'symptoms on increased activity' (z=-3.439 p=0.001) when compared to females.

Maternal Age

Kruskal-Wallis Testing revealed differences in domain scores for infants born to mothers of different maternal ages for 'symptoms without colds' ($\chi^2 = 16.22 \text{ p} < 0.05 \text{ DF} = 6$). Post hoc testing revealed Infants born to mothers aged 25-29 years had higher scores than infants born to mothers aged 35-39 years (p=0.011) but lower scores than those infants born to mothers aged 40-44 (p=0.006). Infants born to mothers aged 45 or older scored lower than those aged 40-44 years (p=0.04).

IMD

Kruskal-Wallis Testing revealed differences in domain scores for infants living in different IMD deciles for 'day time symptoms' (χ^2 =18.196 p=0.033 DF=6) and 'symptoms without colds' (χ^2 =23.557 p=0.005 DF=6). Post hoc analysis showed 'day time symptoms' were higher in those

who lived in IMD deciles one (p=0.043), five (p= 0.034), six (p=0.017), seven (p=0.025) and eight (p=0.011) compared to those who lived in IMD decile three. Likewise those who lived in IMD decile nine had higher 'day time symptoms' scores than those who live in IMD decile one (p=0.025), two (p=0.007) and three (p=0.11) and four (p=0.002). For 'symptoms without colds' those who lived in IMD decile four had lower scores than those who lived in IMD decile one (p=0.046) and five (p=0.034). Those who lived in decile nine had higher scores than those who lived in deciles one to four (p=0.035, p=0.005, p=0.013, p=0.004 respectively).

Maternal Education

Kruskal-Wallis Testing revealed differences in domain scores for infants born to mothers with different levels of education for 'other respiratory symptoms' (χ^2 =16.162 p=0.006 DF=5). Post hoc testing showed infants born to mothers with 'other' qualifications had higher scores than: infants born to mothers with a degree or postgraduate degree (p=0.007), infants born to mothers who attended higher education (p=0.001) and infants born to mothers who had GCSE's (p=0.002). Infants born to mothers who had attended higher education and those born to mothers with only GCSE's had lower scores than infants born to mothers with A-levels (p=0.028 and p=0.031).

Preterm Birth

Infants who were born preterm had higher domain scores than term babies for 'night-time symptoms' (z=-2.466 p=0.014), 'symptoms with colds' (z=-2.049 p=0.04), and for 'other respiratory symptoms' (z=-3.006 p=0.003).

Smoking Exposure

Those infants who were exposed to ETS in pregnancy from other household smokers had lower scores than those who were not exposed for the domains 'symptoms without colds' (z=-2.2882 p=0.004) and 'symptoms on increased activity' (z=-3114 p=0.002). Those who were exposed to any tobacco smoke during pregnancy whether that be maternal smoking or other household smokers showed the same result (z=-2.838 p=0.005 and z=-3.122 p=0.002). Those who were exposed to tobacco smoke exposure in the last six months showed lower domain scores for 'daytime symptoms' (z=-2.371 p=0.018), 'symptoms with colds' (z=-2.531 p=0.011) and 'symptoms on increased activity' (z=-2.118 p=0.034) than those who were not exposed.

Breastfeeding

Those who were breastfed showed lower scores on 'increased activity' than those who were not breastfed (z=-2.202 p=0.028).

Family History of Atopy

Those with a family history of atopy scored higher for 'symptoms without colds' (z=-2.18 p=0.029) and 'symptoms on increased activity' (z=-2.255 p=0.029) compared to infants without and family history of atopy.

Nursery Attendance

Those who attended nursery had higher scores for 'daytime symptoms' (z=-4.169 p<0.001), 'night-time symptoms' (z=-4.609 p<0.001), 'symptoms with colds' (z=-5.706 p<0.001), 'symptoms without colds' (z=-3.117 p=0.002) and 'other respiratory symptoms' (z=-3.649 p=0.012) compared to those who did not attend nursery.

Respiratory Symptom Prevalence in Months 10-16

At 16 months of life 96.0% (n=264) of the cohort had respiratory symptoms in the previous three months. Colds were the highest reported respiratory symptom with 91.3% (n=251) of the cohort having at least one cold in the previous three months. The lowest reported respiratory symptom was 'noisy breathing from the back of the throat' with only 16.4% (n=45) of the cohort reporting 'noisy breathing from the back of the throat'. The prevalence of the respiratory symptoms in the previous three months per 100 infants are shown in figure 63.



Figure 63: Respiratory symptom prevalence in the LBBS at 16 months

Respiratory symptom prevalence in Those Exposed and Unexposed to Risk Factors

Maternal Age

Kruskal-Wallis testing showed there was variation in the prevalence of cough (χ^2 =16.963 p<0.05 DF=6), rattily chest (χ^2 =13.05 p<0.05 DF=6), and 'noisy breathing not from the chest' (χ^2 =13.01 p<0.05 DF=6) in infants born to mothers of different ages. Post hoc analysis showed that infants born to mothers aged less than 20 had a lower prevalence of cough than those born to mothers ages 20-24 (χ^2 =17 Fisher's exact test p=0.007 DF=1), 35-39 (χ^2 =5.07 Fisher's exact test p=0.039 DF=1) and 40-44 (χ^2 =8.11 Fisher's exact test p=0.016 DF=1). Those born to mothers aged 20-24 had a higher prevalence of cough than those born to mothers aged 20-24 had a higher prevalence of cough than those born to mothers aged 20-24 had a higher prevalence of cough than those born to mothers 30-34 (χ^2 =5.32 Fisher's exact test p=0.020 DF=1). Infants born to mothers aged 25-29 had a higher prevalence of rattily chest than infants born to mothers ages 35-39 (χ^2 =5.64 p<0.05 DF=1) and 40-44 (χ^2 =7.81 p<0.05 DF=1) and infants born to mothers aged 30-34 had a higher prevalence of rattily chest than those born to mothers aged 40-44 (χ^2 =4.79 p<0.05 DF=1). Infants born to mothers aged 25-29 had a higher prevalence of 'noisy breathing not from the chest' than infants born to mothers aged 30-34 (χ^2 =6.00 p<0.05 DF=1).

Maternal Qualification

Kruskal-Wallis testing showed there was variation in the prevalence of 'noisy breathing not from the chest' (χ^2 =17.64 p<0.05 DF=5) and rapid breathing (χ^2 =14.51 p<0.05 DF=5) in infants born to mothers with different levels of education. Post hoc analysis revealed infants born to mothers who attended higher education had a lower prevalence of 'noisy breathing not from the chest' than those with A-levels (χ^2 =6.52 Fisher's Exact Test p=0.030 DF=1) and those with no education (χ^2 =8.69 Fisher's Exact Test p=0.003 DF=1). Likewise, those with A-levels had a lower prevalence than those with GCSE's (χ^2 =8.52 Fisher's Exact Test p=0.006 DF=1) and those born to mothers with GCSE's had a lower prevalence than those with no education (χ^2 =6.23 Fisher's Exact Test p=0.039 DF=1) or other qualifications (χ^2 =12.24 Fisher's Exact Test p=0.020 DF=1). Those born to mothers with other qualifications had a higher prevalence of rapid breathing than those born to mothers with a degree or post graduate degree (χ^2 =11.53 Fisher's Exact Test p=0.003 DF=1), those who attended higher education (χ^2 =12.29 Fisher's Exact Test p=0.023 DF=1), those with A-levels (χ^2 =5.98 Fisher's Exact Test p=0.031 DF=1) and those with GCSE's (χ^2 =5.54 Fisher's Exact Test p=0.044 DF=1).

Smoking Exposure

Those infants who's mothers smoked in pregnancy had a higher prevalence of snoring compared to those whose mother did not smoke (χ^2 =11.56 p<0.05 DF=1). Infants that were exposed to any smoking in pregnancy whether that be maternal or other household tobacco

smokers had a lower prevalence of SoB (χ^2 =4.17 p<0.05 DF=1) but a higher prevalence of snoring (χ^2 =4.84 p<0.05 DF=1). Those infants exposed to tobacco smoke in the last six months had a higher prevalence of snoring than those who were not exposed to tobacco smoke (χ^2 =7.35 p<0.05 DF=1).

Gestation

Infants that were born preterm had a higher prevalence of snoring (χ^2 =7.64 p<0.05 DF=1), 'noisy breathing from the throat' (χ^2 =6.26, Fisher's exact test p<0.024 DF=1) and 'noisy breathing not from the chest' (χ^2 =8.14 Fisher's exact test p=0.010 DF=1).

Breastfeeding

Infants who were breastfed for any duration of time had a lower prevalence of cough than infants who were not breastfed (χ^2 =11.92 p<0.05 DF=1).

Nursery Attendance

Infants who attended nursery had a higher prevalence of cough (χ^2 =13.13 p<0.05 DF=1), rattily chest (χ^2 =12.93 p<0.05 DF=1), snoring (χ^2 =5.55 p<0.05 DF=1), 'noisy breathing from the throat' (χ^2 =7.47 p<0.05 DF=1), 'noisy breathing not from the chest' (χ^2 =8.43 p<0.05 DF=1) and rapid breathing (χ^2 =6.74 p<0.05 DF=1) than those infants who did not attend nursery.

Health Care Service Attendance in Months 10-16

In the three month prior to being 16 months old 33.09% (n=91) of the cohort attended health care services with respiratory symptoms. 6.55% (n=18) attended the hospital with respiratory symptoms. GP attendance was higher in males (χ^2 =10.61 p<0.05 DF=1), infants with a family history of atopy (χ^2 =4.61 p<0.05 DF=1 and infants who attended nursery (χ^2 =15.06 p<0.05 DF=1). There was no clear trends between health care attendance and maternal age or maternal education at 16 months therefore these graphs have not been shown.

The Impact of Respiratory Symptoms on QoL in Months 10-16

The impact of respiratory symptoms on the QoL of the infant and their mothers are summarised in the figures 64 and 65. The QoL of 45.8% (n=126) of the infants in the cohort was effected by their respiratory symptoms and 49.1% (n=135) of the cohort's mothers were effected by their infants' respiratory symptoms.



Figure 64: The impact of respiratory disease on the infant's QoL 16 months in the LBBS (per 100 infants)



Figure 65: The impact of respiratory disease on the infant's mother's QoL at 16 months in the LBBS (per 100 infants)

Mothers reported that males had a worse QoL due to their respiratory symptoms than females (z=-2.99 p=0.003). Infants who attended nursery had a worse QoL due to respiratory symptoms than those who did not attend nursery (z=-3.699 p<0.001). Inexplicably at this time point, infants who were exposed to tobacco smoke in pregnancy whether that be maternal or from other household smokers had a better QoL reported than those who were not exposed to tobacco smoke in pregnancy (z=-2.047 p=0.041).

Mothers of male infants reported a worse QoL due to their child's respiratory symptoms than mothers of females (z=-2.711p=0.007). Mothers of infants who attended nursery had a worse QoL due to their child's respiratory symptoms than mothers of infants who did not attend nursery (z=-3.435 p=0.001). Mothers who live in households with more than one child reported a worse QoL due to their child's respiratory symptoms than mothers than mothers who lived with only one child (z=-2.929 p=0.003).

Spearman's rank correlation analysis showed that higher respiratory symptom scores were correlated with higher QoL domain scores indicating a worse QoL (r_s =0.78 p<0.001). This is shown in figure 66.





Figure 66: Scatter diagram of respiratory symptom score and quality of life score at 16 months
3.5.4-Questionnaire 4-22 Months

Cohort Profile, Exposures and Risk Factors at 22 Months

Nursery Attendance, Other Household Children and Sharing a Bedroom

At 22 months of age 56.81% (n=146) of the cohort attended nursery. 49.42% (n=127) of the cohort lived in households with other children. Unfortunately a technical problem with the questionnaire meant that there was a large amount of missing data with regards the question number of other household children and this detail cannot be reported with confidence.27.63% (n=71) of the cohort shared a bedroom at 22 months of age. 64.8% (n=46) of these infants shared their bedroom with their parents, 39.4% (n=28) of these shared with at least one other child and 7.04% (n=5) shared with another family member.

Household Smoking

15.18% (n=39) of the cohort were exposed to household tobacco smoke in months 16-22; see figure 67. 5.1% (n=2) of these smokers smoked inside the home, 92% (n=36) smoked outside, 5.1% (n=2) smoked in the car and 10.3% (n=4) smoked elsewhere. These smoking rates were significantly lower than the smoking prevalence in Liverpool in both 2012 (χ^2 = 11.55 Figure 67: Household smoking in the LBBS p<0.05 DF=1) and 2015 (χ^2 = 8.89 p<0.05 DF=1).



cohort at 22 months

Underlying Medical Conditions

By 22 months, 15.18% (n=39) of the children in the cohort had underlying medical conditions. The underlying health conditions in the LBBS at 16 months are shown table 18.

Type of Underlying Condition	Underlying Condition	Frequency
Cardiac (3)	Congenital Heart Disease	3
Gastrointestinal (1)	Not otherwise specified	1
Neurological (2)	Hypotonia	1
	Not otherwise specified	1
	Eczema	18
	Not otherwise specified	4
	Allergies	2
Other (28)	Congenital Talipes Equinovarus	1
	Developmental Delay	1
	Hip Dysplasia	1
	Trisomy 21	1
Respiratory (6)	Asthma	5
	Bronchomalacia	1
Urinary (2)	Vesico-ureteric Reflux	1
	Tubular Sclerosis	1

Table 18: Underlying medical conditions in the LBBS at 16 months

Total LRSQ Score

At 22 months the lowest total score was 0 and the highest score was 68. The mean score was 14.1, the median was eight and the mode was 5. The distribution of the Totals LRSQ scores at 22 months is shown in the figure 68.





Total LRSQ Score in Those

Expose and Unexposed to Risk Factors and Exposures

Summary statistics comparing demographics, confounding factors, exposures and risk factors are shown in table 19. Those infants born to mothers with a degree or post graduate degree had higher LRSQ scores than those born to mothers who only attended higher education (p=0.003) and had GCSE's (p<0.001). Those born to mothers who attended higher education had lower score than those born to mothers who had no qualifications

		25%	Median	75%	Mean	Mode
Infant's Sov	Male	3	6.4	15.94	11.47	4
mant's sex	Female	4	8	20.08	14.54	5
Multiple Dirth	Single Birth	4	8	17	13.18	5
	Multiple Birth	3	13.24	19.11	11.99	3
Ethnicity	White	3	13.24	19.11	11.99	3
Ethnicity	Other	5.54	12	20	13	1
	<20	-	-	-	-	-
	20-24	6.3	10	17	12.43	10
	25-29	3	5	13	10.86	4
Maternal Age	30-34	4	7	17.79	14.61	0
	35-39	4.85	8	16.73	13.43	8
	40-44	3.42	14	20.58	13.78	14
	>=45	-	-	-	-	-
	1	4	88	17	13.18	0
	2	2.04	7.92	18.14	11.59	2
	3	37	7	14.85	10.8	0
	4	3.01	6	9	7.08	6
	5	3.38	7.76	18.71	14.03	1
IMD	6	3.45	12.62	28.4	15.05	1
	7	5.12	11.87	28.27	20.26	1
	8	5.12	11.87	28.27	20.26	1
	9	4.09	5.33	21.37	10.71	5
	10	13	50.43		24.25	0
	Dearee or Hiaher	5	11	25	16.32	5
	Higher Education	1	7	10.68	8.71	1
Matornal	Alevel	3	10.75	16.73	10.82	3
Qualification*	GCSE	2	4	10	6.95	0
	Other	2 54	10	13.64	10 53	10
	None	7.45	15.3	35.27	20.41	6
	No	4	8	17	13 35	0
Premature Birth	Yes	3 27	6.2	18.21	10.82	2
	No	3.27	8	17.47	13 /1	5
Low Birth Weight	Ves		6.13	13.17	8 33	2
	No	1	8	13.17	13.16	0
Pregnancy	Ves	1	8.67	26	12.95	1
Other Household	No	2	7.82	16	12.55	5
Smoking In	Kee	5	7.02	25.5	12.15	3
Pregnancy*	Yes	5	13	25.5	17.33	U
Any Smoke Exposure	No	4	8	17.07	12.52	5
in Pregnancy	Yes	3	10.33	17.63	15.33	1
Home Smoking	No	4	8	18	13.61	0
	Yes	3	5.52	16.6	10.21	1
Breastfeeding	No	3	8	14	11.26	0
	Yes	4	8	18	13.78	0
	Never	3	8	14	11.26	0
Breastfeeding Time	<1 Month	2	7	11.76	13.62	1
	1-4 Months	4	14	24.27	16.43	4
	>4 Months	4	8	16.36	12.66	5
Family History of	No	5	10	19.1	13.93	8
Atopy	Yes	3	6.96	16	12.54	4
Nursery Attendance*	No	2	7.96	15.9	11.11	0
	Yes	5	9	19.79	14.98	5
Other Household	No	4	8	17	13.67	5
Children	Yes	3	8	17.76	12.63	1
Shares a Podroom	No	4	8	17.63	13.81	4

Table 19: Comparison of LRSQ scores between different exposure/risk factor groups in the LBBS at 22 months

*indicates significant result (p<0.05)

(p=0.01). Those born to mothers with A-Levels had higher scores that those born to mothers with GCSE's (p=0.045) however those born to mothers with GCSE's had lower scores than infants born to those with no qualifications (p=0.005). This is shown in figure 69. Those who were exposed to smoking in pregnancy but whose mother who didn't smoke had a higher LRSQ score than those who were not exposed (z=-2.102 p=0.036); see figure 70. Those who attended nursery had higher LRSQ scores than those who did not attend nursery (z=-1.975, p=0.048); see figure 71.



Maternal Education

Figure 69: Total LRSQ scores at 22 months by maternal education



Figure 70: Total LRSQ score at 22 months by other household smoking in pregnancy

Figure 71: Total LRSQ scores at 22 months by nursery attendance

LRSQ Domain Scores in Months 16-22

The summary statics of the domain scores for months 16-22 are shown in the table 20. 'Symptoms with colds' and 'night-time symptoms' were the highest scoring domains. 'Symptoms on increased activity' and 'other respiratory symptoms' were the lowest scoring domain. Domains seven and eight are QoL domains and are discussed elsewhere.

	Min	Median	Max	Mean	SD
Domain 1 Day Time Symptoms	0	1	11	2.1	2.3
Domain 2 Night-time Symptoms	0	2	13	2.9	2.7
Domain 3 Symptoms with Colds	0	3	21	4.4	4.1
Domain 4 Symptoms without Colds	0	0	11	0.7	1.6
Domain 5 Symptoms on Increased Activity	0	0	10	0.7	1.7
Domain 6 Other Respiratory Symptoms	0	0	9	0.6	1.4
Domain 7 Childs Quality of Life	0	0	16	1.4	2.4
Domain 8 Mothers Quality of Life	0	0	16	1.5	2.5

Table 20: Summary statistics of domain scores at 22 months

Sex

Females had significantly higher 'symptoms with colds' (z=-2.22 p=0.027) and 'symptoms without colds' than males (z=-1.99 p=0.047).

Ethnicity

Infants of other ethnicity had higher scores for 'symptoms without colds' (z=-2.27 p=0.023) and 'symptoms on increased activity' (z=-3.55 p<0.001) than white infants.

Maternal Education

Kruskal-Wallis Testing revealed differences in domain scores for infants born to mothers with different levels of education for 'daytime symptoms' (χ^2 =22.76 p<0.001 DF=5), 'night-time symptoms' (χ^2 =25.74 p<0.001 DF=5), 'symptoms with colds' (χ^2 =16.67 p=0.005 DF=5), 'symptoms without colds' (χ^2 =13.31 p=0.021 DF=5) and 'symptoms on increase activity' (χ^2 =11.49 p=0.042 DF=5).

Post Hoc Testing Domain 1- Day Time Symptoms

Infants born to mothers with a degree or a post graduate degree scored higher than infants born to mothers who attended higher education (p=0.011) or had GCSE's (p<0.001). Infants born to mothers with no qualifications had higher scores than those born to mothers who attended higher education (p=0.005), had A-levels (p=0.033) or GCSE's (p<0.001).

Post Hoc Testing Domain 2- Night-time Symptoms

Infants born to mothers with a degree or a post graduate degree scored higher than infants born to mothers who attended higher education (p=0.023). Infants born to mothers with GCSE's had lower scores than infants born to mothers with a degree or post graduate degree (p<0.001), attended higher education (p=0.013), had A-levels (p=0.007), had other qualifications (p=0.031) or no qualifications (p=0.002).

Post Hoc Testing Domain 3- Symptoms with Colds

Infants born to mothers with a degree or a post graduate degree scored higher than infants born to mothers who attended higher education (p=0.003) or had GCSE's (p=0.003). Those with no qualifications had higher scores than those who attended higher education (p=0.034) and those with GCSE's (p=0.019).

Post Hoc Testing Domain 4- Symptoms without Colds

Infants born to mothers with a degree or a post graduate degree scored higher than infants born to mothers who attended higher education (p=0.006) or had GCSE's (p=0.017). Those with no qualifications had higher scores than those who attended higher education (p=0.046) and those with GCSE's (p=0.004).

Post Hoc Testing Domain 5- Symptoms on Increased Activity

Infants born to mothers with a degree or a post graduate degree scored higher than infants born to mothers who attended higher education (p=0.022) or had GCSE's (p=0.042).

Smoking in Pregnancy

Those who were exposed to other household tobacco smoke in pregnancy but not maternal tobacco smoke had higher scores for 'daytime symptoms' (z=-2.528 p=0.011) and 'symptoms with colds' (z=-1.97 p=0.049).

Household Smoking

Those who were not exposed to household smoking in the last six months had higher scores for 'other respiratory symptoms' then those who were exposed (z=-2.009 p=0.045).

Breastfeeding

Those who were breastfed had higher scores for 'night-time symptoms' than those who were not breastfed (z=-1.967 p=0.049). Further analysis showed that this difference was between those who were breastfed for between one and four months and those who were never breastfed (p=0.003). Those breastfed for greater than four months did not have significantly higher scores (p=0.192).

Nursery Attendance

Those who attended nursery had higher scores for 'night-time symptoms' (z=-2.956 p=0.003) and 'symptoms with colds' (z=-1.964 p=0.049) than those who did not attend nursery.

Other Household Children

Those who lived with other household children had lower scores on 'increased activity' than those who did not have other household children (z=-2.218 p=0.027).

Respiratory Symptom Prevalence in Months 16-22

93.8% (n=241) of the cohort had respiratory symptoms in the previous three months at 22 months of age. Colds were the highest reported respiratory symptom with 85.6% (n=220) of the cohort having at least one cold in the previous three months. The lowest reported respiratory symptom was 'noisy breathing not from the chest' with only 9.34% (n=24) of the cohort reporting 'noisy breathing not from the chest'. The prevalence of the respiratory symptoms in the previous three months per 100 infants are shown in figure 72.



Figure 72: Respiratory symptom prevalence at in the three months prior to 22 months old in the LBBS

Respiratory symptom prevalence in Those Exposed and Unexposed to Risk Factors

Sex

Females had higher prevalence of wheeze (χ^2 =4.85 p<0.05 DF=1) and colds (χ^2 =5.38p<0.05 DF=1) than males at 22 months of age.

Maternal Age

Kruskal-Wallis testing showed there was variation in the prevalence of colds (χ^2 =13.15 p<0.05 DF=5) in infants born to mothers of different ages. Post hoc analysis showed that infants born to mothers aged 20-24 had a higher prevalence of colds than those born to mothers aged 25-29 (χ^2 =4.62 Fisher's Exact Test p=0.0495 DF=1). Those born to mothers aged 35-39 had a higher prevalence of colds than those born to mothers aged 40-44 (χ^2 =5.46 Fisher's Exact Test p=0.033 DF=1). Those born to mothers aged 25-29 and 30-34 had a lower prevalence of colds than those infants born to mothers aged 35-39 (χ^2 =10.09 p<0.05 DF=1 and χ^2 =5.83 p<0.05 DF=1 respectively).

Maternal Qualification

Kruskal-Wallis testing showed there was variation in the prevalence of wheeze (χ^2 =12.06 p<0.05 DF=5), cough (χ^2 =13.62 p<0.05 DF=5), rattily chest (χ^2 =18.79 p<0.05 DF=5) and snoring (χ^2 = 17.15 p<0.05 DF=5) in infants born to mothers with different levels of education. Post hoc analysis revealed infants born to mothers with a post graduate degree had a higher prevalence of wheeze, cough, rattily chest and snoring than those who attended higher education (χ^2 = 5.28 p<0.05 DF=1, χ^2 = 8.54 p<0.05 DF=1, χ^2 =7.07 p<0.05 DF=1 and χ^2 =4.13 p<0.05 DF=1 respectively) and those with GCSE's (χ^2 =5.47 p<0.05 DF=1, χ^2 =6.32 p<0.05 DF=1, χ^2 =10.91 p<0.05 DF=1 and χ^2 =7.89 p<0.05 DF=1 respectively). Those with no qualifications had a higher prevalence of wheeze, rattily chest and snoring than those with GCSE's (χ^2 = 5.37 Fisher's Exact Test p=0.040 DF=1, χ^2 = 5.63 Fisher's Exact Test p=0.037 DF=1 and χ^2 =10.0 Fisher's Exact Test p=0.003 DF=1 respectively). Those with infants born to mothers who attended higher education and those with GCSE's had a lower prevalence of rattily chest than those who had 'other' qualifications (χ^2 =5.97 Fisher's Exact Test p=0.030 DF=1 and χ^2 =9.78 Fisher's Exact Test p=0.004 DF=1 respectively). Infants born to mothers with no education had a higher prevalence of snoring than those who attended higher education (χ^2 =7.53 Fisher's Exact Test p=0.006 DF=1) as did those born to mothers with Alevels compared to those born to mothers with GCSE's (χ^2 =4.86 p<0.05 DF=1).

Smoking Exposure

Unexpectedly in this cohort those exposed to tobacco smoke in the last six months had a lower prevalence of SoB than those who were not exposed (χ^2 =4.52 p<0.05 DF=1).

Nursery Attendance

Those who attended nursery had a higher prevalence of cough (χ^2 =9.20 p<0.05 DF=1), SoB (χ^2 =5.08 p<0.05 DF=1), snoring (χ^2 =8.95 p<0.05 DF=1) and colds (χ^2 =9.28 p<0.05 DF=1) than those who did not attend nursery in the last six months.

Health Care Service Attendance at 22 Months

23.74% (n=61) of the cohort attended the GP in the previous three months at 22 of life with respiratory symptoms. 5.45% (n=14) attended hospital with respiratory symptoms. GP attendance was increased in those who were exposed to tobacco smoking in pregnancy but not maternal smoking (χ^2 =4.32. p<0.05 DF=1). Health care use in different maternal age categories and maternal education categories are shown in figures 73 and 74.



Figure 73: Percentage of the cohort that attended their GP or hospital with respiratory symptoms at 22 months of life by maternal age group



Figure 74: Percentage of the cohort that attended their GP or hospital with respiratory symptoms at 22 months of life by highest maternal qualification attained

The Impact of Respiratory Symptoms on QoL in Months 16-22

The impact of respiratory symptoms on the QoL of the infant and their families in months 16-22 are summarised in figures 75 and 76. The QoL of 41.2% (n=106) of the infants in the cohort was affected by their respiratory symptoms and 40.9% (n=105) of the cohort's mothers were affected by their infant's respiratory symptoms.



Figure 75: The impact of respiratory disease on the infant's QoL at 22 months in the LBBS (per 100 infants)



Figure 76: The impact of respiratory disease on the infant's mother's QoL at 22 months in the LBBS (per 100 infants)

Kruskal-Wallis testing revealed differences in QoL scores of infants born to mothers with different levels of education (χ^2 =19.69 p=0.001 DF=5). Post hoc testing revealed those with a degree or postgraduate degree had a worse QoL due to respiratory symptoms than those born to mothers who attended higher education (p=0.003) had GCSE's (p=0.009) and other

qualifications (p=0.049). Those infants born to mother with no qualifications had a worse QoL than those infants born to mothers with other qualifications (p=0.015), GCSE's (p=0.006), and those who attended higher education (p=0.006).

Kruskal-Wallis testing revealed differences in QoL scores of the families of infants born to mothers with different levels of education (χ^2 =19.69 p=0.001 DF=5). Post hoc testing revealed those families with infants born to mothers with a degree or postgraduate degree had a worse QoL due to respiratory symptoms than those born to mothers who attended higher education (p=0.003) or had GCSE's (p=0.009). Those families with infants born to mothers with no qualifications had a worse QoL than those infants born to mothers with other qualifications (p=0.015), GCSE's (p=0.006), A-levels (p=0.046) and those who attended higher education (p=0.006). Those families with infants born to mothers with A-levels had a worse QoL than those who had GCSE's (p=0.024) and those who had attended higher education (p=0.044).

Spearman's rank correlation analysis showed that higher Respiratory symptom scores were correlated with higher QoL domain scores indicating a worse QoL (r_s =0.756 p<0.001). This is shown in figure 77.



Figure 77: Scatter diagram of respiratory symptom score and quality of life score at 22 months

3.6-Longitudinal Analysis

The longitudinal analysis of the LBBS data is shown below. Longitudinal analysis was performed by multilevel mixed effects model analysis. The final multi-level model had three levels; 1.IMD decile, 2.Maternal Education, 3.Questionnaire. The development of this final model is described in section 2.2.8. The final multi-level models and their coefficients and significance levels are shown in the tables below. The longitudinal analysis has allowed the following aim of this thesis to be achieved;

• To determine the change in the LRSQ scores over time and the variability in this change between different population groups in the LBBS.

3.6.1-Total LRSQ Score

Nursery attendance, being male and living in a household with other children increased total LRSQ score over time. Sharing a bedroom and breastfeeding for on increasing amount of time, up to four months of age, decreased total LRSQ scores over time. See table 21.

Exposure	Increase/Decrease Score	Coefficient	95% CI	р
Nursery	Increase	5.44	3.75-7.13	<0.001
Infant's Sex	Boys > Girls	-2.02	-3.430.62	0.005
Sharing a Bedroom	Decrease	-1.75	-3.210.28	0.020
Increasing Breastfeeding Time	Decrease	-0.75	-1.340.16	0.013
Other Household Children	Increased	2.11	0.67-3.56	0.004

Table 21: Multi-level mixed e	effects model for	Total LRSQ scores
-------------------------------	-------------------	--------------------------

3.6.2-Day Time Symptoms

Attending nursery and being male increased day time symptom score over time. Breastfeeding for an increasing amount of time, up to four months of age, decreased day time symptom score over time. See table 22.

Table 22: Multi-level	mixed effects	model for	domain 1 (dav time	symptoms)

Exposure	Increase/Decrease Score	Coefficient	95% Cl	р
Nursery	Increase	0.91	0.66-1.17	<0.001
Infant's Sex	Boys > Girls	-0.31	-0.530.10	0.004
Increasing Breastfeeding Time	Decrease	-0.11	-0.200.018	0.018

3.6.3-Night-time Symptoms

Attending nursery, being male, increasing number of household children and increasing birth weight increase night-time symptom score over time. Breastfeeding for an increasing amount of time, up to four months of age, decreased night-time symptom score over time. See table 23.

Exposure	Increase/Decrease Score	Coefficient	95% Cl	р
Nursery	Increase	1.03	0.73-1.32	0.001
Infant's Sex	Boys > Girls	-0.43	-0.670.18	<0.001
# Household	Increase	0.27	0.10-0.43	0.002
Children				
Birth Weight	Increase	0.00020	0.0001-0.0004	0.038
Increasing	Decrease	-0.10	-0.20-0.008	0.071
Breastfeeding Time				

Table 23:Multi-level mixed effects model for domain 2 (night-time symptoms)

3.6.4-Symptoms with Colds

Nursery attendance, being male, living with other household children and increasing birth weight increased symptoms with colds over time. Sharing a bedroom reduces symptom with colds over time. See table 24.

Table 24:Multi-level	mixed effects	s model for	domain 3	(symptoms	with colds)
				(0)	

Exposure	Increase/Decrease Score	Coefficient	95%CI	р
Nursery	Increase	1.68	1.25-2.11	<0.001
Infant's Sex	Boys > Girls	-0.32	-0.670.03	0.077
Shares a Bedroom	Decrease	-0.72	-1.090.35	<0.001
Other Children	Increase	0.47	0.11-0.83	0.011
Birth Weight	Increase	0.0001	-0.00003-	0.079
			0.0005	

3.6.5-Symptoms without Colds

Nursery attendance and being born preterm increased symptoms without colds over time. Breastfeeding for an increasing amount of time, up to four months of age, decreases symptoms without colds over the first 22 months of life See table 25.

Exposure	Increase/Decrease Score	Coefficient	95% CI	р
Increasing Breastfeeding Time	Decrease	-0.14	-0.210.07	<0.001
Nursery	Increase	0.25	0.06 - 0.43	0.011
Preterm Birth	Increase	0.18	-0.03- 0.39	0.099

Table 25:Multi-level mixed effects model for domain 4 (symptoms without colds)

3.6.6-Symptoms on Increased Activity

Nursery attendance, being male, and a family history of atopy increased symptoms on increased activity scores over time. Breastfeeding for an increasing amount of time, up to four months of age, and increasing gestational age decreased symptoms on increased activity scores over time. See table 26.

Exposure	Increase/Decrease Score	Coefficient	95% CI	р
Nursery	Increase	0.33	0.15-0.52	<0.001
Increasing Breastfeeding Time	Decrease	-0.10	-0.170.03	0.003
Infant's Sex	Boys > Girls	-0.21	-0.370.06	0.08
Gestation	Decrease	-0.03	-0.062- 0.004	0.082
FH Atopy	Increase	0.14	-0.02-0.30	0.079

Table 26: Multi-level mixed effects model for domain 5 (symptoms on increased activity)

3.6.7-Other Respiratory Symptoms

Being male increased other respiratory symptoms scores over time. Breastfeeding, for any amount of time, and increasing gestational age decreased other respiratory symptoms scores over time. See table 27.

Table 27: Multi-level mixed effects model for domain 6 (other respiratory sym	ptoms)
---	--------

Exposure	Increase/Decrease Score	Coefficient	95% CI	р
Infant's Sex	Boys > Girls	-0.21	-0.380.038	0.016
Breastfeeding	Decrease	-0.07	-0.14- 0.003	0.059
Gestation	Decrease	-0.03	-0.07-0.0009	0.057

3.6.8-Childs QoL

Nursery attendance and increasing number of household children decreased the infant's QoL due to respiratory symptoms over time. Sharing a bedroom, breastfeeding for an increasing

amount of time, up to four months of age, and unexpectedly, household tobacco smoke exposure improved the reported infant's QoL over time. See table 28.

Exposure	Increase/Decrease Score	Coefficient	95% CI	р
Nursery	Increase	0.63	0.40-0.87	<0.001
# Household Children	Increase 0.38		0.24-0.50	<0.001
Sharing a Bedroom	Decrease	-0.31	-0.510.10	0.004
Increasing Breastfeeding Time	Decrease	-0.09	-0.180.01	0.023
Home Smoking	Decrease	-0.34	-0.600.08	0.011

 Table 28:Multi-level mixed effects model for domain 7 (infant's quality of life- higher score indicates worse quality of life)

3.6.9-Family QoL

Nursery attendance, other household children, being male, and a FH of atopy decreased the parents' QoL over time. Parents to infants who were breastfed for an increasing amount of time, up to four months of age, and shared a bedroom had an improved QoL over time. See table 29.

Table 29:Multi-level mixed effects model for domain 8 (mother's quality of life- higher score indicates worse quality of life)

Exposure	Increase/Decrease Score	Coefficient	95% CI	р
Nursery	Increase	0.62	0.33-0.90	<0.001
Other Household Child	Increase	0.42	0.17-0.67	0.001
Infant's Sex	Boys > Girls	-0.29	-0.530.05	0.019
Increasing Breastfeeding Time	Decrease	-0.11	-0.210.007	0.035
FH Atopy	Increase	0.25	0.004-0.49	0.046
Shares Bedroom	Decrease	-0.25	-0.50- 0.005	0.055

3.7-Questionnaire Validation

This section describes the validation of the LRSQ in the LBBS. This allowed the following aim of this thesis to be achieved;

• To validate the LRSQ in a longitudinal study.

The internal consistency of the LRSQ was assessed using Cronbach's alpha coefficients. The questionnaire demonstrated acceptable to good internal consistency. Night-time symptoms and 'other respiratory symptoms' showed acceptable internal consistency (α >0.07) and the remaining domains showed good internal consistency (α >0.8). None of the domains showed excellent internal consistency (α >0.9). The internal consistency of each of the domain is shown in table 30.

Domain	Cronbach Alpha	Internal Consistency
1- Day Time Symptoms	0.81	Good
2- Night-time Symptoms	0.77	Acceptable
3- Symptoms with Colds	0.86	Good
4- Symptoms without Colds	0.81	Good
5- Symptoms on Increased Activity	0.85	Good
6- Other respiratory symptoms	0.78	Acceptable
7- Child's QoL	0.85	Good
8- Family's QoL	0.86	Good

Table 30: Cronbach's alpha coefficients for the LRSQ domains in the LBBS cohort

Chapter 4- Discussion

The aims of the current thesis were to;

- To create a profile of the infants and mothers enrolled in the LBBS and make a comparison with the population of Liverpool and those born in the LWH.
- To describe respiratory symptoms of preschool children using the LRSQ from birth until 22 months of age.
- To determine the differences in LRSQ scores in the different populations at each questionnaire time point.
- To determine the change in the LRSQ scores over time and the variability in this change between different population groups in the LBBS.
- To validate the LRSQ in a longitudinal study.

These aims have been achieved by analysing the LBBS data using the methods describes in section 2.2. The results found in chapter three are discussed below. The results are summarised and then discussed in line with previous findings. Unexpected results and the potential reasons for them are discussed in detail.

4.1-Recruitment and Response Rates

Recruitment was performed by MPhil students in the LWH and by respiratory physicians during CF diagnosis consultations. The use of face-to-face recruitment in the LBBS, rather than using posters and advertising alone, is likely to have increased participation. Recruitment was reduced over Christmas and there were recruitment breaks for one week during April 2013 and throughout August 2013. These breaks were due to a lack of available recruiters during holidays and at other times due to the lack of MPhil students being available to take ownership of the study.

Recruitment from a tertiary centre that provides care to the whole population of Liverpool has given the study access to the majority of births in the study area. The use of the LWH also meant routine data was available on all the eligible births in the LWH. This data allowed the comparison between the LBBS cohort and those born in the LWH. It was shown that although there were some discrepancies between the LBBS cohort and LWH births, the cohort was fairly representative of the white population of the infants born in the LWH. The majority of discrepancies could be reduced with weighting during analysis, which makes the results of the LBBS more valid.

18.84% of all the eligible births in the LWH declared their interest in participating in the LBBS. This is low compared to the 69% expression of interest that was produced during the pilot study of the recruitment strategy, however the research team may not have fully anticipated the workload required for both recruitment and administration of the study. This workload increased as the study grew and this may be reflected in these recruitment rates. Other birth cohort study recruitment strategies have led to recruitment of between 50% and 98% of the eligible population (135, 144); regional cohort studies report between 84.3% and 50% recruitment (141, 144). Recruitment rates in the current study are therefore disappointing when compared to other similar studies.

A significant factor in the poor recruitment rate is that the students did not have access to all the eligible births, as they were only present in the LWH four days a week. Infants who had an uneventful delivery and a short postnatal stay are likely to have been missed. Figures of all the eligible births in the LWH hospital, obtained from Professor Ben Shaw, included those infants who died very shortly after birth; the recruitment team are likely to have missed these. As the study progressed, the recruitment team reduced from two persons to one; the workload for one person to run a cohort study and recruit participants simultaneously means that it is likely to have been impossible to recruit all of the eligible births. Previous studies have used midwives and GP's to recruit mothers into the studies and have often involved antenatal or postnatal questionnaires at time of delivery. This means mothers who attend for any form of antenatal care are asked to participate in the study and eligible births are less likely to be missed. Using midwifes to help with recruitment during mothers prenatal care in the current study would have meant eligible participants were less likely to have been missed by the recruitment team. The LBBS did not have a translated version so mothers who did not speak an appropriate level of English could not be consent or reliably complete the questionnaire and would not be included. The use of a translated version of the LRSQ would increase access to the study, and may reduce the under representation of the minor ethnic groups.

Recruitment at birth in the LBBS reduced the long term recall bias of the risk factors during pregnancy; the mothers only had to wait four months to describe these risk factors which is a relatively short period of time when compared to some other birth cohort studies. Recruitment was throughout all seasons of the year removed any seasonal variations in birth and disease patterns. This is a strength of the study that many birth cohorts (which only recruit for one week of the year) do not possess (134-136).

Overall, recruitment in the LBBS was disappointing, however the response rates shown the LBBS can be accounted for and is understandable given the limitation of personnel and the workload required during recruitment for a study of this scale.

Response rates in the LBBS overall are somewhat disappointing. 29.6% of the mothers who declared their interest in the study, equating to 5.57% of all the eligible births, gave consent and completed questionnaire one when their child was four months old (defined as study enrolment). Of those enrolled, 54.0% responded at 10 months, 39.6% responded at 16 months and 37.0% responded at 22 months. Drop out from the study was initially high, but has reduced as the study has progressed.

Drop out was reduced in two ways. Three reminders were sent to participants at weekly intervals if they had not completed a questionnaire and a birthday e-card was sent to the participant on their infant's birthday. The use of reminders and birthday e-cards has been used elsewhere and shown to be beneficial (49) so it is likely to have enhanced the participation in this study. The initial response rate of 29.6% is disappointing in comparison to other birth cohort studies. This may be due to a combination of contributing factors. Along with the low recruitment rate in the LBBS, mothers had to wait four months after declaring their interest in the LBBS before giving consent and completing the first questionnaire. In other study's, the initial assessment of risk factors for disease has been assessed before birth or immediately postnatally by midwives (136, 141, 143). Waiting four months before beginning the study may have contributed to the initial drop out of interested mothers. The Breathmobile study used a \$25 school supply gift voucher as an incentive to participate which successfully increase response rates to their guestionnaire (95). Mothers mentioned this during feasibility testing for the study as a way of increasing response rates. This was unfortunately not feasible in the current study. When previously used the LRSQ produced response rates of 64% and 56% when sent by post (1, 16). Although this is somewhat higher than the response rate in the LBBS the comparison must be must with caution due to the difference in study designs.

Drop out from questionnaire one to questionnaire four is fairly high compared to other similar studies. The ALSPAC study was able to trace 97.4% of its cohort at two years giving a 2.6% drop out over the first two years compared to 63% drop out in the current study. The highest drop out in a similar study was 15.7% in the first year and 26.1% over the first five years. Similar to the results for recruitment, this may be due to limited personnel being involved with the study. Other cohort studies are linked to national databases or involved schools to track changes of address and have personnel dedicated to reducing attrition. The current study relies heavily on email addresses and participants giving correct information. It has not had the resource to fully investigate participant who did not respond to questionnaires to maximise participation. Other cohort studies, particularly national birth

cohorts, have produced low attrition rates, but have had the benefit of tracing the participants using their schools, something that is just not possible in a preschool cohort; these studies showed a much higher dropout rate once participants leave school and become harder to trace (134-136). Likewise, in the other cohort studies there were massive efforts to improve recruitment and reduce attrition rates by contacting and tracking missing participants (142). This has not been possible in this study for a variety of reasons, mostly due to the limited personnel in the study team.

Another contributing factor to the high dropout may be that the study changed to a fully online study; this reduces the access for the whole population of the cohort as not all of the cohort may have computer access. This adds a potential source of bias. On the other hand as shown in the Aberdeen Cohort Study, the use of an online system can be more accurate and prone to fewer mistakes than traditional paper based questionnaires which may have benefited the study overall (138).

In previous studies a good relationship between the study and participants has been built to reduce attrition. This has been done using birthday e-cards, newsletters and a website (49). This is an area where the LBBS may have neglected. The fully online system sent out birthday e-cards and the study had a website and asked if the participants would like to receive a newsletter, the website was very rarely updated and newsletters were not sent regularly to the cohort. This is likely again to have been down the workload for a very small research team. Taking this further the study could have used social media to build a better relationship with the cohort, and better advertise the study. The use of advertisement before the study in the IoM cohort study lead to very high participation rates and low attrition rates (140).

Although unintentional, once participants missed a questionnaire they were not sent any further questionnaires. This will have led to a drop out in participants who had not intended to leave the study, but had not completed a questionnaire for a variety of reasons. Had this not been the case the dropout rates earlier on in the study may have been significantly less. Between questionnaires three and four drop out reduced dramatically. This is due to participants who had not completed the previous questionnaire being invited back into the study; inviting participants back to the study after missing a questionnaire has been done before in the ALSPAC study to reduce attrition (139).

Conclusion

Overall, recruitment and retention in the LBBS has been disappointing. This is most likely to be due to the limitation of personnel involved in the study, and due to the large time waited

from expression of interest shortly after birth to enrolment at four months of age. This large amount of dropout has caused bias within the study that required adjustment during analysis. Results and conclusions taken from the study must be approached with this in mind.

4.2-Profiles of Mothers in the Liverpool Baby Breathing Study

Mothers in the LBBS were older, less deprived, less likely to smoke in pregnancy and more likely to breastfeed than the mothers to eligible births in the LWH and the population of Liverpool. They were more highly educated than the population of Liverpool.

National cohort studies have previously produced cohorts with similar characteristics to the population of England, however smaller regional studies have found it more difficult to be representative of the study population. For example in the ALSPAC study, mothers were more likely to be white, own a home and car, be married and have a higher socioeconomic status and live in an overcrowded home that the eligible population (139, 162). The BiB cohort contained older, more educated mothers (49, 163, 164). The PIAMA study's cohort of mothers was more highly educated and more likely to breastfeed than the population of the Netherlands (144). The differences between the LBBS cohort and the eligible population may be characteristic of small cohort studies. It could be argued that the more highly educated population are less likely to smoke in pregnancy and more likely to breast-feed and this is has produced bias in the LBBS. This is, however, a fairly sweeping statement and would need further investigation before making formal conclusions. Another reason for these discrepancies in smoking and breastfeeding could be due to social desirability bias. It is well documented that socially desirable characteristics are over reported in questionnaires and undesirable characteristics are underreported in questionnaire (165). The over reporting of breastfeeding and underreporting of smoking have both been shown in previous studies (166, 167).

In a study of infants with severe bronchiolitis, successful follow-up was more likely in: those with private health care, those who had a higher income, females and from households with fewer than three children (168). In the LBBS the mothers that responded were less likely to be deprived which mimics the bronchiolitis study. The importance of socioeconomic status in a respiratory study in Liverpool is imperative due the deprived nature of the population of Liverpool (13) and the well documented risk of socioeconomic status on respiratory health (56, 65, 66). The LBBS cohort was less deprived than that of the population of Liverpool and the eligible births in the LWH. This was accounted for during analysis of this study so conclusions were valid and could be applied to the population of Liverpool. This was done by factor weighting during cross-sectional analysis and by using IMD decile as a level during the

longitudinal multi-level mixed effects model analysis. The use of IMD deciles in the LBBS has unfortunately led to relatively few participants in the least deprived deciles. This may have skewed the data during analysis. Analysis of the LBBS may have been better and produced results more concordant with previous findings if quintiles had been used.

It was shown that over the course of the study that less deprived, older, more educated mothers who breastfed and did not smoke in pregnancy were more likely to remain in the study. These differences were reduced to some extent by adjusting the data using factor weighting at each time point during cross-sectional analysis so as to represent the eligible births in the LWH. Using IMD decile and maternal qualification as model levels accounted for this in the longitudinal analysis of the study. Unfortunately, with the exception of socioeconomic status, the inconsistency in drop out was not completely removed from the study. Therefore any conclusions made in the study, especially as the study has grown older, must take this into consideration.

4.3-Profile of Infants in the Liverpool Baby Breathing Study

The LBBS cohort had a similar male to female ratio and a similar proportion of multiple births to the population of Liverpool. The ethnicity of the cohort was predominantly white, with a higher proportion of mixed ethnicity baby's compared to the population Liverpool. This is likely to be due to the questionnaire only being available English. Therefore the conclusions drawn from this study must be used with caution when applying them to infants of the underrepresented, minor ethnicities. Poor representation of the minor ethnicities has been seen in previous cohort studies in England (135, 169).

The LBBS cohort had more preterm babies compared to all the births at the LWH during the recruitment period. This however could be due to preterm babies having increased length of stay in hospital after birth due to complications and monitoring. This would mean they were more likely to be recruited compared to term babies whose stay in hospital after birth is shorter. The LBBS cohort had a similar proportion of low weight births in the cohort compared to all the eligible births in the LWH during the recruitment period. However it had a higher proportion of low weight births compared to the population of England. In 2014, Liverpool as a whole had a similar proportion of low birth weight births compared to all the births in England (170). The higher proportion of low birth weight births in the LBBS may be due to the recruitment centre (the LWH) being a tertiary centre for the whole of Liverpool. Mothers to low birth weight babies may be more likely to attend the LWH than their general district hospital and therefore are more likely to be recruited into the study. Babies born with

a low birth weight are more likely to have an increased duration of postnatal stay in hospital so are more likely to be recruited into the LBBS.

Overall although the profile of the infants enrolled in the LBBS did not represent all the eligible births in the LWH or that of Liverpool. However the LBBS cohort is a good representation of the white population of the Liverpool given the methods of recruitment and workload for the small research team.

4.4- The Clinical Relevance of the LRSQ Score

The LRSQ is a measure of parental reported symptoms and impact of quality of life. It is a measure of what is parentally important and is not a clinical measure of disease severity. Therefore a clinically significant difference in scores between those exposed and not exposed to risk factors is difficult to determine. What difference in questionnaire score represents a parentally important difference is yet to be formally established. Further studies and discussions with mothers to determine a parentally important difference and possibly a clinically significant difference in LRSQ score would be beneficial. It would make future use of the questionnaire more clinically relevant and meaningful.

This being said, the LBBS study has shown which risk factors may increase or decrease the LRSQ score in preschool infants. An increased LRSQ score may mean clinically that the infant may have a larger variety of symptoms, more frequent symptoms or a combination of the two. A larger the difference in LRSQ score between those exposed and unexposed to risk factors is more likely to be a clinically significant as this indicates more frequent or a larger variety of symptoms in an infant. A larger difference in score may be the difference between having symptoms 'every day' rather than 'a few days', and this difference may be in multiple symptoms. A statistically significant result may however, be due to a small increase in LRSQ score. This may not be clinically significant. A small increase in score may be due to having one symptom 'some days' rather than 'a few days'; this difference is clearly open to interpretation and may not be clinically significant when measuring symptoms over the previous 3 months.

Interpretation of the LBBS results must be approached with this in mind. The LBBS has successfully shown which risk factors and exposures may affect the LRSQ score in infants. This statistically significant change in score may not be clinically significant or parentally important. However, an accumulation of many risk factors which have an effect on LRSQ score is more likely to cause a parentally important and clinically significant change in symptoms.

4.5-Cross-sectional Analysis

The cross-sectional analysis of the data in the LBBS has allowed the description of respiratory symptoms in preschool children from birth until 22 months of age in Liverpool. Analysis compared LRSQ scores, symptom prevalence and health care service use between those exposed and those not exposed to risk factors for respiratory disease at each of the four time points studied.

4.5.1-Sex

Male infants consistently had more respiratory symptoms than females from birth to 16 months of age. Females however, had more respiratory symptoms than males at 22 months of age. Males had higher LRSQ scores at 4, 10 and 16 months. At four months, males had a higher prevalence of SoB and noisy breathing and attended their GP more frequently with respiratory symptoms than females. At 10 months, males had a higher prevalence of cough and SoB and were more likely to attend hospital with respiratory symptoms than female infants. At 16 months, males were still more likely to attend the hospital with respiratory symptoms than female infants. By 22 months however, females scored higher for 'symptoms with colds' showing a swing in the prevalence of respiratory symptoms from males to females. The increased respiratory symptoms in males is had a detrimental effect on their QoL at 10 and 16 months and their families QoL throughout the first three questionnaires.

In previous work it has been shown that males wheeze more frequently and are more likely to have a physician diagnosis of asthma and be hospitalised due to their asthma than females throughout the first five years of life (25, 46). It has been shown previously that males from Liverpool are at an increased risk of asthma (151). Although most of the previous findings concentrate solely on wheeze and asthma diagnosis, the LBBS has given similar results in showing that males have more respiratory symptoms earlier on in life than females. The prevalence of noisy breathing was higher in males at four months than females. This may well have been wheeze that was wrongly identified as 'noisy breathing from the throat' or "noisy breathing not from the chest' due to lack of parental expertise in identifying wheeze.

4.5.2-Multiple Birth

There is very little published work on the effect of multiple births on respiratory symptoms. No differences were observed in the LBBS between infants born as part of a multiple birth and those born as single births. Due to the very small population of infants from multiple births in this cohort, this study would not be suitable from which to draw conclusions. A further study with a larger population of multiple births would be more appropriate.

4.5.3-Ethnicity

Due to the underrepresentation of minor ethnic groups, the over representation of white and mixed race infants in the cohort and the unequal drop out between different ethnic groups, the cohort was split into 'white' and 'other ethnic groups'. Therefore conclusions relating to the ethnicity must be treated with caution. This uneven drop out, very extreme grouping, along with the small sample size of 'other ethnic groups' may have caused the results to be unreliable and inconsistent with previous findings.

Although there was no difference in total LRSQ score or domain scores between white and other ethnicities at four months, white infants had a higher prevalence of rattily chest, snoring, and rapid breathing compared to other ethnicities. At 10 months, white babies had a higher LRSQ scores compared to other ethnicities. The prevalence of wheeze and rattily chest was higher in white babes than other ethnicities. These results were mimicked in the QoL scores for white infants and their families. At 22 months, other ethnicities had higher scores for 'symptoms without colds' and 'symptoms on increased activity'.

The millennium cohort study found that in the first three years of life Black Caribbean's reported higher rates of 'ever asthma' and wheeze compared to the white population of Britain, whereas Bangladeshi population reported a lower prevalence of 'ever asthma' (171). In the analysis of the LBBS cohort we combined these two populations so any discrepancies in respiratory symptoms between these ethnicities may have been missed. The Leicester respiratory cohort observed that South Asian infants had a lower prevalence of wheeze than the white infant population of England but they attended hospital more frequently (172). Again in the current study if this relationship was present it may well have also been missed due to the grouping of the populations.

4.5.4-Maternal Age

At four months, infants born to younger mothers had higher LRSQ scores than children born to older mothers. The prevalence of wheeze, rattily chest, SoB and rapid breathing was higher in infants born to younger mothers than those born to older mothers. The percentage of infants attending GP and hospital decreased as maternal age increased, possibly reflecting confidence of older mothers to manage minor illness in the youngest of infants. Similarly at 10 months, infants born to younger mothers had higher LRSQ scores than those born to older mothers. There were a few exceptions to this rule. Infants born to mothers aged over 45 years had higher scores 'with colds' than infants born to mothers aged 35-44 and higher scores 'on activity' than those born to mothers under 20 years. At 16 months the trends between younger mothers and infant respiratory disease became less consistent and convincing; there was no clear trend in the results when comparing age groups. At 22 months drop out caused the loss of the categories of mothers under 20 and over 45 years. At 22 months there was no obvious trend between respiratory symptoms and maternal age.

The general trend throughout the study has been that infants to younger mothers had more respiratory symptoms than infants born to babies of older mothers, particularly in the first 10 months of life. In previous work there is mixed consensus of the effects of maternal age on respiratory health in infancy. It has been shown that infants to younger mothers are more at risk of LRTI in the first four months of life (173), and that infants born to mothers at the extremes of reproductive age were associated with early remitting wheeze. This being said, the same study showed low maternal age was protective of persistent and relapsing wheeze (77). Our study agrees with these findings somewhat. At early survey points in the LBBS (4m & 10m) the relationship between maternal age and respiratory disease is more convincing than in the second half of the study (16m & 22m). The inconsistent relationship between maternal age and respiratory symptoms in the second half LBBS may be due to the small sample size of the age groups at the extremes of maternal age. Drop out caused the loss of infants born to mothers aged less than 20 year and over 45 years. Because of this, findings related to the extremes of maternal age must be treated with caution. This relationship shown between maternal age and infant respiratory disease in the first 22 months of life may be due to confounding factors which are more apparent in younger mothers than in older mothers. This however, requires further investigation. In the early stages of the study younger mothers attended GPs and hospitals with their infants due to respiratory symptoms more frequently than older mothers. Although this matches the trend of respiratory symptoms in the LBBS, it must be questioned if this trend is due to parenting experience or if there is a physiological explanation for this.

4.5.5-Maternal Education

At four months the prevalence of rattily chest and SoB was lower in those infants born to mothers with a higher level of education compared to those born to mothers with a lower level of education. Likewise health care attendance was higher in those infants whose mothers had less education. The only exception to this was that no infants born to mothers with no qualifications attended hospital in the first four months of life. This may be due to lack of access to healthcare, the small sample size of infants born to mothers with no qualifications or the due to poor symptom recognition by the least educated mothers. At 10 months the general trend in the results was that those born to mothers with a higher education had lower scores for 'daytime symptoms', 'symptoms without colds' and 'symptoms on increased activity'. The only exception to this trend was that infants born to mothers with A-levels had higher scores than those who had no qualifications and other qualifications for 'symptoms with colds' and 'symptoms on increased activity'. These trends were mimicked in GP and hospital attendance and QoL domains for both the infant and their family. However those who had no qualifications had no attendance to health care; this could be due to similar reasons discussed at four months. At 16 months infants born to mothers who had a higher level of education had a lower prevalence of 'noisy breathing not from the chest' and rapid breathing than those born to mothers who were less well educated.

In general total LRSQ score decreased with increasing education at 22 months. Contrastingly domain analysis shows increased scoring in both the most educated and the least educated. This trend was mimicked in symptom prevalence and QoL scoring. For example infants born to mothers with a post graduate degree had a higher prevalence of wheeze, cough, rattily chest and snoring than those who attended higher education and those with GCSE's where as those with no qualifications had a higher prevalence of wheeze and rattily chest and snoring than those with GCSE's. Likewise infants born to mothers with a degree or postgraduate degree and no qualifications had a worse QoL due to respiratory symptoms than those born to mothers who attended higher education, had GCSE's or other qualifications.

These findings at 22 months are interesting. It could be explained by the increased drop out in less educated mothers through the study, however this would not fully account for the increased scoring at 22 months in the more educated families. Maternal education will have a significant impact on many risk factors what have been previously described in the literature and a combination of these factors could contribute to the respiratory health of the cohort. Although there has been very little published regarding the influence of maternal education on their infant's respiratory health, other factors, which could be linked to maternal education, have been investigated. It has been shown that poor housing increases respiratory disease in the first two years of life (57) and children from poorer families had an increased risk of LRTI. Likewise the prevalence of childhood asthma was increased in children from families with low socioeconomic status (56, 65, 66). Poverty in early childhood was associated with increased risk of asthma attacks by four years (174). A western diet with high levels of trans fatty acids may increase the risk of asthma, whereas a diet high in fish, fruit and vegetables may be protective (75, 76). The current study may agree with these previous findings if maternal education was to be used as a marker of the above risk factors for respiratory disease. The increase in respiratory symptoms in the more educated at 22 months

may be due to other factors described in the literature for which maternal education could be a marker for. Perhaps this may be a marker for exposure to excessively clean environments during pregnancy and excessive household cleaning chemicals during infancy. These exposures are associated with increased wheezing, persistent wheezing and lung function abnormalities (175) in line with the hygiene hypothesis of asthma (67). Although the above may provide some explanation for the influence of maternal education on respiratory disease in the LBBS, these are very sweeping statements and these concept needs exploring further before any formal conclusions can be made.

4.5.6-Gestation

Preterm infants were shown to have more respiratory disease from birth to 16 months of age than term infants. Preterm infants had a higher total LRSQ score and a higher prevalence of rapid breathing at four and 10 months of age. Preterm infants scored higher for a variety of domains, and had a higher prevalence of snoring and noisy breathing at 16 months. These results were mimicked in the QoL of both the infant and their family at 10 months of age. Interestingly and contrary to what might be expected, preterm infants had a lower prevalence of colds at four months; this result is difficult to explain and may need further investigation. It may result from reduced community exposure due to protective parenting. Disregarding the decreased prevalence of colds at four months, similar results have been replicated in previous studies. It has been shown that preterm birth is associated with wheezing, the use of inhalation therapy and hospitalisation due to respiratory disease in the first year of life (50) and this increase in wheeze had been shown to be persist for the first seven years of life(49). Although we were not able to show that that wheeze specifically was increased in the first year of life we were able to show that noisy breathing was increased in the first year of life. Contrary to this, there has been some mixed evidence on the frequency of respiratory symptoms in premature babies (51) and this discrepancy may account for the lower prevalence of colds at four months in preterm babies in this study. This finding could also be due to the fairly small sample size in the preterm birth category.

4.5.7-Birth weight

Infants born with a low birth weight had a higher incidence of colds and were more likely to use health care services due to respiratory symptoms than those who were born with a normal birth weight at four months and at 10 months. This is somewhat replicated in previous findings. It has been shown that babies with smaller birth weight have reduced lung function at six weeks of age (176) and are at an increased risk of wheezing disorders in the first five years of life (46, 55). Although the current study did not manage to replicate the exact symptoms shown in previous work, it did show that infants born with low birth weight had more respiratory symptoms than those born at normal birth weight. In the LBBS the sample size of low birth weight infants was still fairly small and drop out caused a further reduction in this sample as the study progressed. This could account for the findings in the current study not being as prevalent and convincing as in previous work.

4.5.8-Breastfeeding

At four months children who were breastfed scored lower for 'symptoms on increased activity' than children who were not breastfed and those who were still being breastfed at four months had lower night-time symptom scores than those who were either not breastfed or breastfed for fewer than four months. The prevalence of SoB and healthcare attendance was lower in those that were breastfed than those who were not breastfed. At 10 months those who were breastfed had lower scores for symptoms without colds than those who were not breastfed. Those who were breastfed for over four months had the lowest scores, indicting a beneficial dose effect for duration of breastfeeding. The prevalence of SoB, noisy breathing from the throat and health service attendance was lower in babies that were breastfed. At 16 months those who were not breastfed. Infants who were breastfed had a lower prevalence of cough than infants who were not breastfed. However at 22 month those who were breastfed had higher scores at night-time than those who were not breastfed. Overall, those who were breastfed had lower respiratory symptoms than those who were not breastfed and increasing breastfed had lower respiratory symptoms than those who were not breastfed.

Similar results were shown in the Southampton's women's cohort were breastfeeding was shown to be protective of adverse respiratory symptoms; this was proportional to length of time of breastfeeding (7). Likewise breastfeeding, particularly exclusive and prolonged breastfeeding, was shown to be protective against hospitalization for LRTI in the first three years of life. It was estimated that 25% of LRTI requiring hospitalization could have been avoided by breastfeeding (173, 177). Although we cannot show the cause of hospital admissions with current data, the study has been able to show that breastfeeding has reduced hospital attendance due to respiratory symptoms by 11.1% at four months and 9.1% at 10 months. In previous work it had been shown that infants who were breastfeed were more likely to have asthma (72) which could not be shown in the LBBS as there was no structured assessment of the participants by health care professionals as part of the study protocol. It is reported that breastfeeding has no long-term effect on adult lung function (178); in the LBBS cohort the beneficial effect of breastfeeding is reduced by 22 months. This

could be due to the reduction in beneficial effect of breastfeeding or alternatively could be due to the uneven drop out between the breastfeeding and non-breastfeeding groups. Although weighting of the data reduced this uneven drop out somewhat, it didn't completely eliminate this. The implications of these findings on clinical practice are that breastfeeding where possible should be encouraged due to the beneficial effect on respiratory health in infants in the first 16 months of life.

4.5.9-Smoking Exposure

Smoking exposure in Pregnancy

In the first four months of infancy, GP attendance was more likely in those infants who were exposed to any source of tobacco smoke in pregnancy. Paradoxically at 10 months those infants who were exposed to any tobacco smoke in pregnancy had lower 'scores without colds' than those who were not exposed to tobacco smoke in pregnancy. At 16 months those who were exposed to smoking in pregnancy from other household smokers had a lower LRSQ score than those who were not exposed. Those infants who were exposed to smoking in pregnancy but a higher prevalence of snoring. At 22 months those who were exposed to smoking in pregnancy but whose mother who didn't smoke had a higher LRSQ score than those who were not.

Although some results in the current study agree with the well documented adverse relationships between smoke exposure in pregnancy and infant respiratory health, the results from the current study are not convincing. The results produced between 10 months and 22 months are contradicting what has been previously published. Maternal smoking in pregnancy has been repeatedly shown to increase the incidence of wheeze and asthma diagnosis up to six years of life (5, 42, 43, 179). The lack of concordance in the current study are likely to be due to the underrepresentation of the population who were exposed to smoke in pregnancy in the study cohort when compared to the population of Liverpool. Although weighting of the data reduced this under representation somewhat, the population exposed to smoking in pregnancy was still smaller than in the population of Liverpool. Drop out was higher in the exposed group than in the non-exposed group; again weighting reduced this unequal drop out but it could not completely eliminate the differences. The exposed group was also fairly small in comparison to the non-exposed group and by 22 months the exposed group sample size was incredibly small which also could account for the unexpected results. All this said, the results at 22 months were as expected, with the exposed group showing higher LRSQ scores than the non-exposed group and it will be intriguing to see if this trend continues in the future questionnaires in the study.

Environmental Tobacco Smoke Exposure after Birth

The proportion of the cohort that was exposed to household tobacco smoke after birth decreased through the study. This could have been due to the reducing prevalence of smoking in Liverpool over the study period (180) or due to the increased drop out in smokers as the study aged. It is likely to be a combination of the two. There was no difference in respiratory symptoms between those who were exposed to ETS and those who were not in the first 10 months of life. Paradoxically, at 16 months however those who were exposed to ETS exposure in the last six months showed lower domain scores for 'daytime symptoms', 'symptoms with colds' and 'symptoms on increased activity' than those who were not exposed. Those exposed had a higher prevalence of snoring than those who were not exposed to tobacco smoke. At 22 months those infants exposed to tobacco smoke had a lower prevalence of SoB than those who were not exposed.

Like smoke exposure during pregnancy, these results were not as expected considering how well documented the negative impact of ETS exposure is on infant respiratory health. Infants exposed to ETS have been shown to be more likely to have a LRTI in the first three years of life, younger at the time of their first infection and require more antibiotic treatment and admissions to hospital for LRTI than those who are not exposed to ETS (6, 45, 46). Household smoking is associated with an increase incidence of wheeze in the first (47) and second year of life (5) and an increase in asthma and allergic symptoms in the first two years of life and a decrease in remission from these symptoms (48). Like smoke exposure during pregnancy the smoking population in the LBBS cohort was largely under represented, and although weighting of the data reduced this to some extent there was still a difference between the population of the cohort exposed to ETS and the rates of smoking in Liverpool. Drop out in the study was also higher in the population exposed to ETS than in the population that was not exposed to ETS. A study in Canada showed that smoking mothers under reported their child's cough (181). Liverpool mothers who smoked may have systematically underreported symptoms in the LBBS. All these factors combined could be responsible for the unexpected results found in the current study, however it is hard to justify this. The results shown in the current study with regards tobacco smoke exposure were not expected.

4.5.10-Family History of Atopy

Over the first 16 months, those with a FH of atopy scored higher in a variety of domains than those without a family history of atopy. Infants with a family history of atopy had significantly higher total LRSQ scores at four months. They had a higher prevalence of noisy breathing from the throat and noisy breathing not from the chest than those with no family history of atopy. Their mothers reported a worse QoL. At 10 months these infants had a higher prevalence of colds. GP and hospital attendance was more likely in those with a family history of atopy. At 16 months those with a family history of atopy were more likely to attend the GP with respiratory symptoms.

In previous studies, atopy had been linked with an increase in wheeze and asthma particularly around the age of two years(70). The current study population is not yet two years of age so it will be interesting to see how the prevalence of wheeze is effected in future sweeps of the study. In the first four months of life there was an increase in noisy breathing in infants with a family history of atopy. This could have been wheeze misinterpreted by the mothers however this cannot be used to make formal conclusions.

Further analysis should be undertaken to look at differences in which family member had atopy to see differences in respiratory symptom patterns. Others' work has shown that maternal asthma is a risk factor for wheeze in males at two years and above and females in five years and above whereas paternal asthma was risk factors for males only between two and five years of age (71).

4.5.11-Nursery Attendance

Nursery attendance increased over time as recorded at each of the four questionnaires. The biggest increase in attendance was between 10 and 16 months and is likely to be due to mothers retuning to work after maternity leave. In the first four months of life very few infants attended nursery. Those who attended nursery in the first four months reported a higher prevalence of snoring, and were more likely to attend the GP with respiratory symptoms than those who did not. From 10 months onwards those who attended nursery had higher total LRSQ scores and scored higher in a variety of domains than those who did not attend nursery. They also had a higher prevalence of a variety of symptoms from 10 months onwards and were more likely to attend the GP at 10 and 16 months. At 10 and 16 months both the infant and their families had a worse quality of life due to respiratory symptoms than those who did not attend nursery.

The association between nursery attendance and respiratory symptoms in preschool infants was the most consistent finding in the LBBS; similar domain scores and the prevalence of similar symptoms were consistently increased throughout the study. The smaller impact of nursery on respiratory symptoms in the first four months of life is likely to be due to the low numbers of attendees at this time.

Similar findings have been published in previous work; infants who attended nursery were more likely to have LRTI in in the first year of life (45) and throughout infancy (173) than those who did not attend. Early nursery attendance was associated with increased airway symptoms in the first four years of life (74). Others have reported that nursery had no effect on respiratory symptoms after four years and have described reduced prevalence of wheeze at five years (182) but it remains to be seen if this will be replicated in the current study. Our study finds that nursery attendance is associated with an increase in respiratory symptoms over the first 22 months of life, however its remains to be seen how nursery attenders will be affected in the LBBS after 22 months of life. Previous studies have shown that there is no long-term detrimental effect of nursery attendance on respiratory health, and that it may be protective after four years of age (74). Our findings should be approached with caution when using them to decide whether an infant should attend nursery, and must be balanced with potential benefits for child development and allowing opportunities for mothers to return to employment.

4.5.12-Sharing a Bedroom

Over time the proportion of the cohort sharing a bedroom decreased, as did the proportion of the infants who shared a bedroom who shared with their parents.

At four months sharing a bedroom had a protective effect on infants' respiratory symptoms. Those who shared a bedroom had a lower total LRSQ score and lower scores with colds than those children who slept alone. Infants who shared a bedroom had a lower prevalence of cough, SoB and colds than children who slept alone. This was mimicked in the QoL scores. In contrast, GP attendance was more likely in infants who shared a bedroom. By 10 months this protective effect had reduced, by which time infants who shared a bedroom only had a lower prevalence of snoring. By 16 months there was no difference between those who shared a bedroom and those who did not.

In previous literature there is very little evidence on the effect of bedroom sharing on respiratory health in infancy. From our study, there may be an association between sharing a bedroom with parents and respiratory symptoms. The decrease in respiratory symptoms in the bed sharing population throughout the study and the reduction in the proportion of the cohort sharing a bedroom with their parents may show this. A formal relationship cannot be concluded from this study however and further investigation of this trend is required.

In some cases sharing a bedroom may be associated with household overcrowding which is associated with increased respiratory disease (58). Household overcrowding can be due to low socioeconomic status (183). In this study the opposite association is seen between respiratory disease and bedroom sharing which could be due to the cohort sharing a bedroom by choice of the parents rather than due to household overcrowding or low socioeconomic status. The reduction of infants sharing a bedroom throughout the study, particularly in those who share a bedroom with their parents could be proof of this. The lack of information on the household overcrowding in the cohort makes it impossible to make formal conclusions regarding this although may highlight an area for further research.

4.5.13-Household Children

At four months infants who lived in households with other children had higher scores with colds and had a higher prevalence of colds than children who lived with no other children. GP and hospital attendance due to respiratory symptoms was higher in those who lived with other children. Both the infants who lived with other children and their mothers reported a worse QoL. In contrast, children who lived with other children reported a lower prevalence of snoring than those who did not live with other children. By 10 months the association between living with other household children and increasing respiratory symptoms was no longer present.

Although there was no difference in respiratory symptoms, infants who lived with other household children were still deemed to have a worse QoL at 10 months and their mothers reported a worse QoL at 16 months. At 22 months infants that lived with other household children reported lower scores on increased activity than infants who did not live with other children.

Previous work has shown that having other siblings at home was associated with decreased risk of asthma diagnosis between 4.5 and 14 years of life years of life (59-61). Although this is outside the age range of the current study, the LBBS has shown that there is no increased risk of wheezing or noisy breathing in households with other household children. In contrast household overcrowding has been shown to increase the risk of lower respiratory tract infection in children (58). In the current study, in the first four months of life in infants who shared a household with other infants were at an increased risk of colds. Unfortunately in the current study we do not have information household overcrowding. However, households with other infants are more likely to be over crowed than households with only one infant. This means the findings in the LBBS could be in line with previous research. Further research would be required before formal conclusions could be made. The decrease in scores on increased activity at 22 months may be in line with the hygiene hypothesis of asthma (68, 69). It will be intriguing to see if this relationship develops further as the study continues.

4.6-Longitudinal Analysis

The longitudinal analysis of the LBBS data has allowed the study team to determine the change in the LRSQ scores over time and the differences of this change in those exposed to risk factors by creating multi-level mixed effect models.

The longitudinal analysis of questionnaire scores using multilevel mixed effects models has been scarce in previous study reports. The majority of cohort studies have analysed the prevalence of specific symptoms in exposed and non-exposed groups over time by univariate and multivariate regression analysis. The use of multilevel mixed effects models in the preschool age group seems to be quite novel.

We have shown that over time nursery attendance, being a male and the presence of other household children increases total LRSQ scores over time most dramatically. The biggest protective factors over time are sharing a bedroom and breastfeeding. Although analysis in this way is scarce in the preschool age group, these findings would agree in general with previous research as discussed earlier.

The LBBS agrees with studies by others that infants who attended nursery had an increased prevalence of an array of respiratory symptoms at various age groups up to four years of life (74, 173, 184). The LBBS is novel in showing that these symptoms increase over time. The change in scores over time may however be due to the low numbers of infants attending nursery at four months. The effect of nursery attendance on respiratory symptoms over time may therefore not be as large as this study portrays.

Likewise males have been shown to have an increase in various respiratory symptoms at various time points through the first five years of life when compared to females in previous work. The Tucson study found that the peak of wheeze in males was between six to eight years of age (45) which unfortunately is outside the scope of this study. It will be interesting to see of the effect of being male on respiratory symptoms over time is as large as the study progresses further towards its end at five years of age.

The presence of other household children has been shown to increase respiratory symptoms over 22 months. As before, there has been little work on the presence of other household children and its effect on respiratory disease in preschool children. What research that has been done is conflicting: having other siblings at home was associated with decreased risk of asthma diagnosis but household overcrowding has been shown to increase LRTI in pre-school children (59-61). Taking all this into account it may be appropriate to conclude that ours is a novel finding and may require further investigation in the future to allow formal conclusions to be made.

There is very little prior evidence on sharing a bedroom and its impact on preschool children's respiratory health over time. This study has shown a reduction of LRSQ score over time in those that shared a bedroom compared to those who did not. This phenomenon is difficult to explain. One explanation for this may be the significant change in the sample sharing a bedroom over time in the cohort; there is a large reduction in those who share a bedroom, particularly in the cohort sharing with their parents. It may be this changing sample, and infants moving between exposed and unexposed groups throughout the study that has caused this finding. A further study to look into this relationship in the future would be helpful in explaining this relationship.

Breastfeeding has been shown to reduce LRSQ score over time. This is as expected as there is a large amount of literature showing the benefits effect of breastfeeding on infant's respiratory health (40, 41, 185). However other studies found this benefit to be fairly short lived and has no long-term effects on adult lung function (39, 178). It will be interesting to see if the protective effects on breastfeeding over time persist beyond 22 months of age or are reduced as previous literature has indicated.

4.7-Questionnaire Validity

The internal validity of the LRSQ was deemed as acceptable to good in this study. This is similar to that seen in previous studies that have used the LRSQ. The comparison to previous studies is shown in table 31.

Domain	LBBS	Powell et al. (1)	Trinick et al. (16)
1- Day Time Symptoms	0.81 (Good)	0.95 (excellent)	0.76 (Acceptable)
2- Night-time Symptoms	0.77 (Acceptable)	0.80(Good)	0.64 (Poor)
3- Symptoms with Colds	0.86 (Good)	0.95(excellent)	0.66 (Poor)
4- Symptoms without Colds	0.81 (Good)	0.95(excellent)	0.81 (Good)
5- Symptoms on Activity	0.85 (Good)	0.95(excellent)	0.79 (Acceptable)
6- Other Respiratory Symptoms	0.78 (Acceptable)	0.95(excellent)	0.80 (Good)
7- Childs QoL	0.85 (Good)	0.91(excellent)	0.89 (Good)
8- Family's QoL	0.86 (Good)	0.87 (Good)	0.79 (Acceptable)

Table 31: Comparison of the internal validity of the LRSQ in the LBBS compared to previous studies
Comparison to previous studies shows that although the internal constancy is not as good as described by Powel et al. in some domains, it is better in most parts than Trinick et al. Trinick et al. used the LRSQ on older children with Cystic Fibrosis (CF). The fact the current study has shown better internal consistency than Trinick et al. shows the LRSQ may be more consistent in assessing individuals of a whole preschool population rather than specifically CF patients, or is more consistent when used in longitudinal studies rather than in a one-off assessment. It may appear that Powell et al. shows the LRSQ to be more consistent in their study than the current study however Powell et al. combined domains one, three, five and sixe to assess day time symptoms which may have improved the internal consistency of these domains. LRSQ has shown acceptable to good internal consistency in the LBBS. This adds strength to the study and proves that the LRSQ is an acceptable measure of respiratory symptoms in longitudinal analysis of a predominantly white, preschool population in Liverpool.

4.8 The Use of Contemporary Technology in the LBBS

In June 2015, AFC[®] was suspended and the study changed to using JotForm[®] for questionnaire completion and response storage. At this point, the LBBS became an online study that was fully automated. MailChimp[®] was used for the distribution of emails containing the questionnaire link to the cohort.

The change to the fully automated online system had a number of advantages, although it may have reduced the access to the study for those without email access. The use of a fully automated online system is a novel concept in a study of this kind. It has reduced drop out due to migration and moving house as participants can be tracked by email. However this does not account for people who change email address. Another advantage for the online system is that the entire questionnaire must be fully completed for the questionnaire to be returned to the research team. This means all data is fully completed and is of good quality. There is only a small amount of missing data in the study. A small proportion of maternal ages are missing, as this question was added to the study at a later date. Where this was missing efforts were made by the research team to input this data at a later date. Some data on the number of household children from questionnaire three and four is missing. This was due to a problem with the questionnaire formatting and has now been resolved. The use of a fully automated online system also reduced the heavy workload for the small research team, which has allowed their efforts to be focused elsewhere. This has improved the quality of other aspects of the study including analysis.

The change to a fully online automated system, although it added benefit to the study by reducing the workload of the research team and improving accuracy of questionnaire

completion, reduced the access to the study for people without Internet access. Given widespread ownership of smart phones, this is likely to be a very low number of participants in the study, however it is still a source of bias within the study that must be taken into consideration.

4.9-Strengths of the LBBS

There are many strengths that the LBBS possesses that have meant that overall the study has successfully investigated the natural history of maternal reported respiratory disease in preschool children from birth to 22 months in Liverpool. The use of questionnaires has allowed the assessment of many risk factors in the study at one time and the majority (although not all) of the results in the study have been shown to be replicated to some extent in previous work. The consistency with previous findings adds strength and validity to the findings of this study. The use of the LRSQ is a strength of the study as it has been previously validated in this age group and in the Liverpool population. It was shown to have acceptable to good validity in the current study.

Although recruitment rates were fairly low compared to other cohort studies, there is still a respectable sample size in this study with an adequate number of infants being exposed and not exposed to the majority of risk factors and the cohort was fairly representative of the white population of the infants born in the LWH. The use of a pilot study to optimise recruitment, questionnaire design and overall running of the study has allowed the study to run at an optimal level given the lack of personnel and funding for the study. The study was accessible and acceptable to the participants (2, 3). The use of both emails and postal questionnaires in the early stages of the study has increased the accesses to the study and allowed people without accesses to email to participate. This has reduced sampling bias in the study.

The use of a tertiary centre for recruitment gave the study access to the majority of births in Liverpool. All year round recruitment reduced seasonal variations of birth. The use of routine birth data allowed the comparison between the LBBS cohort and those born in the LWH.

Recruitment at birth reduced recall bias of the risk factors during pregnancy that were studied in the LBBS. Recall bias was reduced throughout the LBBS by using a short time period between questionnaires and by asking mothers to only recall from the previous three months.

4.10-Weaknesses of the LBBS

The LBBS has similar weaknesses that are shown by the majority of birth cohort studies. The use of questionnaires introduces recall bias into the study although this was minimised by asking mothers only to recall from the previous three months and having regular questionnaires. Longitudinal studies are also prone to attrition; this was very prominent in the LBBS. Although there were efforts made to reduce attrition, the lack of and changing personnel in the LBBS research team is likely to have contributed to the large attrition rates compared to similar studies. The lack of ownership of the study was due to university holidays and the changeover of the study from student to student, year on year. Drop out in the LBBS was unequal in exposed and unexposed groups, particularly in the more deprived, those who were exposed to smoking and those who were not breastfed. The large and uneven attrition in the LBBS has added confounding factors to the study.

In the LBBS there is no formal assessments of the infants by health care professionals, neither is there serological confirmation of RTIs. The study relies fully on the infants' parental assessment of respiratory symptoms that may not reflect a medical assessment; however this study deliberately focuses on the parental experience.

Participation rates in the study were low and there are many sources of sampling bias in the study. The infants in the study are more likely to have been hospital for longer periods which means the proportion of 'non-healthy' babies are likely to be higher than the population of Liverpool. Similarly, there was no translation of the questionnaire, which meant that participants needed to speak sufficient English to participate. This led to an under representation of the minor ethnic groups in the LBBS. Babies with social issues or involved with the authorities were also not recruited into the study. Although the use of the LWH as a base for recruitment gave access to the majority of births in Liverpool this did not allow for the recruitment of any infants born at home or those born in the other general district hospitals in the area.

Like seen in pervious birth cohort studies the profile of the mothers and infants in the LBBS was not representative of the population of the eligible births in the LWH and that of Liverpool. Mothers were older and lived in postcode with a more affluent IMD decile than the general population of Liverpool. Infants were more likely to be of white ethnicity, be preterm, be breastfed and less likely to have been exposed to smoke during pregnancy.

Drop out in the LBBS was not uniform; younger, more deprived mothers who did not breastfeed and who smoked in pregnancy were more likely to drop out of the LBBS. This was

accounted for somewhat by weighting of the study during analysis and the use of multi-level models to analyse the study, however these differences were not completely removed from the study. This weakness of the study reduces the validity of the results when applying them to the population of Liverpool.

The cohort has very strict inclusion criteria, including only a very small well defined population specific to that of Liverpool; this means the findings from the LBBS study must be used very cautiously when applying them to the whole population of England.

Although the majority of findings in this study agree with previous literature there are some discrepancies in the findings especially with regarding smoke exposure. These discrepancies are a weakness of the study given how well documented the effect of smoke on infants respiratory disease is. The lack of findings concordant with previous work regarding smoking is a concerning and may mean the findings of the LBBS unfortunately must be approached with caution.

4.11-Future Recommendations

Future recommendations for the remainder of the study are to continue the analysis of the study in a similar way as in this thesis. This may however, become more difficult as drop out increases and exposed and non-exposed groups become smaller and fewer student are willing to undertake intercalated MPhil research degrees. It will be interesting to see if the protective effect of breastfeeding disappears completely in the future of the study and if smoking exposure becomes a significant negative risk factor for respiratory health preschool infants. It will be interesting to see if evidence supporting the hygiene hypothesis of allergy and asthma becomes more apparent as the study continues.

In the future another study of this kind would be recommended with an improved study design and recruitment strategies. The use of the LRSQ in a longitudinal study using an online automated system was a novel, exciting and successful concept. It is disappointing the study could not produce higher initial recruitment rates and lower attrition rates, otherwise this would have created a larger and more representative cohort study whose results would have been more likely to be concordant with previous findings. To improve the study design a bigger research team would be required with less changes of personal and changes in ownership of the study. Members should be dedicated to different rolls within the study whether that be recruitment, administration or analysis. The use of the fully automated online system from the beginning for the study, with no postal questionnaires may reduce

the workload of the research team and therefore make the running of a study of a future study of this kind more efficient.

In the future a translated version of the LRSQ should be produced and validated; this would allow studies like the LBBS to recruit from minor ethnicities and remove language barriers in studies using the LRSQ This would reduce the under representation of minor ethnicities in future studies.

Further studies investigating the roll of household overcrowding, multiple births, maternal education, maternal age and bedroom sharing on the respiratory health of preschool infants are recommended to further investigate the findings of the current study. There current study was not sufficient to assess these variables with confidence.

The LRSQ should be further analysed to identify questions that have reduced the internal validity of the questionnaire. The improved questionnaire may be more appropriate for use in longitudinal studies of this kind. It would also be beneficial to work with parents to develop what a clinically significant difference or a parentally meaningful difference in symptom questionnaire scores would be. This would make the questionnaire and findings of this study more clinically relevant.

Chapter 5-Conclusions from the LBBS

A profile of the mothers, families and infants in the LBBS was created and compared to the population of Liverpool and eligible births in the LWH. Mothers in the LBBS were older, less deprived, higher educated, less likely to smoke and more likely to breastfeed than the population of Liverpool. Throughout the study, younger, more deprived, less educated mothers who smoked and did not breastfeed were more likely to drop out of the study. This profile of mothers, who were more likely to participate and be retained in a birth cohort study, has been seen in previous work and was expected. To adjust for this effect, the study analysis was weighted so the deprivation in the study was identical to that of all the eligible births in the LWH. Infants in the LBBS were more likely to be of white or of mixed ethnicity and be preterm than the population of Liverpool and the births in the LWH.

The LBBS has described risk factors and protective factors for respiratory disease in the preschool population of Liverpool in the first 22 months of life. Risk factors (shown in either cross-sectional or longitudinal analysis) for respiratory disease are: being male, being born to a younger mother, having a family history of atopy, nursery attendance, preterm birth, low birth weight and living with other household children. Protective factors for respiratory disease were sharing a bedroom and breastfeeding. There was a beneficial positive effect of prolonged breast-feeding. There appears to be a complex relationship between maternal education and respiratory disease in preschool infants in Liverpool that warrants further investigation. The findings in the LBBS for the relationship between tobacco smoke exposure and infant respiratory disease conflict in part with previous findings.

Respiratory symptoms had an impact on the QoL of both the infants and their mothers in the LBBS. Respiratory symptom scores were consistently correlated with QoL scores throughout the study. Respiratory symptoms most commonly affected the infant's sleep through the first 22 months of life. In the first four months of life the biggest effect on the infant's mother was causing worry. There after the biggest effect on the infant's mothers QoL was disturbing their sleep.

Being male, living with other household children, being preterm and attending nursery all reduced QoL in infants due to their respiratory symptoms at various time points through the study. Maternal education had an impact on the infants QoL due to their respiratory symptoms but there was no clear trend. Sharing a bedroom increase the infants QoL at four months as did smoking exposure in pregnant at 16 months; this finding is again disappointing

due to the well documented negative impact of smoking in pregnancy on infants respiratory health.

Mothers of males, of infants with a family history of atopy, who had more than one child, of infants who were born preterm, and of infants who attended nursery had a worse QoL due to their infant's respiratory symptoms. Like infants QoL, maternal education had an impact on mothers QoL but the trend of relationship was not clear and needs further investigation.

The LRSQ was shown to be a valid questionnaire to assess maternal reported respiratory symptoms in infants up to 22 months of age in a longitudinal study of infants from Liverpool. The LRSQ had acceptable to good internal consistency for all domains in the LBBS.

Appendix

Appendix 1- Search terms used for the literature review of birth cohort studies

A literature review was conducted between September 2016 and January 2017. Database 'Scopus' was used with unrestricted epoch. Reference lists of papers were checked to find National, International and Regional Birth Cohort Studies. Cohort study findings with regards respiratory disease in preschool children was identified using the cohort profile papers.

	Search Terms	Results
1	Birth Cohort Study	39,460
2	British Birth Cohort Study	1,141
3	Respiratory Birth Cohort Study	2,995
4	Respiratory British Birth Cohort Study	64
5	Preschool Birth Cohort Study	7,380
6	Preschool Respiratory Birth Cohort Study	897

Appendix 2- Summary of Cohort Studies found during literature review; analysis of study design and key findings.

National British Cohort Studies

The 1946 British Cohort Study

The 1946 British Cohort study was initiated by the Royal College of Obstetricians and the Population Investigation Committee to investigate the falling national fertility rate and the impact of the cost of parenthood. It also aimed to determine the distribution of obstetric and midwifery services nationally and determine their impact on the health of mothers and infants and the reduction of premature deaths. The study was funded by the Nuffield Foundation and The National Birthday Trust fund (134, 186). Data collection and analysis is now in its 70th year and continues (187). Those born between the 3rd and 9th of March 1946 in England, Scotland and Wales were eligible for recruitment (n=16,695). 13,687 (81.98%) mothers completed a questionnaire at delivery. Subsequently 5,362 children (39% of the enrolled cohort) have been followed up. The sample of children was geographically distributed in line with national population; analysis has shown shows that the population of the cohort is similar to the national population (134).

In the first four years 12.6% of the cohort was lost. This increased to 19.9% by 15 years, 36.2% by the end of 35 years, and 43.2% by 53 years. These losses were due to death, refusals, living abroad, emigrations and being unable to trace participants. At 53 years 73% of the participants had given information at 17 or more of the 20 data collection points. Only 7% had taken part in ten or less. The study holds high response rates considering the length of its follow up. The study maintains annual contact with the cohort using birthday cards. This has allowed changes in address or contact information to be updated throughout the study (134).

The study found benefit from health visiting and infant care services. The study described the geographical and socioeconomic variation in cost of childbirth, health and survival (134). Subsequently, data has been collected from 5,362 of these children matched to the characteristics of the national population. Other smaller samples of the cohort have been followed including a sample of first born offspring and women between 47 and 54 years of age (186).

Poor housing, parental respiratory disease and air pollution were associated with increased respiratory disease in the first two years of life. Risk factors associated with respiratory disease in the cohorts first offspring were been born to manual working class parents,

parental respiratory disease and parental smoking (57). Deoxyribonucleic acid (DNA) was obtained from the cohort at 53 years which found that that carriers of the gene SERPINA1 Z and S (who have low alpha-1 antitrypsin levels) had increased risk of LRTI in infancy but this had no effect on adult respiratory health (188).

The 1958 British Cohort Study

The 1958 British Cohort Study aimed to identify factors linked to the non-improving national stillbirth rate. It was funded by the National Children's Bureau. All mothers that gave birth in England, Wales and Scotland during one week in 1958 were eligible. 98.76% (n=17,416) of eligible mothers gave access to their maternal health records and completed a questionnaire (135).

Attrition in the 1956 birth cohort was relatively low. Of the 17,416 participants in the cohort 99.3%, 71.1% and 51.8% were successfully contacted at 7, 23 and 55 years respectively (135, 189). Children were located during follow up throughout childhood using their schools. Follow up at 23 years used the subject's most recent address for contact (169). Data collection at 23 years was significantly smaller due to participants leaving school, making contact more difficult. Attrition was more common in those from disadvantaged backgrounds, poor readers, those with short stature, non-whites and children in social care, so the retained cohort is no longer representative of the population (135, 169). At 42 years, refusal had resulted in 13.2% of the cohort's drop out (135).

The study found that smoking was associated with stillbirth, neonatal mortality and reduced birth weight (190). 35% of childhood wheeze resolved by early adulthood, however 5% of the cohort had persistent symptoms and the remainder had intermittent symptoms. Recurrence of wheeze was more common in active smokers. Pneumonia, asthma and wheezing before seven years of age was associated with a cough at 23 years (169). Wheezing between the age of seven and 16 years was associated with childhood pneumonia, eczema and hay fever, whereas wheezing between the ages 17 and 33 years was associated with personal smoking, hay fever and maternal smoking (191). A history of childhood pneumonia was associated with a lower forced expiratory volume in one second (FEV1) and forced vital capacity level (FVC) that is not reversed by salbutamol (192).

The 1970 British Cohort Study

The 1970 British Cohort Study was established in 1970 and underwent its tenth sweep in July 2016 (193). It was funded by the National Birthday Trust and Royal College of Obstetricians and Gynaecologists. The 1970 British Cohort Study set out to examine the association

between maternal characteristics and neonatal morbidity (136, 193). Later, sweeps aimed to assess the factors determining child health, development, transitions into adult life and subsequent health, abilities, attainment and employment in adulthood (136).

All born during one week in April 1970 were eligible to join the study. Eligible births were assessed by midwives using a parental interview and a physical examination. Recruitment led to the enrolment of 16,571 births in the study, 95.9% of the eligible births. Those born in Northern Ireland were dropped from the study after the initial questionnaire (193).

Cohort members were traced for follow up, using school records, at 5, 10 and 16 years 78.9%, 89.8% and 70.1% of the original sample responding at each time point. The increase in responses between five and 10 years was due to immigrants born in the recruitment period being added to the study (135, 136, 169, 193). Although this has allowed the investigation of the effects of immigration on health, the addition of immigrants to the cohort has changed the cohort's demographics. Caution should therefore be taken when assessing trends in this study over time. After 16 years there was a large drop out (54.3% of the original cohort remained in the study at 26 years) due to participants leaving school, and therefore becoming harder to trace (136).

Infants who were male, had mothers who smoked and had a low birthweight were more likely to wheeze in the first five years of life (46). Maternal smoking was associated with a higher incidence of bronchitis and hospital admissions for LRTI during the first five years of life and increased the incidence of wheeze by 1.8% in the first 10 years of life (194). The incidence of wheezy bronchitis had a dose response relationship with maternal smoking; the incidence was 14% in children whose mothers smoked four cigarettes a day compared to 49% in those who smoked 14 cigarettes a day (195). 80% of children who had wheeze in the first five years of life (46). A higher incidence of wheeze before five years increased the risk of continuing to wheeze at 10 years but there was no association between the age of first attack and prognosis (196). A damp home environment was associated with increased cough and phlegm production at 10 years of age (56). Those born under 2kg were nearly twice as likely to be diagnosed with asthma at 26 years when compared to those born between 3kg and 3.5kg (197). Wheezing was reported more often in those who were not breastfed; this relationship did not persist when adjusted for parental asthma (198).

The Millennium Cohort Study

The Millennium Cohort study began in 2000 and aimed to collect a range of data assessing health, childcare, education and employment and their inequalities. It is funded by the Economic and Social Research Council and the UK Government (137, 199). Children born in England and Wales between 1st September 2000 and 31st August 2001 and in Scotland and Northern Ireland between 24th November 2000 and 11 January 2002, alive at nine months and eligible to receive child benefit (as identified by the government child benefit records) were eligible to join the study (137, 199). This criteria and recruitment strategy has given almost universal coverage of the UK population (missing only asylum seekers). The use of a yearlong recruitment period compensates for seasonal variation in the outcomes of birth. The study sample was representative of the UK population (137). Recruitment led to a cohort of 18,827 children at baseline. Hard to research populations were intentionally over sampled to ensure an adequate sample size for analysis. Analysis of these small sub populations is a strength of the study, however this disproportionate sampling needed adjusting for during analysis of the whole population (137).

Follow up occurred in participants who remained in the UK at three, five, seven and 11 years. Response was 82.8%, 82.2%, 73.5% and 70.6% at each sweep of data collection respectively (137). Attrition rates are relatively low considering the time period of the study. Dropout was due to non-response, refusal, non-contact, emigration and death. Dropout was higher in ethnic minorities and disadvantaged areas (137). This disproportionate attrition was adjusted for during analysis.

Data was collected using Computer Aided Self Completion Interviews and assessment by healthcare professionals which has covered a large range of data using standardized measures (199). This combination of data collection has increased the quality of the data and is a strength of the study.

The millennium cohort has led to some key respiratory related publications. Exclusive and prolonged breastfeeding protected against hospitalization for LRTI in the first eight months of life. Breastfeeding for any amount of time showed some benefit (177). It was estimated that 27% of LRTI requiring hospitalization could have been avoided by exclusive breastfeeding and 25% prevented with partial breastfeeding. This protective effect diminished soon after withdrawal of breastfeeding (177). Male sex, maternal age, body mass index (BMI), atopy, smoking during pregnancy, preterm birth, breastfeeding, exposure to other children and household furry pets were independently associated with both early

remitting and persistent wheezing over the first seven years of life (79). Compared to the white population, in the first three years of life, 'Black Caribbean' and 'Bangladeshi' infants reported a higher and lower rate of 'ever asthma' and wheeze respectively. These findings may be due to socioeconomic and cultural differences (171). After adjusting for confounding factors there was a relationship between household income and respiratory health in children (66). Poverty in early childhood was found to be associated with increased risk of asthma attacks and chronic respiratory illness by four years of age (174). Children born to parents with subfertility and who had a longer time to conception were at an increased risk of asthma and used more asthma medications at both five and seven years of age (200). Obese children used more medications for respiratory conditions than children wither lower BMI between five and 11 years of age (201).

The millennium cohort profile has allowed mapping of the trajectories of asthma symptoms in the first seven years of life. This has shown that those with high levels of wheeze and atopic symptoms have an increasing risk of wheeze over the first five years of life that then reduces until seven years. This reduction is less than those without atopic symptoms. In those with low levels of wheeze, asthma symptoms reduce from birth to seven years, more so in the first five years of life (202). Between the 1970 cohort and Millennium cohort, there has been an increase in allergic conditions, including asthma (203).

Regional British Cohort Studies

Aberdeen Children of the 1950's Cohort Study

The Aberdeen Children of the 1950's study was established and funded by the American Association for the Aid of the Crippled Child to research the cause of cognitive problems in childhood. Aberdeen was chosen due to its high standard of maternal record keeping in the Aberdeen Maternal and Neonatal Databank (204). This contains research standard records on around 85% of births in Aberdeen (138, 205). Aberdeen also routinely collects data from home visits, school records, childhood growth data, immunizations data and social background data. Through the study birth and obstetric data was combined with an assessment of height, weight, cognition, family socioeconomic circumstances, attitudes, and behaviours and a teacher's assessment of behaviour. Mortality data, hospital admissions, cancer registrations were also included. Further follow up of the cohort has been designed to look at various specific hypothesis (138, 205).

All children born in Aberdeen between 1950 and 1956 who were in primary school in 1962 (aged between six and 12 years of age) and their siblings were eligible for the study. 99% of

the eligible population of Aberdeen was traced during initial data collection (n=12,013), along with 7,080 second generation births. Response rates of 64% to the postal questionnaire in 2002-2003 were achieved. Response was highest amongst females with married, non-manual working class parents and fewer siblings. There were 611 deaths in the cohort by the end of September 2005 (138).

This cohort is almost a complete population of a well-defined area with high rates of follow up in the early stages. This was achieved by using the Aberdeen Maternal and Neonatal Databank. This is a unique feature of the study that increased the quality of the data. This, along with the use of Christmas cards, was used to track changes of address to increase participation throughout the study period (138). The cohort has given a large insight into cognition, the natural history of coronary heart disease and mental health (138, 205), but there were no publications on respiratory disease in preschool children.

Avon Children of the 90's Study

The Avon Longitudinal Study of Parents and Children (ALSPAC), also known as "Children of the 90's", is part of the European Longitudinal Study of Pregnancy and Childhood (ELSPAC). ELSPAC was established by the WHO. It was designed to investigate how phenotypes combine with environmental factors to influence health and development of children and their mothers. The Medical Research Council, the Wellcome Trust and the University of Bristol fund the study. Data collection is now in its 25th year (162, 206).

14,541 pregnancies were opportunistically recruited by midwives between 1st April 1991 and 31st December 1992 from a defined area in South West England. Pregnant women who moved to the area were eligible until the point of delivery. Those who moved away and had not completed a questionnaire in the third trimester were excluded (139, 162, 207). Children born during the recruitment period that were missed were added to the study at seven and eight years of age. This has led to 15,390 children being enrolled in the study by eight years.

Enrolled mothers were more likely to be white, own a home and car, be married and live in overcrowded house when compared to census data, despite having a higher average socioeconomic status than the population of Britain (139). The children in the cohort at 16 years were more likely to be female, of white ethnicity, higher educated and less likely to be eligible to free school meals than the national population (162).

Attrition was highest during the children's infancy and increased again as the participants entered adulthood, however because children were free to re-join the study up to 18 years of age the cohort population has remained fairly large. Of the 14,062 eligible live births 97.4%

and 84.4% were followed up at two years and 18 years respectively. 100 children had died, 2018 were untraceable and 782 withdrew from the study. During adolescence only 48.2% responded to all 12 data collection sweeps. 75% of the cohort had responded at least once (162).

Over 1200 papers have been published using data from this cohort. Babies that sleep prone and on their side had an increased incidence of respiratory symptoms at six months (208). This relationship was no longer apparent by 18 months (175). An important finding that led to change in National guidance was that Sudden Infant Death Syndrome was reduced in babies who slept on their back (209). Children of mothers who had greatest use of household chemicals in pregnancy were at an increased risk of persistent wheeze and lung function abnormalities (210). Maternal smoking in pregnancy increase infant wheezing during early childhood, regardless of postnatal ETS exposure (43). The prevalence of wheeze was also higher in children born to mothers that consumed paracetamol in the third trimester (211). The prevalence of asthma was higher in children whose mother had anxiety in pregnancy (212). Wheezing and eczema was more prevalent between 30 and 42 months of life in those who were exposed to excessively clean environments after adjustment for all confounding factors (67). The incidence of asthma was higher in seven and eight year olds with a lower socioeconomic status (65). The Filaggrin gene was associated with asthma, eczema and sensitization to allergens (213).

Born In Bradford

The BiB Cohort Study began in 2007 in response to the high rates of child morbidity and mortality in Bradford. It aimed to identify modifiable risk factors that cause or contribute to ill health and produce health promotion interventions in this deprived multi-ethnic population. The study collaborates with the European Child Cohort network, which allows links with other national and international birth cohort studies (49, 163, 214). The largely biethnic cohort has allowed comparisons of health outcomes to be made between two ethnic groups (49).

All births between March 2007 and November 2010 at the Bradford Royal Infirmary were invited into the study. The recruitment rate was 64% (13,776 pregnant woman and 3,448 of their partners). The recruited population had a lower proportion of younger mothers and a higher proportion of South Asian and nulliparous mothers. There were differences in gestational age, however this was due to the recruitment strategy of the study because those

born before 28 weeks gestation were not recruited. Despite this the cohort was shown to be representative of the population in Bradford making the cohort valid (164).

The BiB study has a good relationship with the participants and the hospital. This relationship was maintained using birthday cards, newsletters, and a study website (49). The BiB cohort was split into many sub cohorts during investigation. The BiB1000 sub-study showed response rates of 77%, 75%, 74%, 70% and 70% at 6, 12, 18, 24 and 26 months respectively with 47% of the cohort completing all follow up. The majority of attrition in this cohort was due to not being able to contact the participants even with the use of telephone calls, reference to electronic health records and opportunistic visits to family homes (215).

There have been around 140 publications produced from this cohort. Maternal smoking (16.4%) and drinking (30.6%) rates in pregnancy were relatively low compared to white British mothers (33.7% and 67.4% for respectively). While babies born to mothers of Pakistani ancestry were lighter and they were less likely to be exposed to risk factors of sudden infant death syndrome (49).

The study has shown that between birth and three years of age the incidence of wheeze is higher in boys than in girls and low birth weight increases the risk of wheezing disorders (25).Rapid growth between three and 12 months of age increased the risk of wheeze (55).

The Isle of Man (IoM) Cohort Study

The IoM Birth Cohort study began as part of the ELSPAC study. The aims specific to the IoM study were to measure the prevalence of diseases in children and their parents, detect health advantages and disadvantages, to study the effects of migration on the population and to determine any island specific features of that may influence health (140). After a period of publicity on the Isle of Man, every mother expecting a baby between 1st January 1991 and 30th June 1992 was invited to join the study (n=1,384). This led to 1,314 live births being enrolled in the study. Mothers leaving the island (n=161) were not followed up, however children who had moved to the island born in the recruitment period were opportunistically added to the study at seven years. This led to 1,411 participants available for follow up at seven years. Between 15 and 16 years, 1,125 participants were available for follow up (140, 216).

The main source of attrition in this cohort was emigration off the island. This large proportion of drop out is a particular weakness of the study however the publicity for the study and its robust data collection methods have produced near 100% response rates (140). The inclusion criteria allows for the analysis of the whole island population and comparison between the

Manx Population and those who have immigrated to the island; this adds a unique element to this study. The high participation rate in this cohort shows the benefit to publicity and structured recruitment when compared to opportunistic recruitment, such as used in the AVON Cohort.

Funding by various charitable organisations on the island has not been sufficient. Although data is complete it is not ready for analysis (140, 216). Because of this, no publications assessing the IoM alone have been published. There are however, publications comparing the IoM Cohort with the ALSPAC cohort (217).

The Newcastle Thousand Baby Study

The Newcastle Thousand Families 1947 Birth Cohort was established in response to the high infant mortality rate in Newcastle. The original aim of the study was to identify risk factors of infection in infancy and has continued as a longitudinal study to assess health, education and family life throughout childhood (141, 218). It was funded by the Newcastle City Health Department, the City Health Committee and the Nuffield Foundation (141, 218).

The cohort consisted of 1,146 (84%) children born in May and June of 1947 in Newcastle upon Tyne who were followed through their first year of life. The original cohort was representative of the social spectrum of the city (141). Data collection began in pregnancy using midwife's reports and continued every six weeks throughout infancy and quarterly up until five years of age. Follow up included assessments by paediatricians and health visitors. The cohort was tracked by NHS central registry enabling mortality figures to be updated. This added accuracy to figures reported in publications. The cohort was also linked to GP and hospital records (141, 218).

In the first year of the study 44 participants died and 125 participants left the study area. 73.9% and 65.4% of the cohort were followed until the end of the 5th and 15th year of life respectively. At 50 years 89% of the surviving cohort were traced; 50.1% returned a questionnaire and 40.0% (n=412) attended for a clinical examination (141). Although these are high attrition rates compared to other cohort studies the most recent follow up showed that the cohort was representative of the original sample for the majority early life factors. The sex of the cohort is no longer representative of the original sample (141). The high attrition rates, along with a fairly small sample size reduce the strength of this study as a whole.

There is very little published work addressing paediatric respiratory health from this cohort, however work has been published on long term respiratory function and of overall infant's

health. Key findings included: Children from poorer families with household over-crowding were at the highest risk of respiratory infection (58). Children who had a severe lung infection before five years of age were seven times more likely to develop chronic lung disease in adulthood (58). Breastfeeding had no long-term beneficial effects on adult lung function (219).

The Southampton Women's Cohort

The Southampton Women's and Children's survey was established in 1998 by the University of Southampton and the Medical Research Council to investigate the effects of dietary and lifestyle factors on the health of mothers and their children. The study now follows the children born into the study to determine pre-natal risk factors for chronic disease (142, 176).

Between 1998 and 2002 GP's in Southampton recruited female patients between 20 and 34 years of age by sending them a leaflet which was followed up with a telephone call. A trained nurse performed the initial assessment of interested women at their home. To maximise response rates, a repeat letter was sent to those who did not respond to the telephone call. If there was still no response the home was visited. The study was publicised locally at events and in supermarkets to increase participation. These efforts led to at 75% participation rate (142, 176). 12,583 women gave pre-pregnant profiles and were followed throughout pregnancy. Participants and their GP's were asked to contact the study team if they or their patient become pregnant. This recruitment strategy led to 3,158 babies available for follow up (176). Response rates were 95%, 93%, 86% and 81% at six months and one, two and three years respectively (142).

The study has a small respiratory specific sub-study of 150 babies who underwent lung function testing, by means of rapid thoracoabdominal compression, at six weeks of age and completed symptom diaries in the first year of life. Urine samples were obtained from the cohort of infants and skin prick testing was conducted on their parents. Dust samples were collected from participant's homes. (142, 176).

This sub-study found that greater maternal adiposity was associated with increased transient wheeze in childhood, however it had no association with asthma or allergy (220). Babies who had smaller birthweights and those who had high weight gain postnatally had poorer lung function at six weeks of age (176). Greater relative weight and adiposity gains were associated with increased wheeze at three years (54). Poorer lung function at six months was associated with a higher incidence of wheezing in the first five years of life (221).

Breastfeeding was shown to be protect against respiratory disease. This protective effect was proportional to the duration of breastfeeding (7).

This cohort has a good participation produced a large sample for a regional cohort study. However there was a small age range invited to the study so a large proportion of the pregnant population was missed. Likewise the design of the study meant that those who were not registered with a GP or don't attend GP's regularly were less likely to be invited to the study. These are significant sources of selection bias. Although the use of GP's will have reduced the drop out, there was likely to be a significant number of women who forgot to or intentionally did not contact the study team when they become pregnant (142, 176).

International Collaborative Birth cohort studies

The European Longitudinal Study of Pregnancy and Childhood (ELSPAC)

The European Longitudinal Study of Pregnancy and Childhood (ELSPAC) was established by the WHO in Europe in 1985 to identify the factors influencing health in childhood in European Countries (222). It is coordinated in Bristol by the ALSPAC team (206). The aims of the study were to determine the biological, psychological, social, economic and environmental factors that influence pregnancy, birth, health and development (223). The study was established in the Isle of Man, Czech Republic, United Kingdom, Slovakia and Ukraine and partly in Greece, Spain, Croatia and Estonia. It was agreed to follow the cohort until 21 years of age and potentially beyond, however poor funding in various countries has meant that continuing data collection was not possible (224). Recruitment occurred in Great Britain, the Isle of Man and Czech Republic between 1991 and 1992 and varied from 1992 to 1995 in the remaining countries (223, 225). The cohort contains over 40,000 children and their families and at 17 years the study participation rate remained relatively high considering the time consuming nature of the questionnaire at around 50% (225).

To date there has only been a few publications produced as a result of the ELSPAC study. The majority of publications have only included data from a single country. This is disappointing considering the scale and the potential impact of collaboration in this study. Findings from the ALSPAC study and Isle of Man Cohorts have been discussed earlier.

Respiratory Specific Birth Cohort Studies

Manchester Asthma and Allergy Study (MAAS)

The Manchester Asthma and Allergy Study (MAAS) was established in 1995 to understand the development of asthma and allergies in young children. 3,618 mothers and 2,172 fathers were originally recruited at routine antenatal booking appointments. Parents underwent skin prick testing to common allergens to divide the cohort of infants into risk groups, each of which had different study aims (24, 143, 226). There was no comment on of the cohort's representation of the eligible population of Manchester.

1,499 parents consented to the study, which led to 1,085 children being enrolled into the study (2). 6.9% were lost to follow up in the first year and 8.3% were lost by three years. 969 children (89.3%) remained in the cohort at five years (226). Although the sample size is not particularly large compared to some other cohort studies, attrition rates were low over the first five years of the study. Follow up was by means of symptom questionnaires at one, three and five years assessing asthma, eczema, rhinitis and medication use and by objective measures of lung function at three and five years. GP notes were reviewed for formal diagnoses (24, 143, 226). The use of risk-stratified groups by means of skin prick testing is a unique aspect to this study.

Children in the 'high risk' group who wheezed had a significantly worse lung function at 1 month of age (227). Persistent wheeze after three years was predicted by poor lung function (228). Those children whose environment was controlled to reduce allergen exposure, used significantly fewer asthma medications than those whose environment was not controlled (229). Sensitisation to allergens was associated with a higher prevalence of asthma (228). Dust mite exposure was associated with increased bronchial airway reactivity regardless of prior sensitisation to dust mites, though asthma was more severe in those who were sensitised (78). Similarly, sensitization to dogs and cats was associated with asthma (230). ETS was associated with an increased prevalence of asthma (230) and increases the risk of wheezing in the first year of life (47). Nursery attendance was associated with reduced wheeze at five years. This association was strongest in those who attended nursery between six and 12 months of age (73). Significantly more wheezers between three and five years of age had been prescribed antibiotics in the first year of life for LRTI when compared to non wheezers. This may indicate a roll for early antibiotic exposure in the aetiology of wheeze (231). Exposure to particulate matter and nitrogen dioxide was associated with small but significant reductions in lung volume growth in children (232).

The Prevention and Incidence of Asthma and Mite Allergy (PIAMA) Birth Cohort Study

The Prevention and Incidence of Asthma and Mite Allergy (PIAMA) Cohort was established in response the increasing prevalence of asthma in the Netherlands in the 1980's and 1990's. (144). Recruitment ran between 1996 and 1997 in prenatal clinics in North, Middle, and Southwest regions of the Netherlands. Mother's atopic status was assessed using a screening questionnaire which allowed their child to be grouped by risk of atopy. This led to 3,963 children (~50% of those invited) being enrolled into the study. Children were split across two sub studies; one investigating the effects of dust mites and the other investigating the natural history of respiratory disease (233). Mothers of the PIAMA study were more highly educated and less likely to be from a non-western population than the population the Netherlands. They were however, just as likely to breastfeed and had a similar maternal age (144). The children were followed up with 10 questionnaires and medical examinations in the first eight years. At eight years 92% of the participants were still involved in the study. In 2012 and 2013 participants underwent anthropometric measurements, lung function tests, and blood tests; 800 (around 20%) of the original cohort participated (144, 234).

House dust mite impermeable covers provided a reduction in asthma symptoms for the first two years of life (235). Asthma was increased at eight years in children whose mother ate peanuts daily during pregnancy (236). Birth by caesarean section conferred a higher risk of asthma at eight years, more so in those with allergic parents (237). Lower birth weight increased the risk of respiratory symptoms in the first five years of life and this relationship was strongest in those exposed to ETS (238). Breastfeeding reduced asthma at eight years of age regardless of a family history of atopy (178). Exposure to microbial contaminants in early life may have reduced asthma and wheezing in the first four years of life (239). Frequent consumption of products containing milk fat reduced the risk of asthma symptoms in the first three years of life (240). Nursery attendance increased the risk of LRTI in the first year of life (184) and was associated with an increase in respiratory symptoms until four years of age but fewer respiratory symptoms thereafter (74). Air pollution was associated with an increased diagnosis of asthma in the first year of life (241) and at eight years of age (242). In contrast to previous thoughts, the PIAMA study described that those with nocturnal dry cough in early childhood were nearly twice as likely to be diagnosed with asthma at school age (243).

The Tucson Children's Respiratory Study

The Tucson Children's Respiratory Study was established in 1980. It examined the relationships of risk factors in the first three years of life, including LRTI, with chronic lung disease in childhood and early adulthood. The study recruited 78% (1,246 children) of all those born in Arizona between May 1980 and January 1984 (244). Participating mothers tended to be married, older, have spent longer in education, have more household members with a FH of atopy and have other older children. All other demographics were comparable to the population of Arizona (244). Drop out was 13% in the first five years. 78% of the cohort

remained traceable during 2002 and 60% of these were still living in Tucson. These are relatively low attrition rates considering the length of tim(49, 139, 140)e the study has been running (45).

A large amount of data assessing risk factors was collected prenatally and biological samples were taken at birth. In the first three year there were four data collection points which included questionnaires, skin prick tests and pulmonary function tests with airway challenge studies. Lung function was measured using rapid thoracic compression, helium dilution, forced oscillation measurements and tidal breathing analysis. During any acute LRTI the participants were asked to contact the research team for further assessment and serology tests. Between three years and 16 years of age, there were a further four data collection sweeps (45, 244). The integration of questionnaires, skin prick testing, objective measures of lung function and formal assessment of RTI's, has created a novel dataset of high quality. The assessment during infections is a novel and unique feature of this study.

LRTI occurred in 32% of infants in the first year of life. 17% and 12% of infants had a LRTI in the subsequent two years. RSV was the most common cause of LRTI followed by Parainfluenza virus (245). Infants who had LRTI in the first three years of life demonstrated reduced lung function at six years and had a higher incidence of asthma at 11 years (45). Breastfeeding for at least one month reduced LRTIs in the first four months of life (173). Infants born to younger mothers and nursery attenders were more at risk of LRTI (173). Infants with mothers who smoked were three times more likely to have a LRTI in the first three years, and the age of their first infection was lower (246).

The proportion of the cohort with an asthma diagnosis increased with every sweep of data collection until 16 years of age. The highest incidence of wheeze in males was between six and eight years of age after which it decreased. Contrastingly the incidence of wheeze in females increased after 10 years. 59% of children that wheezed before three years had stopped wheezing by six years of age (45). The diagnosis of asthma after 11 years of age was not associated with wheeze in the first three years of life, however it was associated with wheezing between six and 11 years. Children who lived with other siblings had reduced incidence of asthma symptoms between six and 13 years (60). Infants who were exclusively breastfed and have mothers with asthma are more likely to have asthma and atopic disease (72).

Infants who wheezed in the first three years of life had lung function measurements indicating smaller than normal airways before their first LRTI. Those who wheezed in the first

year of life and had one LRTI by three years of age on average had reduced lung function compared to those who did not wheeze. These observations form the hypothesis that future wheeze and LRTI may be predicted by premorbid reduced lung function in early childhood (247).

The Tucson study lead to the description of three phenotypes of wheeze. "Transient wheezers" wheezed during the first two to three years of life and had no or few episodes thereafter. They were unlikely to have atopic risk factors and had reduced lung function at three years which improved (but had not completely resolved) by six years (45).

"Non-atopic wheezers" continued to wheeze beyond the third year of life after a LRTI in early life. They had a lower lung function at six years; it was unclear if this abnormality is present before the initial LRTI. It was hypothesised that this wheeze is due to an increased response to a viral infections due to abnormal control of airway tone. This abnormality may reduce with increasing age (45).

"Atopic wheezers" were likely to have their first wheezing episode in the first six years of life. Atopic wheezes had the greatest reduction in lung function between three and six years and had the lowest level of lung function at six years of age (possibly due to ongoing damage to the lungs). They had a high level of serum IgE and were more likely to have an atopic mother (45).

Appendix 3- Summary of Large Non-Birth Cohort Respiratory Studies

The European Community Respiratory Health Survey (ECRHS)

ERCHS was established in 1990 in response to the increasing prevalence of asthma in Europe in the 1980's. Currently there have been three phases of the study (248).

ECRHS 1 included 600 participants aged between 20 and 44 years, randomly selected from 56 centres across Europe. Those who responded to an initial questionnaire were invited for detailed respiratory assessment (including a questionnaire and spirometry with bronchial response tests) (249). The questionnaire included 71 items assessing respiratory symptoms, exposure to risk factors of repository disease, the use of health services and medication use. The questionnaire was based on two validated respiratory questionnaires; the International Union against TB and Lung Disease questionnaire and the American Thoracic Society questionnaire (249).

ECRH2 was a nine year follow up study which aimed to determine the incidence and prognosis of allergy, asthma, hay fever and eczema, describe the distribution of exposures and risk factors, and identify sub groups that are more susceptible to the diseases in question. ECRH2 also established a blood bank for DNA extraction (250). The questionnaire was modified to include questions from the ISSAC questionnaire and to represent the Global Initiative for Asthma guidelines more closely (250).

ECRH3 was a 19-year follow up study. It aimed to describe the change in respiratory symptoms over time. It assessed patterns of IgE sensitisation as the participants aged and if atopic status and asthma are associated with reduced lung function and Chronic Obstructive Pulmonary Disease (COPD) development. It also described associations between obesity, physical exercise, asthma and lung function (248).

The ECRHS has shown that there is a large variety in the prevalence of asthma throughout Europe. It is highest in English speaking countries and lowest in the Mediterranean and eastern European regions (251); this correlates with warm winters and cold summers (252). There has been an increase in the diagnosis of asthma attacks and medication use but not in the prevalence of symptoms over this time. This could be due to better diagnosis leading to more people being labelled as asthmatic, or the increased use of effective treatments (253). The ECRHS has strongly supported the hygiene hypothesis of asthma and allergy; children with pets had less atopy and hay fever, and it was beneficial to grow up on farm (251). Children who had early exposure to other children, larger families, shared a bedroom and owned a dog in childhood were less likely to become atopic in adulthood (254). Other factors that were associated with increased asthma in adulthood included indoor mould growth (255), passive smoking (256) and parental smoking, low maternal age, severe respiratory infection in early life, and long term dog keeping (257). It has been shown that respiratory symptoms have a detrimental effect on the QoL (258).

The International Study of Asthma and Allergy in Childhood (ISAAC)

The ISAAC aimed to describe the worldwide variation in prevalence and severity of asthma, rhinitis and eczema, obtain baseline measures of these diseases and provide a framework for future aetiological studies of environmental, lifestyle and genetic studies of these diseases (105).

The study had three phases:

- Determining the (variation of) worldwide prevalence of diseases in six to seven year olds and 13 to 14 year olds (105).
- 2- Investigate possible aetiological and risk factors for the diseases (259).
- 3- Repetition of phase 1 to assess trends in severity and prevalence (260).

The study has shown that there is worldwide variation in the prevalence of the diseases in question, which worldwide variations of smoking and air pollution cannot be made responsible for (261). Population level variation in susceptibility to allergic sensitisation may party be responsible for the variations in prevalence. Countries with an incidence of Tuberculosis showed a lower prevalence of asthma (75). The prevalence of asthma was lower in rural areas than urban ones, being lowest amongst farmers (75) and in households with higher levels of dust endotoxins (64). In contrast, there was an association between pertussis and measles and asthma and rhinitis (262). Respiratory symptoms were shown to be higher in damp households (263). Asthma was associated with an increase in indoor humidity and was lower in areas with higher altitude and a larger variation in temperature (63). Eczema correlated positively with increasing latitude and negatively with increasing temperature (63).

Diet may also play a role in the aetiology of asthma. Areas with higher trans fatty acid consumption showed higher levels of asthma, rhinitis and eczema (75). In contrast, areas with higher fish, fruit and vegetable consumption, a 'Mediterranean diet', showed lower levels of wheeze (76). Obese and overweight children were shown to have higher levels of wheeze and increase airway obstruction (264). Breastfeeding was shown to be protective of wheeze in non-atopic children and this was most apparent in non-affluent areas. There was no association shown between breastfeeding and allergy (265). Phase three has shown a

worldwide increase in the prevalence for all of the diseases in question in children aged between six and seven years. However asthma was reduced in children aged 13 to 14 years of age, more so in areas with a high original prevalence (260, 266).

The Leicester Respiratory Cohort

The Leicester Respiratory Cohort was set up in 1990 due to the lack of respiratory studies in preschool children. It aimed to determine the prevalence, natural history, risk factors, phenotypes and trends of asthma and other respiratory disorders in preschool children. Leicester's large South Asian population allowed detailed study of respiratory disorders in this ethnic minority (108).

Two cohorts were selected at random from the Leicestershire health authority child health database. The 1990 cohort contained 1,650 children born between 1985 and 1990 and the 1998 cohort consisted of 8,700 children born between 1993 and 1997 (108). The selection was random and the sample was stratified so the cohort contained selected numbers of children of each age group up to four years old. The cohorts were shown to be representative of the population of Leicestershire (108) although the participants that responded tended to be of higher social economic class and have a higher maternal age (108). Participants were sent questionnaires base on validated questionnaires at baseline and every few years thereafter and were invited for physical examination in the early school years. Baseline data was 100% due to the use of routine data. Response rates for the 1990 cohort were 86.2% in 1990, 79.1% in 1998 and 57.3% in 2003. For the 1998 cohort there were 78.3% in 1998, 59.6% in 2001 and 48.6% in 2003. The high attrition rate was due to change of address with nearly half of the participants moving house at least once during the study (108, 109, 267, 268). Around 60% of those invited for laboratory studies attended (108).

In 1990 11% of the cohort had a doctor diagnosis of asthma, 16% wheezed with 13% of these having repeated attacks; in 1998 the percentage of the cohort with doctor diagnosed asthma and wheeze was 19% and 29% respectively (268). 10% of children snored regularly and nearly half of these continued to do so at four years (269). It was shown that fewer than half of preschool wheezers continue to wheeze at school age and those with a chronic cough tend not to wheeze two to four years later (270). In contrast to what was previously though, vaccinations were not associated with an increased risk of wheeze and appeared to be protective against late onset wheeze. It has been hypothesised that allergen induced wheeze and infection induced wheeze may be completely different entireties (271). Breastfeeding was associated with better lung volume in those born to asthmatic mothers. Those that were

breastfed for four months had better peak flows irrelevant of a FH of asthma (272). This study has helped to develop a simple tool for identifying the risk of continuing to wheeze at school age in preschool wheezers (273). The South Asian population had less wheezing than the white population but had a higher rate of admission to hospital (172).

The Wythenshawe Community Asthma Project (WYCAP)

WYCAP is a long-term prospective observational study of respiratory illness and asthma which observes two cohorts from two GP practices in Manchester. It was designed to develop a method for identifying those with asthma and COPD in the community both known and unknown to medical services and to evaluate the cost of treating these individuals (274, 275).

Three postal respiratory questionnaires were sent to 11,206 patients between 1993 and 1999. The questionnaire was based on the European Community Respiratory Health Questionnaire with additional questions to assess FH of asthma and current smoking. Participants were likely to have asthma if they scored four or more. Non responders were sent two reminders four and eight weeks after the initial questionnaire (274).

13.8% of patients aged 16 years or over had asthma like symptoms and around half of these were unknown to health services and may benefit from further investigation or management (274). The questionnaire showed a good positive predictive value in identifying those who clinicians thought would benefit from a trial of medication (275). The workload of assessing these unknown patients would be unfeasible and a method of prioritising patients was successfully established using the 1995 and 1999 data (276). It was estimated that 10% of the children might have had asthma that was unknown to medical services. The questionnaire was deemed to be valid in identifying children who may have asthma and may benefit from clinical review in the five to 15 year old population when compared to expert opinion (277). The number of children that would require clinical review was between 3.5% and 8%; this would significantly increase the workload of GP practices (277).

It has been shown that from 1993 to 1999 the prevalence of wheeze, being woken by cough and the recipient of asthma medication has increased in adults but decreased in children; this was more likely to be due to an increase in COPD in adults rather than the reduced asthma in children (278).

Appendix 4- Birth cohort studies not discussed in the current thesis

Study	Country	Year	Cover	Participants	Outcome Aims
All Babies in South East Sweden (279)	Sweden	1997	Regional	17055	To determine the in importance of environmental factors in diabetes development, how genetic and environmental factors interact, whether interventions change lifestyle and reduce obesity and diabetes, and build a biobank for studies on diabetes, obesity and immune mediated diseases.
ELSPAC in The Czech Republic (224)	Czech Republic	1991	Regional	7589	To determine Effect of the socioeconomic change in the country in 1989 on health.
Generation R (280)	Netherlands	2002	Regional	9778	To identify early environmental and genetic causes leading to abnormal growth, development, and health during foetal life through to adulthood.
German Multicentre Allergy Study (281)	Germany	1990	Multicentre Regional	1314	To assess the role of different exposures (mite allergen, cat allergen, ETS, infectious disease, vaccinations) in the development of asthma.
Growing Up In New Zealand (121)	New Zealand	2009	Regional	6822	To identify the determinants of developmental trajectories and what optimises these trajectories. To assess development and its association with ethnicity.
Growing Up In Scotland (282)	Scotland	2004	National	14000	To generate Scottish specific data about outcomes through childhood and into adulthood across various domains.
The Amsterdam Born Children and their Development (283)	Netherlands	2003	Regional	8226	To assess associations between maternal lifestyle, medical, psychosocial and environmental conditions during pregnancy and child's health at birth and later life.
The Bern Infant Lung Development Cohort (284)	Switzerland	1999	Regional	364 (2009)	To study the effects of environmental and genetic risk factors on the physiological properties of the developing lungs in healthy individuals.
The Danish National Birth Cohort (285)	Denmark	1996- 2002	National	96986	To determine cause of reproduction failures, pre/perinatal determinants of childhood health and improve infrastructure for future studies.
The Dunedin Multidisciplinary Cohort Study (286)	New Zealand	1972	Regional	1037	To study the nature and prevalence of some developmental and health problems in three year olds and some risk factors for these problems.
The Helsinki Birth Cohort (287)	Finland	1934	Regional	13345	To assess the influences of early growth and living conditions on long term health, including cardiovascular disease, psychological disease, aging and cancers.
The Hong Kong Chinese Birth Cohort (288)	China	1997	Regional	8327	To examine the effect of ETS exposure and breastfeeding on infant health and health care utilisation in the first 18 months.
The INMA Study (289)	Spain	1997- 2008	Multicentre Regional	3887	To describe prenatal exposure to polluting agents in Spain and to evaluate impact of pre and post-natal exposure to polluting agents and the effects of genetics and nutritional factors.
The Krakow Birth Cohort Study (290)	Poland	2002	Regional	493	To determine the effects of PAH components on foetal growth and early childhood health.
The Mater-University Study of Pregnancy and Childhood (291)	Australia	1981	Regional	7223	To investigate the factors associated with adverse pregnancy outcomes.
The Mother and Children's Environmental Health Study (292)	Korea	2006	Multicentre Regional	892	To determine the roles of bio-aerosols, heavy metals and endocrine disrupting chemicals in adverse birth and health

					outcomes. To assess the interaction between environmental factors and genes in disease development.
The National Children's Study (293)	USA	1999	Multicentre National	5608	To study the process of human development from conception to old age.
The North Finland Birth Cohorts (294)	Finland	1996 1985	Regional	1966- 12231 1985- 9479	To examine risk factors for preterm birth and intrauterine growth retardation and subsequent morbidity.
The Norwegian Mother and Child Cohort Study (295)	Norway	1999	National	110000	To build a knowledge base on the causes of serious disease.
The Pelotas Birth Cohort Study (296)	Brazil	1982	Regional	5914	To assess the effects of breastfeeding, and other chronic disease risk factors.
The SELMA Study (297)	Sweden	2010	Regional	2582	To investigate early life exposures during pregnancy and infancy to environmental factors with focus on endocrine disrupting chemicals.
The STEPS Study (298)	Finland	2008	Regional	1827	To determine the precursors and causes of reduced child's health and wellbeing.
The Stockholm Birth Cohort Study (299)	Sweden	2004	Regional	1953- 15117 1980-13838	To study intergenerational life course associations with health disease and mortality.
The Swiss National Cohort (300)	Switzerland	2006	National	7452075	To build and maintain the first nationwide cohort with data linkage and develop measures of socioeconomic status and procedures to identify deaths.
The Toyama Birth Cohort Study (301)	Japan	1990	Regional	8274	To study of lifestyle and health among all children.

Appendix 5- Adult respiratory symptoms questionnaires

	King's Brief Interstitial Lung Disease (K-BILD) health status questionnaire	K-BILD health status questionnaire was developed to assess the health status of patients with interstitial lung disease. It is a 15 item questionnaire which assesses three domains; breathlessness and activity, psychological and chest symptoms. It has been shown to be a valid self-completed questionnaire for the assessment of health status in patients with interstitial lung disease (302). A minimal informant difference has been determined for the response to therapy (303).
ľ	St Georges Respiratory	The St Georges Respiratory Questionnaire is a self-completed
	Questionnaire	questionnaire with 76 items assessing symptoms, their frequency and severity, activity limitation and the impacts on psychological and social functioning. It has been validated, showing good repeatability and sensitivity and is comparable with standardised measures in adults with chronic airflow obstruction (304), asthma (305) and bronchiectasis (306). There has been a COPD (307) and Idiopathic Pulmonary Fibrosis (308) specific versions developed and validated.
	The Asthma Control Test	The asthma control test is a five item questionnaire assessing symptoms, use of rescue medication, and the impact of asthma on everyday functioning (309). It has been shown to be valid and reliable in screening those with poorly controlled asthma in a multinational study (310) and is responsive to changes to asthma control (311).
ĺ	The Chronic Respiratory	The Chronic Respiratory Questionnaire was developed to produce
	Questionnaire	a measure changes of quality of life in COPD patients after treatment. It is a clinician completed questionnaire which assesses dyspnoea, fatigue, emotional function and patient's feeling of control over their disease. It has been shown to be precise, valid and responsive measure of quality of life with good internal consistency (312).
ŀ	The Clinical COPD	The CCQ was developed to aid clinicians to identify the clinical
	Questionnaire (CCQ)	status, activity limitations, emotional dysfunction and clinical improvement in COPD patients. It has 10 items assessing three domains (symptoms, functional wellbeing and emotional wellbeing) by a 7-point Linkert scale. It is a valid, reliable and responsive measure in COPD patients and is useful in identifying 'at risk' patients and assessing improvement in patients after smoking cessation (313).
	The COPD Assessment Test (CAT)	The CAT was developed to help assess health status in COPD patients and enhance communication between clinician and patient. It contains eight items assessing cough, phlegm, chest tightness, breathlessness going uphill or upstairs, activity limitation at home, confidence on leaving the home, sleep and energy (314). It is a sensitive, valid and reliable standardised measure of COPD (315). It has been shown to be useful in measuring pulmonary rehabilitation response (316) and changes in health following acute exacerbations (317).
	The International Union against Tuberculosis Bronchial Symptoms Questionnaire	The International Union against Tuberculosis developed the Bronchial symptoms questionnaire to measure the prevalence and distribution of asthma (318) and has shown good repeatability across Europe particularly in questions that assess asthma and wheeze(319).
	The Leicester Cough	The Leicester Cough Questionnaire was developed to assess the
	Questionnaire	quality of life in adults with Chronic Cough. It is a valid,

	repeatable, 19 item questionnaire assessing QoL over three domains; physical, psychological and social wellbeing (320). It is useful in assessing acute cough (321), non-CF Bronchiectasis (322) and COPD (323).
The Medical Research Council (MRC) Breathlessness Scale	The MRC Breathlessness Scale comprises of five statements describing the disability associated with breathlessness, from no respiratory capacity to almost complete capacity (324). It can be self-completed or completed by an interviewer (324). It has been used for stratification of treatment, survival prediction and disability measurement in COPD Patients (325).
The Quality of Life for Respiratory Illness Questionnaire (QOLRIQ)	The QOLRIQ consists of 55 items assessing seven domains; breathing problems, physical problems, emotions, general activities, situations that triggered or enhance breathing problems, daily domestic activities and social activities/relationships/sexuality (326). It has been shown to be sensitive to change longitudinally and is valid and reliable in measuring QOL in moderate to severe COPD and asthma (327)
The Wisconsin Upper Respiratory Symptom Survey	The Wisconsin Upper Respiratory Symptom Survey is a continuation of work by Jackson et al. in 1960 (328) which created a standardised measure of the symptomatic and functional impact of the common cold (329). The questionnaire contains 44 items assessed on a seven point scale (329) and is valid, responsive, reliable (330) and correlates with biomarkers of respiratory disease (331). A smaller version containing 22 items has been validated (330).

Appendix 6- Study Protocol

Study Protocol

The Liverpool Respiratory Birth Cohort Study

A Prospective, Longitudinal Birth Cohort study using the Liverpool Respiratory Symptom Questionnaire to conduct a biannual assessment of the respiratory symptoms of preschool children born in Liverpool from birth until the age of five years.

> Miss Rosanna Pickles Miss Bethan Griffith Dr Calum Semple Dr Paul McNamara Dr Kevin Southern Professor Ben Shaw

Abstract

Objective: To describe respiratory symptoms of preschool children using the Liverpool Respiratory Symptom Questionnaire (LRSQ) from birth until the age of five years, in Liverpool by bi-annual assessment.

Method: Newborn infants will be recruited during their mother's stay at the Liverpool Women's Hospital (LWH). Prior to discharge, research students will provide information to mothers about the study in the form of a postcard. Mothers will be asked to fill in their contact details including email and phone number indicating their interest in taking part. Completion of the postcard implies consent to be contacted. Postcards may be deposited in collection boxes at the LWH. Once the research team have received the postcards, parents will later be sent an email or letter thanking for their interest in the study. When the baby is four months old an email will be sent to the mother, which will include an option to consent to take part and a link to the initial online questionnaire. Mothers may alternatively opt to receive the questionnaires by post. The questionnaire will then be emailed or posted to mothers six monthly until their child is five years old. Demographic details will be requested on initial enrolment and confirmed or updated during the course of the study. Data will be hosted on-line by Adobe Forms Central software in an encrypted and anonymized format and stored on secure servers at the University of Liverpool.

Analysis: Demographic, exposure and LRSQ data collected by Adobe Forms Central software, which is compatible with the statistical analysis software SPSS[™]. Univariate and multivariate analyses using linear regression analysis will be used to compare domain scores of the LRSQ scores with exposures/variables such as prematurity, birth weight, deprivation and exposure to cigarette smoke in pregnancy and in the household. Structural equation analysis and multinomial regression analysis will also be used to assess any relationship between exposure/demographic variables and respiratory symptoms. Cronbach's alpha coefficients will be calculated to assess internal validity.

Introductions

A literature search of birth cohort studies that explore respiratory symptoms revealed 129 studies conducted between 1961 and 2011 written in the English language. Of these 129 studies 17 separate respiratory birth cohort studies were identified. Among the UK studies identified are the AVON study using the Bristol cohort, the 'Children of the 1950's and the National Child development study. International studies specifically focusing on respiratory symptoms include the Tucson Cohort and the ISAAC study. All use a respiratory questionnaire to assess respiratory symptoms, but none assessed the impact of respiratory symptoms on both preschool children and their parents. The Liverpool Symptom Questionnaire (LRSQ) is specifically designed to assess the impact of respiratory symptoms on preschool children and their parents. This study is the first proposed birth cohort study to use the LRSQ.

Over the last few decades numerous questionnaires have been designed to explore the respiratory symptoms of adults and children. Many studies have since been conducted to examine the validity of these questionnaires. Questionnaires may be delivered by an interviewer, be completed by either the patient themselves or by the parent's of the patient. Self-completion questionnaires have proven to be more economical and also help reduce observer bias when compared to interview questions ¹⁸.

A literature search was conducted using Medline. Keywords searched include 'respiratory' 'symptom' and 'questionnaire'. The results were limited to the dates 1991 to 2011. In total 775 articles were identified and reduced to 69 after reviewing titles and abstracts to determine relevance. The 69 articles related to 36 different respiratory symptom questionnaires for both children and adults and enabled identification of the most commonly used questionnaires.

Questionnaires for respiratory symptoms commonly used in adults include the St George Respiratory Questionnaire (SQRQ), the American Thoracic Society

Standardized Respiratory Questionnaire (ATSq), the Global Allergy and Asthma European Network Questionnaire (GA2LENg) and the MRC respiratory symptom questionnaire. The most commonly used questionnaires in the studies identified are the SQRQ and modified versions of the ATSq. The primary symptoms explored using the questionnaires are cough, wheeze, and breathlessness. Many studies have edited existing questionnaires to include questions on smoking, occupational hazard and various other exposure or occupational respiratory hazards. The most commonly used questionnaire's available specifically for children include the ISAAC questionnaire for ages 6 years to 13 years, and the Test for Respiratory and Asthma Control in Kids (TRACK). Both questionnaires have been validated by numerous studies. Many of the adult questionnaires explore the impact of respiratory symptoms upon the patient. The St Georges Questionnaire for adults addresses the affect of respiratory symptoms upon patient but not upon the family. The Wisconsin Upper Respiratory Symptom Survey (WWRSS 44) includes questions on whether the persons cold has affected their daily activities, work inside and outside their home, interact with others and personal life ¹⁹. The Living with Asthma Questionnaire explores the impact of asthma on the person completing the questionnaire, however this is a questionnaire designed for adults and is specific to Asthma and doesn't necessarily incorporate any other respiratory condition ²⁰.

As discussed many adult questionnaires explore the impact of respiratory symptoms upon the patient and their lives. However, currently no preschool questionnaire for parental completion explores the impact of these respiratory symptoms on the children and their parents. There are also very few respiratory questionnaires validated specifically for the preschool age groups and the LRSQ addresses both these issues.

The Liverpool Respiratory Score Questionnaire (LRSQ) is a validated tool that explores the prevalence of respiratory symptoms in infants and preschool children. This parental completion questionnaire was first designed using established criteria, as a follow up tool for use in neonatal studies where the outcome of children two to three years of age was of interest. A unique feature of the LRSQ is that it also explores the impact of wheeze attacks and other respiratory symptoms upon the child and their family. The LRSQ consists of nine domains. Each domain contains between three and five questions seeking responses scored on a five point Likert scale from "not at all" (score 0) to "every day" (score 4) The first six domains assess respiratory symptoms such as wheeze and cough. The next two assess the effect on the child and their family. The final section asks for details regarding medication, GP/clinic visits, hospital admissions and labels given. All domains ask parents to consider symptoms/effect over the last three months.

Birth Cohort studies are an invaluable tool for studying the epidemiology of specific populations. This study aims to map the natural history of respiratory symptoms of preschool children born in the Liverpool. It is likely to be an invaluable tool to assess the complex relationships between childhood respiratory symptoms and deprivation, premature birth, birth weight, smoking in pregnancy and smoking by household members. Liverpool is recognised as being of the most deprived cities in the England with high rates of cigarette smoking ²¹. This permits for reliable studies into these effects and makes Liverpool an ideal location for a birth cohort study. The majority of births occur at one centre facilitating recruitment.

Work underpinning this study

Three studies have been conducted using the LRSQ. The first, by C V Powell et al (2002) developed and validated the standardised questionnaire ²². After reviewing other questionnaires we find that most do not explore the impact of specific respiratory symptoms, including wheeze, on the child and their family. However, the LRSQ does this well with two of 9 domains exploring the impact of respiratory symptoms on both the child and their family. A relatively small cohort was used for initial exploration of the questionnaire, however the authors justify this as a reasonable number, as other questionnaire used similar figures. After assessing particular areas such as response rates and reliability, the authors demonstrated that it is an acceptable questionnaire, easily completed, with good response rates ²². However they did not attempt to examine the readability in detail or looked at factors, which may affect responses ²².
The second was a cross sectional study that explores respiratory symptoms in Cystic Fibrosis ²³. This study also further validated the questionnaire's external and internal validity. The study showed that the LRSQ has good internal validity across 6/8 domains. It covered an extensive number of symptoms while also maintaining acceptability. It also demonstrated the LRSQ as a potential tool for assessing and monitoring respiratory symptoms in preschool children with cystic fibrosis.

A small, unpublished cross-sectional study has also been conducted using the Liverpool cohort, which used the LRSQ to explore respiratory symptoms in infants following exposure to RSV bronchiolitis. This study again demonstrated good internal validity using the Cronbach's coefficient but also enabled identification on small issues with the design of the questionnaire with the possibility of many improvements that may help with data collection and improve the clarity of the questionnaire.

Justification of this study

A literature search was conducted identifying birth cohort studies that feature a respiratory component. Among the UK birth cohort studies identified were the AVON study a large longitudinal birth cohort, which recruited over 14,000 pregnant mothers from the Bristol Cohort ²⁴. The 'Children of the 1950's' is a large study conducted on the Aberdeen cohort followed children born in the 1950's up until adult life. The British National Child Development Study (BNCDS) started in 1958 and recruited all births within one week in the UK. International studies specifically focusing on respiratory symptoms include the Tuscon Cohort and the International Study of Asthma and Allergy in Children (ISAAC) study. Both of these use a respiratory questionnaire to assess respiratory symptoms, but did not apply it in preschool children.

The LRSQ has demonstrated potential as a tool for assessing respiratory symptoms in preschool children. The validity and utility of the LRSQ have been assessed in two previous studies, demonstrating it to be a useful tool for assessing respiratory symptoms in preschool children however further

validation is required, using a larger cohort of patients. Many large birth cohort studies have been conducted in the UK and Internationally.

This is the first proposed prospective birth cohort study to use the Liverpool Respiratory Symptom Questionnaire. This study will enable a large range of information regarding respiratory symptoms of preschool children and other risk factors predisposing children to respiratory and related disease. It enables numerous future studies to be conducted using the data collected. It allows exploration of respiratory symptoms in relation to demographic details and details of exposure to risk factors which is particularly of interest in Liverpool, as it is considered to be one of the most socially deprived cities in the United Kingdom²¹.

Research Method

Study design

This is a longitudinal birth cohort study using the parent completed Liverpool Respiratory Symptom Questionnaire (LRSQ) to assess preschool children's respiratory symptoms from birth up until the age of five years old. This study will also provide the opportunity for additional cross-sectional studies to be conducted on the patient group recruited and the results gathered. It is important to note that there will be no specific interventions made by the study.

Setting: Children normally resident in Liverpool postcodes L1-38.

Recruitment

We propose a maximum recruitment strategy and aim to recruit as many infants born at the Liverpool Women's Hospital as possible from January 2013. Recruitment will be limited only to parents domiciled within the Liverpool postcodes L1-38.

Newborn infants will be recruited during their mothers stay prior to discharge from the Liverpool Women's Hospital. Research students will personally provide information to mothers about the study verbally and in the form of a postcard while the mothers are at the Liverpool Women's Hospital. Mothers will be asked to fill in their contact details including email and phone number indicating their interest in taking part. Completion of the postcard implies consent to be contacted. Mother's hospital stickers will also be attached to the postcard. In addition, the postcard will also include a QR(2D) bar code which mothers may scan using their smart phones. This QR code will direct participants immediately to an online version sign up version of the postcard.

Once completed, research students will collect the postcards. Alternatively mothers, midwives or volunteers at the Liverpool Women's Hospital may deposit the postcards in a collection box. Once the research team have received the postcards contact details will be uploaded to the database sorted on the University of Liverpool's secure server. Within a week, parents will be sent an email thanking for their interest in the study. The next point of contact will be when the baby is four months old. An email will be sent to the mother, which will include a link to the initial online questionnaire. This email will include; information about the study, consent, questions regarding demographic and exposure details and the LRSQ in a series of separate but easily understandable pages. Mothers may alternatively opt to receive the postal questionnaires, which will be first sent within a week of the baby being four months old.

Recruitment material will make no therapeutic promises, as this is only a descriptive study. The inclusion and exclusion criterion ensures that no one is unfairly excluded from the study. All data collected will be automatically deposited on a password database on a secure server in the University of Liverpool. Where parents have indicated that they would prefer hard copy (paper) correspondence, this will be sent with a stamped address envelope for return. The research student will enter this data by hand. Data will be imported into an SPSS database on a case-by-case basis.

Follow Up

Consent will be obtained when the initial questionnaire is sent, four months after birth. Following that, a second email will be sent six months later and every six months thereafter until the child is five years old. This repeat mailing questionnaire will provide an option for participants to update any contact details and will include demographic and exposure questions and the LRSQ questions.

In addition, on the date of their child's birthday, mothers will receive a personalised email thanking them for taking part in the study and wishing their child a happy birthday. In the case of no reply, two reminder emails will be sent and if no response this will be followed by one contact by telephone and one postal contact as email addresses and telephone numbers may have changed over time. Each contact attempt will be two weeks after previous attempts. For mothers without email addresses or access to computers, reminders will be made by telephone and mailed by post.

With each email sent, mothers will be given an option to sign up to updates regarding recruitment and reports on any preliminarily findings. Families who move out of the area will be asked to continue their involvement. Contact will continue by email and telephone. The questionnaire will include an option to update place of residence among other details.

Data Collection

Data will be linked from the on-line Adobe Forms Central Survey software into a SPSS database on a case-by-case basis throughout the duration of the study. The data collected from postal questionnaires will be inputted into the online survey manually by and subsequently inputted into the database by the research student.

Inclusion criteria

All infants born at the Liverpool Women's hospital, including premature births, where parents are normally resident in Liverpool postcodes (L1-38) regardless of future residence.

Exclusion criteria

Neonates born to parents normally resident outside the Liverpool postcodes L1-38.

Babies born to non-English speaking parents.

Variables and Outcomes Measures

Aim

To establish a population based longitudinal birth cohort study conducting a bi-annual assessment of respiratory symptoms of preschool children using the LRSQ from birth to the age of five in Liverpool.

Primary objective: To describe parent reported respiratory symptoms in a population based birth cohort followed longitudinally form birth to five years old using the LRSQ.

Secondary objectives: To examine any association between differences in respiratory symptoms in groups of preschool children with different social and environmental risk and protective factors.

Bias

To minimise withdrawals from the study patients will be given three reminders, two via email and one by telephone contact after initial email of the questionnaire.

Recall bias is not considered to be a problem, as the questionnaire requires parents to report respiratory symptoms in the last three months and this recall period has been validated for this questionnaire In the three previous studies parents have not fully completed the questionnaire, which may introduce reporting bias. Using the online questionnaire may help, as software prompts parents to complete all questions and can give options for null responses rather than leaving 'blanks' on paper.

Demographics and Exposure Variables

Proposed Study Size

We plan a maximum recruitment strategy from the Liverpool Women's Hospital where there are approximately 8,000 births each year. We estimate that approximately one in four mothers will complete the postcard provided. The questionnaire currently has a 13% return rate therefore we expect approximately 260 patients to participate in the study each year. Comparison of demographic data for those participating with census data will allow a check for recruitment bias.

Statistical Methods

The results from questionnaires on Adobe Forms Central will be uploaded to SPSS and analysed using the SPSS Statistical software. Rolling cumulative data analysis will be performed for the duration of five years.

Univariate and multivariate analyses using linear regression analysis will be used to compare domain scores of the LRSQ scores with exposures such as maternal smoking etc. Fishers exact test will be used to determine whether there is any relation between two categorical variables. Structural equation analysis and multinomial regression analysis will also be used to assess any relationship between exposure/demographic variables and respiratory symptoms. Cronbach's alpha coefficients will be calculated to re-assess internal validity. In this study missing data is most likely to result from failure to respond.

Data Sources: Patient Demographic Service

Data will also be collected from linked hospital episode data.

Data management

Data will be collected from questionnaire answers via the Adobe Forms Central Software and linked directly to the programme SPSS for analysis.

Consent

Mothers providing their contact details on the study information postcard implies consent to be contacted. The consent of mothers of patients will be sought when the first questionnaire is emailed or posted to participants. Mothers who are willing to participate will click on the embedded link to the questionnaire will be directed to a screen detailing more information about the study. After reading this they will be given the option to either not to participate and unsubscribe to emails, to contact the research team with any queries or to consent by clicking a button online and inserting their initials. After consenting to take part, mothers will be directed to the questionnaire. Mothers who opted for the postal questionnaire will be asked to complete a consent form alongside the initial questionnaire.

Patients and parents of patients at participating practices have the opportunity to opt out of the scheme at registration and any point thereafter by either contacting the research team or clicking a button on the email sent containing the LRSQ. [In studies involving postal questionnaires where the burdens are

insignificant and sensitive topics are not involved, the REC will normal regard the return of the questionnaire as adequate evidence of consent (IRAS guidance 2012)].

Ethical Issues

This study received REC ethical approval by proportionate review on 08/05/2012. REC Reference: 12/EM/0194

This study also received REC approval of minor amendments on the 08/11/2012.

This allowed use of the QR barcode on the postcard for recruitment ant the use of a poster with the QR code for recruitment and advertisement of the study at the Liverpool Women's Hospital.

No physical intervention will take place. The LRSQ database will include a study ID but not direct identifiers. The study ID and contact details will be kept in an encrypted data file in a secure server hosted by the University of Liverpool. The survey has only been validated in an English Language format. There is no capacity in this student project to develop and validate the LRSQ in other languages.

Risks to Patient

There are few anticipated risks for research participants in this study. Possible risks include a breach in confidentiality with regards to contact details of patients and any personal data. Precautions will be taken to minimize the risk of this data will be stored very carefully. No identifiable data will be included in publications. Security data measures that will be taken include encrypting data with passwords, coding medical conditions and limiting access to study data.

An additional foreseeable risk identified is the risk of sending emails and questionnaires to parents of deceased children. Measures that could be taken include linking to the National mortality database via the Patient Demographic Service. The Alder Hey Children's NHS Foundation Trust's IT department will perform weekly batch searches linking to SPINE via the Patient Demographic Service. This, however, is not a foolproof method as there will be delays between the PDS being notified by the community and by Alder Hey Children's Hospital.

Benefits to patients

There are no direct benefits to research participants taking part in this study. However there we hope that this study will benefit future preschool children by enabling us to identify risk factors associated with particular respiratory symptoms.

Risks / benefits to study

If too few participants are recruited this would compromise the results of the study. There is also the risk of loosing patients to long-term follow-up. If patients are lost the data will still be included in the results.

Research governance

The University of Liverpool will be lead sponsor and the Liverpool Women's NHS Foundation Trust will co-sponsor for the study. The proposed study will be undertaken in accordance with the University of Liverpool's research governance procedures.

Dr MG Semple (Liverpool University), Professor Ben Shaw (Liverpool Women's) will be a joint guarantor for analysis and reports.

The Research Team

Miss Rosanna Pickles Miss Bethan Griffiths Dr MG Semple (Chief Investigator) Professor Ben Shaw Dr Kevin Southern Dr Paul McNamara



Timetable and Milestones

Phase One – completion of protocol and application to IRAS for ethical approval by May 2012

Phase Two – Development of the online questionnaire and consent form.
Prior to the start of the study mothers will be interviewed regarding
questionnaire aesthetics. Once the questionnaire design has been finalised
the questionnaire will then be trialed at the Liverpool Women's hospital.
Mothers will be interviewed while they are completing the questionnaire about
the design, layout, format and content of questions; ease of used and asked
to score the questionnaire out of ten. Mothers will also be asked a about
anything that would motivation them to participate in the study, and encourage
continued participation.

Phase Three – Recruitment of mothers and implementation of the study will begin on the 7th January 2013 and continue for a maximum of five years depending on the success of the study. The study will continue for a further five years after recruitment is complete.

Expertise

- Dr MG (Calum) Semple is a Senior Lecturer in Child Health at the University of Liverpool and Consultant in Paediatric Respiratory Medicine at Alder Hey Children's Hospital.
- Professor B Shaw, Consultant in Neonatal and Respiratory paediatrics at Liverpool Women's Hospital and the Royal Liverpool Children's Hospital.
- Dr Paul McNamara, Senior Lecturer in Child Health at the University of Liverpool and Consultant in Paediatric Respiratory Medicine at Alder Hey Children's Hospital
- Dr K Southern Reader in Paediatric Respiratory Medicine at the University of Liverpool and Consultant in Paediatric Respiratory Medicine at Alder Hey Children's Hospita

Service Use Input

Mothers of the participants will be involved in the development of the study, particularly with the development of the postcard information card and also the design, content and format of the online questionnaire. Research students will interview mothers at the Liverpool Women's Hospital about the aesthetics and format of the questionnaire and postcard. Finalised versions of the questionnaire will be piloted at the Liverpool Women's Hospital. Mothers will be interviewed while completing the online questionnaire. Interview questions will be standardised and prepared prior to seeing the parent. They will be asked to feedback on matters such as appearance and format, ease of completion, and clarity and content of the information and questions.

References

1. Attfield MD. Respiratory Questionnaires.

2. B. Barrett REB, M. P. Mundt, G. R. Thomas, S. K. Barlow, A. D. Highstrom, and M. Bahrainian. Validation of a short form Wisconsin Upper Respiratory Symptom Survey (WURSS-21). Health and Quality of Life Outcomes. 2009;7.

3. Living with Asthma Questionnaire. Available from:

http://www.psy.plymouth.ac.uk/research/mhyland/Scoring%20%20the%20Lliving%20with%20Asthma.pdf.

4. The English Indices of Deprivation 2010.

 Powell CVE, McNamara P, Solis A, Shaw NJ. A parent completed questionnaire to describe the patterns of wheezing and other respiratory symptoms in infants and preschool children. Archives of disease in childhood.
 2002 November 1, 2002;87(5):376-9.

6. R Trinick KS, P Mcnamara. Assessing the Liverpool Resiratory Symptom Questionnaire in Children with Cystic Fibrosis. 2011.

7. Children ALSoPa. Avon Longitudinal Study of Parents and Children. University of Bristol; 2012; Available from: http://www.bristol.ac.uk/alspac/. Appendix 7- Results of PPI Sessions for Questionnaire and Study Development Performed by Previous MPhil Students

Rossana Pickles and Bethan Griffith developed the questionnaires through the use of PPI sessions based at the LWH. Eligibility criteria for the PPI sessions were;

- Inpatients at the LWH
- Home address in postcodes L1-L38,
- Spoke sufficient English to participate.

The properties of the questionnaire that were discussed are discussed below. The prototype questionnaires used in the PPI sessions are shown in the appendix.

Mode of Completion

65% (n=13) mothers said they would complete the online questionnaire on a laptop or personal computer. The remaining mothers said they would complete the questionnaire using another online device. No mothers in the PPI sessions were interested in receiving the questionnaire by post. Mothers made comments on the questionnaire being available in more than one format.

Question Density

Of the three questionnaire density options (see appendix) the majority of participants (n=13) opted for the 'medium density' questionnaire. There was an even split between the high density and low density questionnaires (n=3) (2, 3).

Background

35% (n=7) participants had a preference for a plain background and non-coloured text, 30% (n=6) opted for a plain background with subtly coloured text, 25% preferred a subtly coloured background. 5% (n=1) preferred lots of colour and 5% (n=1) had no preference (2, 3).

Themes, Images and Logos

Of the three theme options (see appendix) 65% (n=13) opted for the theme which was a strip of children holding hands. 20% (n=4) of the participants would prefer images to be kept to a minimum. Mothers said that the images should only be used if relevant to the study and overall the questionnaire should look professional (2, 3).

Font Size

'Arial' was chosen due to its wide availably in software packages. 75% (n=15) of the participants opted for the medium sized font (size 16) (2, 3).

Other Comments

Participants said that it was vital to ensure there was enough information about the study available. The questionnaire should be kept 'short and sweet' and 'response options kept minimal'. Small incentives would make participants more likely to complete the questionnaire (3).

Questionnaire Content

The cohort demographics and infant exposures that would be assessed in the LBBS were decided after discussions with respiratory experts Dr Calum Semple (Alder Hey Children's Hospital) and Professor Ben Shaw (LWH). These are shown in figure 1. Each questionnaire included sections for demographics, risk factors, the LRSQ and health care service attendance (2).

The PPI sessions have let the research team design a questionnaire that is tailored to both the research team and the participants needs.

Infant gender, Postcode/Index of Multiple Deprivations Ethnicity, Gestational age, Birth weight, Mother's highest qualification, Breastfeeding (duration), Nursery attendance, Sharing a bedroom, Number of household siblings, Maternal smoking during pregnancy, ETS exposure (before and after birth), Co-Morbidities FH of atopy

Figure 11: Demographics and Exposures assessed in the LBBS

An example of the final questionnaire is shown in the appendix. It was available for completion online and in a paper version sent via post. There is a medium question density on each page of the questionnaire. The theme used was a strip of children holding hands, with a plain background with subtly coloured text. Images used are only relevant to the study and are proprietary artwork of the study team. A medium font size was used.

With the exception of questionnaire 1, no patient identifiers were stored on the questionnaire platform severs. Questionnaires were linked to participants using a unique identification code which was embedded in emails sent to the participants. This minimised the patient information stored outside the University of Liverpool computer system. This

unique identification code was required to complete the questionnaires and minimised the personal information that was stored on the online questionnaire platform. All exports from the questionnaire platforms were stored on encrypted, password protected files on the University of Liverpool server.

Recruitment Material

The study name was changed from 'The Liverpool Respiratory Birth Cohort' to the "Liverpool Baby Breathing Study" (LBBS) as the research team decided it would be more attractive and understandable to parents. Posters advertising the study, containing a digital quick response code (figure 2), were put in the LWH. Participants who said they would like to join the study during recruitment were given a postcard to complete with their contact Response code used for details, which was to be posted into a post-box by the entrance to



Figure 12: Quick sign up to the LBBS

the wards. The posters and postcards had the same theme as the questionnaire and emails to give the study a professional and recognisable look. The poster and postcard design are shown in the appendix.

Appendix 8- Recruitment Pilot Study Results

A fourweek pilot study for the recruitment of infants into the LBBS was performed by Rosanna Pickles and Bethan Griffith to determine the most optimal yet feasible recruitment strategy. Four different recruitment strategies were each trialled for a one week period. Recruitment strategies and their recruitment rates are shown in table 6.

Strategy	Days	Recruitment Rate
5 Full Days	Monday- Friday	58%
5 Mornings	Monday- Friday	65%
4 Alternate Days	Monday, Wednesday, Friday, Sunday	61%
7 Full Days	Monday- Sunday	78%

 Table 32: Recruitment rates during recruitment pilot study

Recruitment took place on the post-natal wards and the neonatal unit at LWH. Researchers aimed to recruit all the eligible births when on the wards. Children diagnosed with CF were also invited to join the study during their initial consultation with a respiratory consultant.

The pilot study recruited 334 interested mothers from 605 births in the LWH; 332 from postnatal wards and two from CF clinics. 67 mothers were not from eligible postcodes, 119 births were missed due to the research team not being present on the wards, 36 were not approached, 24 were non-English speaking, four were not suitable for recruitment as identified by the clinical team and 21 mothers declined (2, 3). Appendix 9- Development of the Online Questionnaire Distribution Method

Questionnaire Platform

Online survey software's were appraised for the features and services they offer. A shortlist of four software packages was produced by the research team; 1.SelectSurvey.NET (SSN) (ClassApps LLC, Kansas City, Missouri, USA) 2.Google Docs (Google Inc. California, USA), 3.Adobe FormsCentral[®] (AFC[®]) (Adobe Systems Inc. California, USA) .and 4.SurveyMonkey[®] (SurveyMonkey Inc. California, USA) (2, 3). The comparison of the software's is shown in the appendix.

The research team decided to opt for AFC[®] for the questionnaire platform; it provided a better software for the both research team and the participants, allowing the creating of a questionnaire that was outlined acceptable by the PPI sessions. AFC[®] was appraised against SSN in a feasibility study. AFC[®] scored better for first impressions and after questionnaire completion. 10 out for 16 mothers preferred the AFC[®] questionnaire overall (2, 3). In June 2015 AFC[®] was suspended to allow development as a future premium product with charges. This forced the study team to change to use JotForm[®] (JotForm Inc. San Francisco, USA) for questionnaire completion and response storage. It allowed the research team to reproduce the questionnaires from AFC[®], whilst giving the same benefits and security as AFC[®] with enhanced form analysis which would help the research team resolve any issues with submissions. The properties of both AFC[®] and JotForm[®] are shown in the table7. The new questionnaires and the fully online recruitment linked to the automated questionnaire delivery system was developed by Sanjay Patel.

Property	AFC®	JotForm®	
Cost	£105.39/year	\$90/year (~£70/year)	
Usability	Excellent	Excellent	
Skip Logic	Question and Page	Question and Page	
Help Text	Yes	Yes	
Logos and Themes	Unlimited	Unlimited	
Field Validation	Yes	Yes	
Security	SSL Encryption Security	SSL Encryption Security on Secure German Server	

Table 33: Properties of AFC [®] and JotForm[®]

Data Export	SPSS & Excel Data	SPSS & Excel Data Export
	Export	
Analysis tools	Data Summaries	Form Analysis
Support	Telephone and Email Support	Email support and User Forums

Email Platform

Factors that were considered when choosing a mailing platform for the distribution of online questionnaires were cost, deliverability, functionality, usability, security and the ability to personalise emails. The research team decided that MailChimp® (The Rocket Science Group, Atlanta, USA) would be the best platform for the distribution of emails to the cohort. MailChimp® provided a secure online server with two-tiered security and a fully automated system which precisely scheduled emails based on the participant's date of birth with minimal user input. MailChimp® tracked if emails are opened and if links within emails have been clicked; this is a benefit of the study as it will allow any issues or excessive drop out to be dealt with in sufficient time. It allows users to remove themselves from mailing lists which was important for autonomy. Participants could self-enrol into the study using a quick response code on the study advertising posters without the need of the research team. Emails could be branded with the same theme of the questionnaires, allowing all study materials to be identical.

Appendix 10- Prototype Questionnaires



Appendix 11- Consent form and Questionnaire one used in the LBBS





The Liverpool Baby Breathing Study

Dear Parent/Guardian,

Many congratulations on the birth of your new baby!

During your stay at Liverpool Women's Hospital, you spoke to a member of our research team and agreed to be contacted for our study. We understand that this has been a very busy time for you, so we really appreciate that you took the time to speak to us.

As you may recall, The Liverpool Baby Breathing study is a study that explores breathing symtpoms, such as coughs, colds and wheezing in babies and young children. The aim of our study is to understand how these symptoms change over time, and how they affect you an your family. You can read a little more about the study on the first page of the enclosed questionnaire, under the "Further Information" heading.

If you are still happy to take part in the Liverpool Baby Breathing Study, please let us know by completing the attached consent form, and the first questionnaire (it only takes 5-10 minutes of your time!). We have enclosed a stamped envelope so you can return the questionnaire to us at your earliest convenience.

If you provided an email address when you registered your interest in the study, you may have received a link to an online version of the questionnaire. Don't worry, you don't have to fill it in twice – simply complete the online *or* paper version!

If you have any questions or concerns about the study, or are having problems completing the questionnaire, or would no longer like to participate in the study, please contact the research team directly and we will be happy to help!

We look forward to hearing your responses!

Best Wishes

The Research Team

Miss Rosanna Pickles Miss Bethan Griffith Dr Kevin Southern Professor Ben Shaw Dr Calum Semple

Email: babystudy@liverpool.ac.uk Telephone: 01512824532 Post MG Semple, Institute of Child Health, University of Liverpool, Alder Hey Children's Hospital NHS, Eaton Road, Liverpool, L12 2AP





Consent Form



Title of Project: Liverpool Baby Breathing Study (The Liverpool Respiratory Birth Cohort Study)

Name of Researchers:

Miss Rosanna Pickles, Miss Bethan Griffith, Dr Calum Semple, Dr Kevin Southern and Professor Ben Shaw

Please tick box

- I confirm that I have read and understand the further information section dated 01/05/2013 for the above study. I have had the opportunity to consider the information, ask questions and have had these answered satisfactorily.
- I understand that our family and child's participation in the study is entirely voluntary and that we have the right to withdraw at any time without stating a reason and without affecting my care and my family's care in any way.
- I understand that the data collected by this study will be looked at by members of the research team named above and may be scrutinised by regulatory authorities or by the host NHS Trusts.
- 4. I have read and understand the above consent form, I certify that I am the parent/guardian of the child recruited.
- 5. I give permission for our family and my child to be involved in the above study.

Child's Forename (first/given name):	Surname:
Your Child's Date of birth: Day: Month:	Year :
Mother's Forename (first/given name):	_ Surname:
Mothers Signature for Consent:	
Today's Date: Day: Month: Year :	







Welcome, and thank you for your interest in... Liverpool Baby Breathing Study

Questionnaire 1

The following questionnaire asks questions about your baby and what has been happening to him or her over the last three months. It should take **no longer than 10 minutes to complete.** All future questionnaires should take no more than 5 minutes to complete.

This study aims to find out more about the respiratory symptoms, such as wheezing, coughs and colds, that your child experiences, and how they affect you and your family.

Our study will help us understand what makes these symptoms more or less likely to occur and how they change over time.

It is important that every question is answered, even if your child has been perfectly well, with no problems at all.

Thank you.

Further Information 01.05.2013

Why have I been chosen?

We are asking all parents of children born at the Liverpool Women's Hospital who were living within the L1-38 post codes when their child was born to take part. We are particularly interested in the children of Liverpool as there are high levels of respiratory diseases such as asthma and bronchiolitis.

Do I have to take part?

No - it is up to you whether you decide to take part or not. You are free to withdraw at any time, without giving a reason. You and your child's future clinical care will not be affected if you do not wish to take part.

What will it involve (before/during/after?)

-We ask you to complete a questionnaire about your child. We will email you a link to the questionnaire just twice a year for five years.

-This questionnaire should take no longer than 10 minutes to complete.

-We will be asking you a few questions about your family circumstances, and your child's respiratory symptoms. -You can choose to receive updates on how the study is running by email. At the end of the study, we will send you a summary of the results for the whole study.

Are there disadvantages of taking part?

We are not aware of any disadvantages to you or your child. All information will be treated with the strictest confidence.

Are there any advantages for taking part?

Being involved in the study will not benefit your child directly. We hope to help other children in the future by identifying what helps or worsens respiratory symptoms.

If you you have any further questions, please contact the research team directly:

Email: BabyStudy@liverpool.ac.uk Telephone: (0151) 282 4532

Please let us know if you would not like to receive any further contact from us, or if you want to go green and start receiving your questionnaires by email!







About Your Pregnancy, Birth and New Baby You will only need to give us these details once

1. What is your <u>new baby</u> 's first name?
2. What is your <u>new baby</u> 's last name?
3. Is your new baby:
O Male O Female
4. What is your new baby's date of birth?
5. What best describes your new baby's ethnic group or background?
O I would prefer not to say
O White - British
O White - Any other white background
O Mixed or multiple ethnic groups
O African, Caribbean or any other black ethnic group
O Asian - Indian, Pakistani or Bangladeshi or any other ethnic group
O Eastern Asian- Chinese or any other ethnic group
O Other - Please Specify
6. How many weeks pregnant were you when you gave birth? e.g. if you gave birth at 36 weeks 5 days , please write "36"
7. How much did your baby weigh at birth?
8. Did vou smoke at <i>anv time</i> during your pregnancy?
O Yes O No
9. Did any member of your household smoke, <i>anywhere</i> or at <i>any time</i> during your pregnancy?
A household member means someone who sleeps or regularly takes meals at your house.
O Yes O No
Where did these people smoke? <i>Tick all that apply</i>
Inside the home
Outside the home
Inside the car
Inside at another location eg. work/social
Cther
10. Did you breast feed your new baby at any time? This includes expressed breast milk or bottle feeding at the same time
O Yes O No
If yes, how long did you breast feed for?
O Less than 1 month O 1-3 months O I am still breast feeding







About You and Your Household

Your information will be stored securely and will not be shared

11. Which of these qualifications do you have?

O No qualifications

O Completed secondary school

O GCSE/O-Level

O Vocational Training/ Apprenticeship

O Diploma eg. BTEC, GNVQ

O A Levels/Scottish Highers

O Undergraduate Degree

O Postgraduate Degree

O Other (please specify)

12. Have you, your baby's father, or any of your baby's brothers or sisters ever been told by a doctor that they, or you, have of asthma, hay fever or eczema?

O Yes O No

If you answered yes to the question above, who has Asthma, Hay Fever or Eczema?

	Asthma	Hay Fever	Eczema
Me (My baby's mother)	0	0	0
My baby's father	0	0	0
My eldest child	0	0	0
My second eldest child	0	0	0
My third eldest child	0	0	0
My fourth eldest child	0	0	0

13. Does any member in your household smoke, anywhere?

A household member means someone who sleeps or regularly takes meals at your house.Please tick yes, even if they smoke outside

O Yes O No

If yes, where do these people smoke? *Tick all that apply*

Inside the home

Outside the home

Inside the car

Inside at another location eg. work/social

□ Other

14. Does your baby attend nursery/crèche?

O Yes O No







About Your New Baby

15. Do you	u have any oth	ner children who live w	vith you?			
O Yes	O No					
If yes, how	v many other o	children live with you?				
1	2	П 3	4	5	6	7 or more
16. Does	your baby sha	re a bedroom with you	urself or anybody e	lse?		
O Yes	O No					
If yes, who Please tic	o does your ba k all that apply	aby share a bedroom ⁄	with?			
 Parent Other 	or parents	One other chi	ld 🔲 Two	o or more childro	en	
17. Has yo	our new baby	ever seen your GP be	cause of his/her c	nest?		
O Yes	O No					
If yes, how	v many times	has your baby seen yo	our GP because of	his/her chest?		
Once More th	nan 5 times	Twice	Three tim	es 🗖	Four times	Five times
18. Has yo	our new baby	ever been to hospital	because of his/her	chest?		
O Yes	O No					
If yes, how	v many times	has your baby been to	hospital because	of his/her chest	1?	
O Once O More th	nan 5 times	O Twice	O Three tim	les O	Four times	O Five times
19. Does y	your new baby	/ have any long term h	nealth conditions?			
O Yes	O No					
What kind	of long term h	nealth condition does	your new baby hav	re?		
Long te	erm chest (res	piratory) disease?				
Long te	Long term heart disease?					
	Long term kidney disease?					
Diabete						
Other						
Please tel	l us more abo	ut the long term health	n condition(s) your	new baby has:		







During the DAY(when awake) in the last three months::

20. My new baby I	nas been wheezing	(whistling noise co	oming from the che	st)		
O Not at all	O A few days	O Some days	O Most days	O Every day		
21. My new baby I	nas had a cough					
O Not at all	O A few days	O Some days	O Most days	O Every day		
22. My new baby l baby's chest)	nas had a rattly che	est (<i>noise that you</i> o	can hear and feel a	s a vibration, when placing your hands over your		
O Not at all	O A few days	O Some days	O Most days	O Every day		
23. My new baby has been short of breath						
O Not at all	O A few days	O Some days	O Most days	O Every day		

During the NIGHT (when asleep) in the last three months:

24. My new baby	has been wheezing	(whistling noise com	ing from the chest)	
O Not at all	O A few nights	O Some nights	O Most nights	O Every night
25. My new baby	has had a cough			
O Not at all	O A few nights	O Some nights	O Most nights	O Every night
26. My new baby child's chest)	has had a rattly ches	st (noise that you cai	n hear and feel as a	vibration when placing your hands over your
O Not at all	O A few nights	O Some nights	O Most nights	O Every night
27. My new baby	has been short of bro	eath		
O Not at all	O A few nights	O Most nights	O Some nights	O Every night
28. My new baby	has snored:			
O Not at all	O A few nights	O Most nights	O Some nights	O Every night







29. How many colds (re	unny nose, and high temp	<i>perature)</i> has your new ba	by had in the last	three months?	
O 0	1 O 2	O 3	O 4	O More than 4	
When my new ba	iby has HAD A CO	LD in the <u>last three</u>	months:		
30. My new baby has b	een wheezing (whistling	noise coming from the ch	est)		
O Not at all with colds O Every day	O A few days	O Some days	O Most d	lays	
31. My new baby has h	ad a cough				
O Not at all with colds O Every day	O A few days	O Some days	O Most d	lays	
32. My new baby has h child's chest)	ad a rattly chest (noise th	hat you can hear and feel	as a vibration whe	en placing your hands over y	′our
O Not at all with colds O Every day	O A few days	O Some days	O Most d	lays	
33. My new baby has b	een short of breath				
O Not at all with colds O Every day	O A few days	O Some days	O Most d	lays	

When my new baby has NOT HAD A COLD in the last three months:

34. My new baby has been wheezing (whistling noise coming from the chest)						
O Not at all	O A few days	O Some days	O Most days	O Every day		
35. My new baby	has had a cough					
O Not at all	O A few days	O Some days	O Most days	O Every day		
36. My new baby child's chest)	has had a rattly ch	est (noise that you	can hear and feel a	as a vibration when placing your hands over your		
O Not at all	O A few days	O Some days	O Most days	O Every day		
37. My new baby has been short of breath						
O Not at all	O A few days	O Some days	O Most days	O Every day		







When my new baby has been more active (e.g. crawling, walking or when excited) in the <u>last</u> three months:

38. My new bal	by has been wheezir	ng (whistling noise	coming from the cl	hest)
O Not at all	O A few days	O Some days	O Most days	O Every day
39. My new bal	by has had a cough			
O Not at all	O A few days	O Some days	O Most days	O Every day
40. My new bal child's chest)	by has had a rattly c	hest <i>(noise that you</i>	ı can hear and fee	l as a vibration when placing your hands over you
O Not at all	O A few days	O Some days	O Most days	O Every day
41. My new bal	by has been short of	breath		
O Not at all	O A few days	O Some days	O Most days	O Every day

Other problems my new baby may have had in the last three months:

42. My new baby	has had noisy bre	athing that does no	ot seem to come fro	om the chest:
O Not at all	O A few days	O Some days	O Most days	O Every day
43. My new baby	has had fast breat	thing:		
O Not at all	O A few days	O Some days	O Most days	O Every day
44. My new baby	has had noisy bre	athing that appears	s to come from the	throat or back of the throat:
O Not at all	O A few days	O Some days	O Most days	O Every day







How my new baby's chest symptoms actually affected him or her over the last three months:

45. My new bab	y's chest symptoms	have affected his o	or her feeding or ea	ating
O Not at all	O A few days	O Some days	O Most days	O Every day
46. My new bab	y's chest symptoms	have woken up my	y new baby:	
O Not at all	O A few nights	O Some nigh	ts O Most nigh	ots O Every night
47. My new bab	y's chest symptoms	have reduced my	new baby's activity	c.
O Not at all	O A few days	O Some days	O Most days	O Every day
48. My new bab	y's chest symptoms	have made my ne	w baby unusually t	ired:
O Not at all	O A few days	O Some days	O Most days	O Every day

How my new baby's chest symptoms have affected me and my family in the last three months:

49. My new baby's	s chest symptoms h	nave limited MY act	ivities:		
O Not at all	O A few days	O Some days	O Most days	O Every day	
50. My new baby's	s symptoms have re	esulted in adjustme	nts being made to	our family life:	
O Not at all	O A few days	O Some days	O Most days	O Every day	
51. My new baby's chest symptoms have disturbed our sleep:					
O Not at all	O A few nights	O Some nights	O Most nights	O Every night	
52. I have been worried about my new baby's chest symptoms:					
O Not at all	O A few days	O Some days	O Most days	O Every day	







Your Details

Your data will be stored securely and will not be shared

You will only need to give us these details once!

53. What is **your** first name?

54. What is your last name?

55. What is your new baby's father's first name? (optional)

56. What is your new baby's father's last name? (optional)

57. What is your full address

58. What is your postcode?

ſ

Γ

59. What is your main telephone number?

60. What is your **main** e-mail address?

I would like to receive my future questionnaires online

I want to sign up to the Liverpool Baby Study newsletter to receive updates from the study







The Liverpool Respiratory Birth Cohort Study Thank You For Your Participation!

All the information you give us is invaluable to the research team and future children in Liverpool, even if your baby hasn't had any symptoms at all! We will be in touch with you again in around 6 months time.

If you have any questions or would like to discuss the study with the research team, please don't hesitate to contact the research team at babystudy@liverpool.ac.uk or call us on 0151 282 4532

MG Semple Institute of Child Health, University of Liverpool, Alder Hey Children's Hospital NHS FT, Eaton Road, Liverpool, Merseyside L12 2AP



Appendix 12- Poster used in the LBBS



Appendix 13- Postcard Used in the LBBS

Thank you for your	r interest in the	
LIVERPOOL Liverpool Baby B	Breathing Study	Liverpool Women's
The Liverpool Baby Breathing Study will study the re breathing problems) of children born in Liverpool from v	espiratory symptoms (colds, col when they are born until they a	ughs, wheezing and re five years old.
Why are we doing this? We want to find out more about the respiratory sympton experiences and their effect on your family. The aim of symptoms change over time and what makes them mo	ms, such as wheezing and cour our study is to try to understand re or less likely to occur.	ghing that your child d how these
Why have I been chosen? We are asking all parents of children born at the Liverp L1-L38 postcodes only, if they would like to take part. Liverpool as there are high levels of respiratory disease	ool Women's Hospital who are . We are particularly interested i es such as asthma and bronchi	living within the in the children of olitis.
If you agree to take part:		
• We will send you a questionnaire online or by post tw	rice a year, for five years	
The first questionnaire should take less than 10 minutes Fach following questionnaire about take less than 5	tes to complete	
Your decision to participate will not affect you or your	child's future care in any way	
 All responses and personal details will be handled in 	the strictest confidence	
 You may choose to leave the study at any time 		
Thank you for you LIVERPOOL Liverpool Baby I	r interest in the Breathing Study	
The Liverpool Respirate To register your interest in the Liverpool Baby Breathing St member of staff, or pop it in the collection box. Alternatively, y	ory Birth Cohort Study tudy simply fill in your details below you can enter your details online b	v, and give this card to a y scanning the QR code.
Baby's First Name:	Last Name:	
Baby's Date of Birth:	Singleton: 🔲 Twi	n: 🔲 Triplet: 🗌
Mother's First Name:	Last Name:	
Email:	Postcode: L	
Main telephone number:	Please contact me b	y: email 🔲 post 🗖
By providing these details I agree to be contacted b Don't worry, you are not committing yourself to join the We will contact you in four months time to confirm that If you would like any further information, you can con- team directly by emailing BabyStudy@liv.ac.uk or con- lease scan this code sing your OR reader	by the Research team. study at this stage! you are still interested! ontact the research Affin calling (0151) 2824532.	< Hospital Label Here

	SelectSurvey.NET	Google Docs	Adobe® Forms Central	SurveyMonkey®
Price (annually)	University funded	Free	£105.39	£299
Usability	Fair	Good	Excellent	Good
Appearance formatting	Colour	Templates	Unlimited	Templates
Skip Logic - question	No	No	Yes	Yes
Skip Logic – page	Yes	No	Yes	Yes
Question Piping	Yes	No	No	Yes
Answer Pre-Population	No	No	No	Some
Help text	No	Yes	Yes	No
Logos and images	1 (top of page)	No	Unlimited	1 (top of page)
Field Validation	Yes	No	Yes	Yes
Integrated email	Limited	No	No	Limited
Security	Encrypted	Data cloud	SSL encryption	SSL Encryption
Statistical analysis	SPSS export	Excel export	SPSS export	SPSS export
Data Export	Yes	Yes	Yes	Yes
Data Summary	Yes	Yes	Yes	Yes
Progress Bar	Yes	No	Yes	Yes
Telephone/Email Support	Via University	No	Both	Email

Appendix 14- Comparison of Questionnaire Online Platforms

Appendix 15- Data Coding in the LBBS

		1= Questionnaire 1 (4 Months)	
Questionnaire number	Questionnaire	2= Questionnaire 2 (10 Months)	
Questionnaire number	Questionnaire	3= Questionnaire 3 (16 Months)	
		4= Questionnaire 4 (22 Months)	
Pablas Condar	RGENDER	0=Male	
Bables Gerider	BGENDER	1= Female	
Cingle or Multiple Programmy		1=Singleton	
Single of Multiple Pregnancy	WIOLTIP	2=Multiple Birth	
		0= I Would prefer not to say	
		1= White-British	
		2= White- Other	
		3= Mixed/Multiple ethnic groups	
Babies Ethnicity	BETHNIC	4=African/Caribbean/ Other Black	
		5=Asian-	
		Indian/Pakistani/Bangladeshi	
		6=East Asian- Chinese	
		7= Other	
Birth Weight to be completed in pounds or Kg?	BWT	1=Pounds and Ounces	
		2=Kg	
Maternal Smoking in Pregnancy	MSMOKE	0= No	
		1= Yes	
Other Household smoker in pregnancy	OSMOKE	0= No	
		1= Yes	
		1=Inside the home	
	y LSMOKE	2= Outside the home	
Location of other household smoke in pregnancy		3= Inside the car 4=Inside another location	
		(work/social)	
		5= Other	
Broast Food	RE	0= No	
Dicastreed	Ы	1= Yes	
		1= <1 Month	
Breast feeding duration	BFT	2=1-4 Months	
		3= >4 Months	
		0= No Qualifications	
		1= Completed secondary school	
		2=GCSE	
		3=Vocational	
Highest Maternal Qualification	MQUAL	/NVQ	
		4=A-Levels	
		5=Undergraduate Degree	
		6=Postgraduate Degree	
		7=Other	
Index of Multiple Deprivation Decile	IMD	Least Deprived 1-10 Most Deprived	
Nursery Attendance	NURSERY	0= No	
		1= Yes	
--	----------	------------------------------	--
Other children living in the Household?		0= No	
	OCHIED	1= Yes	
Number of other children living in the household	NUMCHILD	1-7(or more)	
Doos the shild share a bodroom?		0= No	
Does the child share a bearbonn:	JDED	1= Yes	
		0=Parent/Parents	
Who does the child share a bodroom with?	WSBED	1= 1 other Child	
who does the child share a bedroom with:		2= 2 or more children	
		3= Other	
Doos the baby bays any other health condition	DUCON	0= No	
bes the baby have any other health condition	внеом	1= Yes	
		1=Respiratory Condition	
		2=Heart Condition	
		3=Kidney Condition	
Type of health condition	THCON	4=Neurological Condition	
		5=Gastrointestinal Condition	
		6= Diabetes	
		7= Other Health Condition	
	FHATOPY	0= No	
is there a family history of atopy?		1= Yes	
	WATOPY	1= Babies Mother	
		2= Babies Father	
		3=My eldest Child	
Who has atopy in the family?		4=My Second Eldest Child	
		5= My Third eldest child	
		6= My 4th eldest Child	
		7= Other	
What atopic conditions does mother have	ΜΑΤΟΡΥ	1=Asthma	
What atopic conditions does father have	FATOPY		
What atopic conditions does eldest child have?	ECATOPY		
Wat atopic conditions does second eldest child have?	SECATOPY	2=Hay Fever	
What atopic Conditions does third eldest child have	ТЕСАТОРУ		

What atopic conditions does forth eldest child have?	FECATOPY	3=Eczema	
What atopic conditions does the their family member have?	ΟΑΤΟΡΥ		
Has there been smokers in the household during the	HSMOKE	0= No	
last 3 months?		1= Yes	
Where do the household smokers smoke?		1=Inside the home	
		2= Outside the home	
	HLSMOKE	3= Inside the car	
		4=Inside another location (work/social)	
		5= Other	
Have you been the GP with a respiratory condition	RCD	0= No	
the last 3 months?	bor	1= Yes	
Have you attended the hospital with respiratory	BHOS	0= No	
conditions in the last 3 months?	0103	1= Yes	

	Symptoms assessed on a Linkert Scale
Not at all	0
A few Days	1
Some Days	2
Most Days	3
Every Day	4

IMD	4 Months	10 Months	16 Months	22 Months
1	1.40	1.43	1.43	1.40
2	0.77	0.69	0.82	0.88
3	0.90	1.23	0.80	0.93
4	0.91	0.81	0.92	0.91
5	0.82	0.80	0.79	0.90
6	0.66	0.77	0.92	0.79
7	0.64	0.56	0.58	0.49
8	0.63	0.48	0.53	0.58
9	1.08	0.86	0.72	0.48
10	0.37	0.23	0.51	0.71

Appendix 16- Cross-Sectional Analysis Weighting Factors

References

1. Powell CVE, McNamara P, Solis A, Shaw NJ. A parent completed questionnaire to describe the patterns of wheezing and other respiratory symptoms in infants and preschool children. Archives of Disease in Childhood. 2002;87(5):376-9.

2. Pickles RE. Design, Development and Initiation of the Liverpool Respiratory Birth Cohort Study: Philosophy University of Liverpool; 2013.

3. Griffith B. The Liverpool Respiratory Birth Cohort Study- Questionnaire Development, Recruitment and Preliminary Results: Masters of Philosophy. University of Liverpool; 2013.

4. Uphoff E, Cabieses B, Pinart M, Valdes M, Anto JM, Wright J. A systematic review of socioeconomic position in relation to asthma and allergic diseases. The European respiratory journal. 2015;46(2):364-74.

5. Lannerö E, Wickman M, Pershagen G, Nordvall L. Maternal smoking during pregnancy increases the risk of recurrent wheezing during the first years of life (BAMSE). Respiratory Research. 2006;7(1):3-.

6. Bielska DE, Kurpas D, Nitsch-Osuch A, Gomółka E, Ołdak E, Chlabicz S, et al. Exposure to environmental tobacco smoke and respiratory tract infections in pre-school children - a cross-sectional study in Poland. Ann Agric Environ Med. 2015;22(3):524-9.

7. Fisk CM, Crozier SR, Inskip HM, Godfrey KM, Cooper C, Roberts GC, et al. Breastfeeding and reported morbidity during infancy: findings from the Southampton Women's Survey. Maternal & Child Nutrition. 2011;7(1):61-70.

8. Robison RG, Kumar R, Arguelles LM, Hong X, Wang G, Apollon S, et al. Maternal Smoking during Pregnancy, Prematurity and Recurrent Wheezing in Early Childhood. Pediatric pulmonology. 2012;47(7):666-73.

9. Lifestyles Statistics Team HaSCIC. Statistics on Women's Smoking at Time of Delivery: England- Quarter 2, July 2015 to September 2015. 2015 16 December 2015.

10. Statistics on Women's Smoking at Time of Delivery: England- Quarter 2, July 2015 to September 2015. Lifestyles Statistics Team, Health and Social Care Information Centre; 2015 16 December 2015.

11. Health Profile- Liverpool. http://fingertipsreports.phe.org.uk/ . Department of Health; 2012. 30/11/2016

12. Breastfeeding initiation and prevalence at 6 to 8 weeks-Quarter 4 2013/14. www.england.nhs.uk: NHS England; 2014. [Cited 23/11/2016]

13. England PH. Health Profile 2016- Liverpool. <u>http://fingertipsreports.phe.org.uk/</u>; 2016 [Cited 6/09/2016].

14. England PH. Interactive Health Atlas of Lung Conditions in England- NHS Liverpool CCG <u>https://fingertips.phe.org.uk</u>: Public Health England; 2016. [Cited 20/11/2016]

15.Davies SC, Lemer C, Hargreaves D, Cheung R, Strelitz J, Weil L, et al. Annual Report of the ChiefMedicalOfficer2012.

https://www.gov.uk/government/uploads/system/uploads/attachment_data/file/255237/2901304 <u>CMO_complete_low_res_accessible.pdf;</u> 2013. [Cited 20/11/2016]

16. Trinick R, Southern KW, McNamara PS. Assessing the Liverpool Respiratory Symptom Questionnaire in children with cystic fibrosis. European Respiratory Journal. 2012;39(4):899.

17. Trinick RE, Bunni L, Thorburn K, Hackett AP, Dalzell M, McNamara PS. An Observational Study Examining the Relationship between Respiratory Symptoms, Airway Inflammation and Bacteriology in Children with Severe Neurodisability. PLoS ONE. 2015;10(4):e0124627.

18. Brooke AM, Lambert PC, Burton PR, Clarke C, Luyt DK, Simpson H. Recurrent cough: Natural history and significance in infancy and early childhood. Pediatric Pulmonology. 1998;26(4):256-61.

19. Hay AD, Wilson AD. The natural history of acute cough in children aged 0 to 4 years in primary care: a systematic review. The British Journal of General Practice. 2002;52(478):401-9.

20. Martinez FD, Wright AL, Taussig LM, Holberg CJ, Halonen M, Morgan WJ, et al. Asthma and Wheezing in the First Six Years of Life. New England Journal of Medicine. 1995;332(3):133-8.

21. Brand PLP, Baraldi E, Bisgaard H, Boner AL, Castro-Rodriguez JA, Custovic A, et al. Definition, assessment and treatment of wheezing disorders in preschool children: an evidence-based approach. European Respiratory Journal. 2008;32(4):1096-110.

22. Sonnappa S. Preschool wheeze: phenotypes and beyond. Pediatric Health. 2010;4(3):267-75.

23. Society BT. Burden of Lung Disease <u>https://www.brit-thoracic.org.uk/working-in-respiratory/burden-of-lung-disease/2006</u> [Available from: <u>https://www.brit-thoracic.org.uk/working-in-respiratory/burden-of-lung-disease/</u>. [Cited 20/11/2016]

24. Custovic A, Simpson BM, Murray CS, Lowe L, Woodcock A, on behalf of the NACMA, et al. The National Asthma Campaign Manchester Asthma and Allergy Study. Pediatric Allergy and Immunology. 2002;13:32-7.

25. Mebrahtu TF, Feltbower RG, Parslow RC. Incidence and Burden of Wheezing Disorders, Eczema, and Rhinitis in Children: findings from the Born in Bradford Cohort. Paediatric and Perinatal Epidemiology. 2016;30(6):594-602.

26. Ricci V, Delgado Nunes V, Murphy MS, Cunningham S. Bronchiolitis in children: summary of NICE guidance. BMJ : British Medical Journal. 2015;350.

27. Taylor S, Taylor RJ, Lustig RL, Schuck-Paim C, Haguinet F, Webb DJ, et al. Modelling estimates of the burden of respiratory syncytial virus infection in children in the UK. BMJ Open. 2016;6(6).

28. Redondo M, Keyt H, Dhar R, Chalmers JD. Global impact of bronchiectasis and cystic fibrosis. Breathe. 2016;12(3):222-35.

29.OrganizationTWH.Childmortalityhttp://www.who.int/maternal_child_adolescent/topics/child/mortality/en/:WHO; 2017 [cited 201709/06/2017].Availablefrom:

http://www.who.int/maternal_child_adolescent/topics/child/mortality/en/.[Cited 22/11/2016]

30. Oostenbrink R, Jansingh-Piepers EM, Raat H, Nuijsink M, Landgraf JM, Essink-Bot ML, et al. Health-related quality of life of pre-school children with wheezing illness. Pediatr Pulmonol. 2006;41(10):993-1000.

 van Gent R, van Essen LE, Rovers MM, Kimpen JL, van der Ent CK, de Meer G. Quality of life in children with undiagnosed and diagnosed asthma. European journal of pediatrics. 2007;166(8):843-8.
 Fleming L, Murray C, Bansal AT, Hashimoto S, Bisgaard H, Bush A, et al. The burden of severe asthma in childhood and adolescence: Results from the paediatric U-BIOPRED cohorts. European

Respiratory Journal. 2015;46(5):1322-33.

33. Benito-Fernández J, Mojica-Muñoz E, Andrés-Olaizola A, González-Balenciaga M, Urrutia-Adan M, Martinez-Indart L, et al. Impact on quality of life by improving asthma control medication in patients with persistent asthma in a paediatric emergency department. European Journal of Emergency Medicine. 2013;20(5):350-5.

34. Brabin B, Smith M, Milligan P, Benjamin C, Dunne E, Pearson M. Respiratory morbidity in Merseyside schoolchildren exposed to coal dust and air pollution. Archives of Disease in Childhood. 1994;70(4):305-12.

35. Venn AJ, Lewis SA, Cooper R M, Hubbard R, Britton J. Living Near a Main Road and the Risk of Wheezing Illness in Children. American Journal of Respiratory and Critical Care Medicine. 2001;164(12):2177-80.

36. van der Zee S, Hoek G, Boezen HM, Schouten JP, van Wijnen JH, Brunekreef B. Acute effects of urban air pollution on respiratory health of children with and without chronic respiratory symptoms. Occupational and Environmental Medicine. 1999;56(12):802-12.

37. Fischer P, Steerenberg P, Snelder J, van Loveren H, van Amsterdam J. Association between exhaled nitric oxide, ambient air pollution and respiratory health in school children. International Archives of Occupational and Environmental Health. 2002;75(5):348-53.

38. Sunyer J, Spix C, Quénel P, Ponce-de-León A, Pönka A, Barumandzadeh T, et al. Urban air pollution and emergency admissions for asthma in four European cities: the APHEA Project. Thorax. 1997;52(9):760-5.

39. Brew BK, Allen CW, Toelle BG, Marks GB. Systematic review and meta-analysis investigating breast feeding and childhood wheezing illness. Paediatric and Perinatal Epidemiology. 2011;25(6):507-18.

40. Wright AL, Holberg CJ, Martinez FD, Morgan WJ, Taussig LM. Breast feeding and lower respiratory tract illness in the first year of life. Group Health Medical Associates. British Medical Journal. 1989;299(6705):946-9.

41. Sonnenschein-van der Voort AMM, Jaddoe VWV, van der Valk RJP, Willemsen SP, Hofman A, Moll HA, et al. Duration and exclusiveness of breastfeeding and childhood asthma-related symptoms. European Respiratory Journal. 2011;39(1):81.

42. Ferrante G, Antona R, Malizia V, Montalbano L, Corsello G, La Grutta S. Smoke exposure as a risk factor for asthma in childhood: a review of current evidence. Allergy and asthma proceedings. 2014;35(6):454-61.

43. Lux AL, Henderson AJ, Pocock SJ, Team tAS. Wheeze associated with prenatal tobacco smoke exposure: a prospective, longitudinal study. Archives of Disease in Childhood. 2000;83(4):307-12.

44. Werhmeister FC, Nunes BP, Loret de Mola C, Gómez-Cofré N, de Oliveira PD, Marco PL, et al. Intrauterine exposure to smoking and wheezing in adolescence: the 1993 Pelotas Birth Cohort. Journal of Developmental Origins of Health and Disease. 2015;6(3):217-24.

45. Taussig LM, Wright AL, Holberg CJ, Halonen M, Morgan WJ, Martinez FD. Tucson children's respiratory study: 1980 to present. Journal of Allergy and Clinical Immunology. 2003;111(4):661-75.

46. Lewis S, Richards D, Bynner J, Butler N, Britton J. Prospective study of risk factors for early and persistent wheezing in childhood. The European respiratory journal. 1995;8(3):349-56.

47. Murray CS, Woodcock A, Smillie FI, Cain G, Kissen P, Custovic A. Tobacco smoke exposure, wheeze, and atopy. Pediatr Pulmonol. 2004;37(6):492-8.

48. Lin Z, Norback D, Wang T, Zhang X, Shi J, Kan H, et al. The first 2-year home environment in relation to the new onset and remission of asthmatic and allergic symptoms in 4246 preschool children. Science of The Total Environment. 2016;553:204-10.

49. Wright J, Small N, Raynor P, Tuffnell D, Bhopal R, Cameron N, et al. Cohort Profile: the Born in Bradford multi-ethnic family cohort study. Int J Epidemiol. 2013;42(4):978-91.

50. Pramana IA, Latzin P, Schlapbach LJ, Hafen G, Kuehni CE, Nelle M, et al. Respiratory symptoms in preterm infants: burden of disease in the first year of life. European Journal of Medical Research. 2011;16(5):223-30.

51. Islam JY, Keller RL, Aschner JL, Hartert TV, Moore PE. Understanding the Short- and Long-Term Respiratory Outcomes of Prematurity and Bronchopulmonary Dysplasia. American Journal of Respiratory and Critical Care Medicine. 2015;192(2):134-56.

52. Xu X-F, Li Y-J, Sheng Y-J, Liu J-L, Tang L-F, Chen Z-M. Effect of low birth weight on childhood asthma: a meta-analysis. BMC Pediatrics. 2014;14:275.

53. Schaubel D, Johansen H, Dutta M, Desmeules M, Becker A, Mao Y. Neonatal Characteristics as Risk Factors for Preschool Asthma. Journal of Asthma. 1996;33(4):255-64.

54. Pike KC, Crozier SR, Lucas JS, Inskip HM, Robinson S, Roberts G, et al. Patterns of fetal and infant growth are related to atopy and wheezing disorders at age 3 years. Thorax. 2010;65(12):1099-106.

55. Mebrahtu TF, Feltbower RG, Parslow RC. Effects of birth weight and growth on childhood wheezing disorders: findings from the Born in Bradford Cohort. BMJ Open. 2015;5(11).

56. Cable N, Kelly Y, Bartley M, Sato Y, Sacker A. Critical role of smoking and household dampness during childhood for adult phlegm and cough: a research example from a prospective cohort study in Great Britain. BMJ Open. 2014;4(4).

57. Mann SL, Wadsworth ME, Colley JR. Accumulation of factors influencing respiratory illness in members of a national birth cohort and their offspring. Journal of Epidemiology and Community Health. 1992;46(3):286-92.

58. Lamont DW, Parker L, Cohen MA, White M, Bennett SMA, Unwin NC, et al. Early life and later determinants of adult disease. Public Health. 1998;112(2):85-93.

59. Cardoso MRA, Cousens SN, de Góes Siqueira LF, Alves FM, D'Angelo LAV. Crowding: risk factor or protective factor for lower respiratory disease in young children? BMC Public Health. 2004;4:19-.

60. Castro-Rodriguez JA, Holberg CJ, Wright AL, Halonen M, Taussig LM, Morgan WJ, et al. Association of Radiologically Ascertained Pneumonia Before Age 3 yr with Asthmalike Symptoms and Pulmonary Function During Childhood. American Journal of Respiratory and Critical Care Medicine. 1999;159(6):1891-7.

61. Strachan DP, Aït-Khaled N, Foliaki S, Mallol J, Odhiambo J, Pearce N, et al. Siblings, asthma, rhinoconjunctivitis and eczema: a worldwide perspective from the International Study of Asthma and Allergies in Childhood. Clinical and Experimental Allergy. 2015;45(1):126-36.

62. Luijk MP, Sonnenschein-van der Voort AM, Mileva-Seitz VR, Jansen PW, Verhulst FC, Hofman A, et al. Is parent-child bed-sharing a risk for wheezing and asthma in early childhood? The European respiratory journal. 2015;45(3):661-9.

63. Weiland SK, Husing A, Strachan DP, Rzehak P, Pearce N. Climate and the prevalence of symptoms of asthma, allergic rhinitis, and atopic eczema in children. Occup Environ Med. 2004;61(7):609-15.

64. Gehring U, Strikwold M, Schram-Bijkerk D, Weinmayr G, Genuneit J, Nagel G, et al. Asthma and allergic symptoms in relation to house dust endotoxin: Phase Two of the International Study on Asthma and Allergies in Childhood (ISAAC II). Clinical and experimental allergy : journal of the British Society for Allergy and Clinical Immunology. 2008;38(12):1911-20.

65. Galobardes B, Granell R, Sterne J, Hughes R, Mejia-Lancheros C, Davey Smith G, et al. Childhood wheezing, asthma, allergy, atopy, and lung function: different socioeconomic patterns for different phenotypes. Am J Epidemiol. 2015;182(9):763-74.

66. Violato M, Petrou S, Gray R. The relationship between household income and childhood respiratory health in the United Kingdom. Soc Sci Med. 2009;69(6):955-63.

67. Sherriff A, Golding J. Hygiene levels in a contemporary population cohort are associated with wheezing and atopic eczema in preschool infants. Arch Dis Child. 2002;87(1):26-9.

68. Rook GA, Martinelli R, Brunet LR. Innate immune responses to mycobacteria and the downregulation of atopic responses. Curr Opin Allergy Clin Immunol. 2003;3(5):337-42.

69. Strachan DP. Hay fever, hygiene, and household size. BMJ : British Medical Journal. 1989;299(6710):1259-60.

70. Indinnimeo L, Porta D, Forastiere F, De Vittori V, De Castro G, Zicari AM, et al. Prevalence and risk factors for atopic disease in a population of preschool children in Rome: Challenges to early intervention. International Journal of Immunopathology and Pharmacology. 2016;29(2):308-19.

71. Tse SM, Coull BA, Sordillo JE, Datta S, Gold DR. Gender- and age-specific risk factors for wheeze from birth through adolescence. Pediatric pulmonology. 2015;50(10):955-62.

72. Wright AL, Holberg CJ, Taussig LM, Martinez FD. Factors influencing the relation of infant feeding to asthma and recurrent wheeze in childhood. Thorax. 2001;56(3):192-7.

73. Nicolaou NC, Simpson A, Lowe LA, Murray CS, Woodcock A, Custovic A. Day-care attendance, position in sibship, and early childhood wheezing: a population-based birth cohort study. The Journal of allergy and clinical immunology. 2008;122(3):500-6.e5.

74. Caudri D, Wijga A, Scholtens S, Kerkhof M, Gerritsen J, Ruskamp JM, et al. Early daycare is associated with an increase in airway symptoms in early childhood but is no protection against asthma or atopy at 8 years. Am J Respir Crit Care Med. 2009;180(6):491-8.

75. Christopher Lai NP. The contribution of ISAAC to the understanding of asthma. Leukotriene Research & Clin Review. 2001;2(1):1-4.

76. Nagel G, Weinmayr G, Kleiner A, Garcia-Marcos L, Strachan DP. Effect of diet on asthma and allergic sensitisation in the International Study on Allergies and Asthma in Childhood (ISAAC) Phase Two. Thorax. 2010;65(6):516-22.

77. Langley SJ, Goldthorpe S, Craven M, Morris J, Woodcock A, Custovic A. Exposure and sensitization to indoor allergens: association with lung function, bronchial reactivity, and exhaled nitric oxide measures in asthma. The Journal of allergy and clinical immunology. 2003;112(2):362-8.

78. Langley SJ, Goldthorpe S, Craven M, Woodcock A, Custovic A. Relationship between exposure to domestic allergens and bronchial hyperresponsiveness in non-sensitised, atopic asthmatic subjects. Thorax. 2005;60(1):17-21.

79. Taylor-Robinson DC, Pearce A, Whitehead M, Smyth R, Law C. Social inequalities in wheezing in children: findings from the UK Millennium Cohort Study. The European respiratory journal. 2016;47(3):818-28.

80. Beydon N, Davis SD, Lombardi E, Allen JL, Arets HG, Aurora P, et al. An official American Thoracic Society/European Respiratory Society statement: pulmonary function testing in preschool children. Am J Respir Crit Care Med. 2007;175(12):1304-45.

81. Roberto DM, Scarlett GA, Ana MI, Laura FT, Pablo BV, Solange CL. Spirometry interpretation feasibility among pre-school children according to the European respiratory society and american thoracic society guidelines. Revista Chilena de Pediatria. 2015;86(2):86-91.

van den Wijngaart LS, Roukema J, Merkus PJFM. Respiratory disease and respiratory physiology: Putting lung function into perspective: Paediatric asthma. Respirology. 2015;20(3):379-88.
 Allen JL, Wolfson MR, Mcdowell K, Shaffer TH, Motley wttaoR. Thoracoabdominal Asynchrony

in Infants with Airflow Obstruction. American Review of Respiratory Disease. 1990;141(2):337-42. 84. Manczur T, Greenough A, Hooper R, Allen K, Latham S, Price JF, et al. Tidal breathing parameters in young children: Comparison of measurement by respiratory inductance plethysmography to a facemask pneumotachograph system. Pediatric Pulmonology. 1999;28(6):436-41. 85. Beydon N. Pulmonary function testing in young children. Paediatric Respiratory Reviews. 2009;10(4):208-13.

86. Calogero C, Simpson SJ, Lombardi E, Parri N, Cuomo B, Palumbo M, et al. Respiratory impedance and bronchodilator responsiveness in healthy children aged 2-13 years. Pediatric Pulmonology. 2013;48(7):707-15.

87. Peták F, Czövek D, Novák Z. Spirometry and forced oscillations in the detection of airway hyperreactivity in asthmatic children. Pediatric Pulmonology. 2012;47(10):956-65.

88. Aurora P, Bush A, Gustafsson P, Oliver C, Wallis C, Price J, et al. Multiple-breath washout as a marker of lung disease in preschool children with cystic fibrosis. Am J Respir Crit Care Med. 2005;171(3):249-56.

89. Baraldi E, de Jongste JC, Gaston B, Alving K, Barnes PJ, Bisgaard H, et al. Measurement of exhaled nitric oxide in children, 2001. E Baraldi and JC de Jongste on behalf of the Task Force. 2002;20(1):223-37.

90. Juniper E, O'Byrne P, Guyatt G, Ferrie P, King D. Development and validation of a questionnaire to measure asthma control. European Respiratory Journal. 1999;14(4):902-7.

91. Juniper EF, Gruffydd-Jones K, Ward S, Svensson K. Asthma Control Questionnaire in children: validation, measurement properties, interpretation. European Respiratory Journal. 2010;36(6):1410-6.

92. Juniper EF, Bousquet J, Abetz L, Bateman ED. Identifying 'well-controlled' and 'not well-controlled' asthma using the Asthma Control Questionnaire. Respiratory Medicine. 2006;100(4):616-21.

93. Skinner EA, Diette GB, Algatt-Bergstrom PJ, Nguyen TT, Clark RD, Markson LE, et al. The Asthma Therapy Assessment Questionnaire (ATAQ) for children and adolescents. Disease management : DM. 2004;7(4):305-13.

94. Kachru R, Morphew T, Kehl S, Clement LT, Hanley-Lopez J, Kwong KY, et al. Validation of a single survey that can be used for case identification and assessment of asthma control: the Breathmobile Program. Annals of allergy, asthma & immunology : official publication of the American College of Allergy, Asthma, & Immunology. 2006;97(6):775-83.

95. Jones CA, Morphew T, Clement LT, Kimia T, Dyer M, Li M, et al. A School-Based Case Identification Process for Identifying Inner City Children With Asthma: The Breathmobile Program. Chest. 2004;125(3):924-34.

96. Liu AH, Zeiger R, Sorkness C, Mahr T, Ostrom N, Burgess S, et al. Development and crosssectional validation of the Childhood Asthma Control Test. Journal of Allergy and Clinical Immunology. 2007;119(4):817-25.

97. Liu AH, Zeiger RS, Sorkness CA, Ostrom NK, Chipps BE, Rosa K, et al. The Childhood Asthma Control Test*: Retrospective determination and clinical validation of a cut point to identify children with very poorly controlled asthma. Journal of Allergy and Clinical Immunology. 2010;126(2):267-73.e1.

98. French DJ, Christie MJ, Sowden AJ. The reproducibility of the Childhood Asthma Questionnaires: measures of quality of life for children with asthma aged 4–16 years. Quality of Life Research. 1994;3(3):215-24.

99. M J Christie, D French, A Sowden, West A. Development of child-centered disease-specific questionnaires for living with asthma. Psychosomatic Medicine. 1993;55(6):541-8.

 Newcombe PA, Sheffield JK, Juniper EF, Petsky HL, Willis C, Chang AB. Validation of a parentproxy quality of life questionnaire for paediatric chronic cough (PC-QOL). Thorax. 2010;65(9):819-23.
 Newcombe PA, Sheffield JK, Petsky HL, Marchant JM, Willis C, Chang AB. A child chronic cough-specific quality of life measure: development and validation. Thorax. 2016;71(8):695-700.

102. Bianca AC, Wandalsen GF, Miyagi K, Camargo L, Cezarin D, Mallol J, et al. International Study of Wheezing in Infants (EISL): validation of written questionnaire for children aged below 3 years. Journal of investigational allergology & clinical immunology. 2009;19(1):35-42.

103. Chong Neto HJ, Rosario N, Dela Bianca AC, Solé D, Mallol J. Validation of a questionnaire for epidemiologic studies of wheezing in infants. Pediatric Allergy and Immunology. 2007;18(1):86-7.

104. Mallol J, García-Marcos L, Aguirre V, Martinez-Torres A, Perez-Fernández V, Gallardo A, et al. The International Study of Wheezing in Infants: Questionnaire Validation. International Archives of Allergy and Immunology. 2007;144(1):44-50. 105. Asher M, Keil U, Anderson H, Beasley R, Crane J, Martinez F, et al. International Study of Asthma and Allergies in Childhood (ISAAC): rationale and methods. European Respiratory Journal. 1995;8(3):483-91.

106. Pearce N, Weiland S, Keil U, Langridge P, Anderson H, Strachan D, et al. Self-reported prevalence of asthma symptoms in children in Australia, England, Germany and New Zealand: an international comparison using the ISAAC protocol. European Respiratory Journal. 1993;6(10):1455-61.

107. Gibson PG, Henry R, Shah S, Toneguzzi R, Francis JL, Norzila MZ, et al. Validation of the ISAAC video questionnaire (AVQ3.0) in adolescents from a mixed ethnic background. Clinical and experimental allergy : journal of the British Society for Allergy and Clinical Immunology. 2000;30(8):1181-7.

108. Kuehni CE, Brooke AM, Strippoli M-PF, Spycher BD, Davis A, Silverman M. Cohort Profile: The Leicester Respiratory Cohorts. International Journal of Epidemiology. 2007;36(5):977-85.

109. Luyt DK, Burton PR, Simpson H. Epidemiological study of wheeze, doctor diagnosed asthma, and cough in preschool children in Leicestershire. BMJ : British Medical Journal. 1993;306(6889):1386-90.

110. Strippoli MPF, Silverman M, Michel G, Kuehni CE. A parent - completed respiratory

questionnaire for 1 - year - old children: repeatability. Archives of Disease in Childhood. 2007;92(10):861-5.

111. Hohmann C, Pinart M, Tischer C, Gehring U, Heinrich J, Kull I, et al. The Development of the MeDALL Core Questionnaires for a Harmonized Follow-Up Assessment of Eleven European Birth Cohorts on Asthma and Allergies. International Archives of Allergy and Immunology. 2014;163(3):215-24.

112. Hartnick CJ, Zurakowski D, Haver K. Validation of a pediatric cough questionnaire. Ear, nose, & throat journal. 2009;88(11):1213-7.

113. Juniper EF, Guyatt GH, Feeny DH, Ferrie PJ, Griffith LE, Townsend M. Measuring quality of life in children with asthma. Quality of life research : an international journal of quality of life aspects of treatment, care and rehabilitation. 1996;5(1):35-46.

114. Everhart RS, Fiese BH. Development and initial validation of a pictorial quality of life measure for young children with asthma. Journal of pediatric psychology. 2009;34(9):966-76.

115. Murphy KR, Zeiger RS, Kosinski M, Chipps B, Mellon M, Schatz M, et al. Test for Respiratory and Asthma Control in Kids (TRACK): A caregiver-completed questionnaire for preschool-aged children. Journal of Allergy and Clinical Immunology. 2009;123(4):833-9.e9.

116. Zeiger RS, Mellon M, Chipps B, Murphy KR, Schatz M, Kosinski M, et al. Test for Respiratory and Asthma Control in Kids (TRACK): Clinically meaningful changes in score. Journal of Allergy and Clinical Immunology. 2011;128(5):983-8.

117. Chipps B, Zeiger RS, Murphy K, Mellon M, Schatz M, Kosinski M, et al. Longitudinal Validation of the Test for Respiratory and Asthma Control in Kids in Pediatric Practices. Pediatrics. 2011.

118. Van De Kant KD, Klaassen EM, Jöbsis Q, Nijhuis AJ, Van Schayck OC, Dompeling E. Early diagnosis of asthma in young children by using non-invasive biomarkers of airway inflammation and early lung function measurements: Study protocol of a case-control study. BMC Public Health. 2009;9. 119. Kairamkonda VR, Richardson J, Subhedar N, Bridge PD, Shaw NJ. Lung function measurement in prematurely born preschool children with and without chronic lung disease. J Perinatol. 2008;28(3):199-204.

120. van de Kant KDG, Koers K, Rijkers GT, Lima Passos V, Klaassen EMM, Mommers M, et al. Can exhaled inflammatory markers predict a steroid response in wheezing preschool children? Clinical & Experimental Allergy. 2011;41(8):1076-83.

121. Hutchinson SG, Mesters I, Van Breukelen G, Muris JW, Feron FJ, Hammond SK, et al. A motivational interviewing intervention to PREvent PAssive Smoke Exposure (PREPASE) in children with a high risk of asthma: Design of a randomised controlled trial. BMC Public Health. 2013;13(1).

122. Storrø O, Øien T, Dotterud CK, Jenssen JA, Johnsen R. A primary health-care intervention on pre- and postnatal risk factor behavior to prevent childhood allergy. The Prevention of Allergy among Children in Trondheim (PACT) study. BMC Public Health. 2010;10:443-.

123. Elphick HE, Ritson S, Rigby AS, Everard ML. Phenotype of acute respiratory syncytial virus induced lower respiratory tract illness in infancy and subsequent morbidity. Acta Paediatrica, International Journal of Paediatrics. 2007;96(2):307-9.

124. Gunawardhana LP, Baines KJ, Mattes J, Murphy VE, Simpson JL, Gibson PG. Differential DNA methylation profiles of infants exposed to maternal asthma during pregnancy. Pediatr Pulmonol. 2014;49(9):852-62.

125. Mattes J, Murphy VE, Powell H, Gibson PG. Prenatal origins of bronchiolitis: Protective effect of optimised asthma management during pregnancy. Thorax. 2014;69(4):383-4.

126. Roberts D, Vause S, Martin W, Green P, Walkinshaw S, Bricker L, et al. Amnioinfusion in preterm premature rupture of membranes (AMIPROM): A randomised controlled trial of amnioinfusion versus expectant management in very early preterm premature rupture of membranes - A pilot study. Health Technology Assessment. 2014;18(21):1-135.

127. Thorburn AN, McKenzie CI, Shen S, Stanley D, Macia L, Mason LJ, et al. Evidence that asthma is a developmental origin disease influenced by maternal diet and bacterial metabolites. Nature communications. 2015;6:7320.

128. Turner SW, Carter J, Danielian P, Chalmers I, McConaghy L, Pacitti N, et al. Protease concentration in amniotic fluid at term and early childhood respiratory symptoms. The Journal of Maternal-Fetal & Neonatal Medicine. 2014;27(4):416-20.

129. Trinick R, Johnston N, Dalzell AM, McNamara PS. Reflux aspiration in children with neurodisability—a significant problem, but can we measure it? Journal of Pediatric Surgery. 2012;47(2):291-8.

130. Mandaliya PH, Morten M, Kumar R, James A, Deshpande A, Murphy VE, et al. Ventilation inhomogeneities in children with congenital thoracic malformations. BMC Pulmonary Medicine. 2015;15(1).

131. Percival E, Bhatia R, Preece K, McElduff P, McEvoy M, Collison A, et al. Reproducibility of serum IgE, Ara h2 skin prick testing and fraction of exhaled nitric oxide for predicting clinical peanut allergy in children. Allergy, Asthma and Clinical Immunology. 2016;12(1).

132. Preece K, Bhatia R, Belcher J, Patchett K, McElduff P, Collison A, et al. The fraction of exhaled nitric oxide improves prediction of clinical allergic reaction to peanut challenge in children. Clinical & Experimental Allergy. 2014;44(3):371-80.

133. Smith AM, Bernstein DI, LeMasters GK, Huey NL, Ericksen M, Villareal M, et al. Environmental Tobacco Smoke and Interleukin 4 Polymorphism (C-589T) Gene: Environment Interaction Increases Risk of Wheezing in African-American Infants. The Journal of Pediatrics. 2008;152(5):709-15.e1.

134. Wadsworth M, Kuh D, Richards M, Hardy R. Cohort Profile: The 1946 National Birth Cohort (MRC National Survey of Health and Development). Int J Epidemiol. 2006;35(1):49-54.

135. Power C, Elliott J. Cohort profile: 1958 British birth cohort (National Child Development Study). Int J Epidemiol. 2006;35(1):34-41.

136. Elliott J, Shepherd P. Cohort profile: 1970 British Birth Cohort (BCS70). Int J Epidemiol. 2006;35(4):836-43.

137. Connelly R, Platt L. Cohort Profile: UK Millennium Cohort Study (MCS). International Journal of Epidemiology. 2014.

138. Leon DA, Lawlor DA, Clark H, Macintyre S. Cohort Profile: The Aberdeen Children of the 1950s Study. International Journal of Epidemiology. 2006;35(3):549-52.

139. Fraser A, Macdonald-Wallis C, Tilling K, Boyd A, Golding J, Davey Smith G, et al. Cohort Profile: the Avon Longitudinal Study of Parents and Children: ALSPAC mothers cohort. Int J Epidemiol. 2013;42(1):97-110.

140. Goodfellow SA, Rolfe EM, Golding J. Cohort profile: The Isle of Man Birth Cohort Study. Int J Epidemiol. 2013;42(5):1246-52.

141. Pearce MS, Unwin NC, Parker L, Craft AW. Cohort Profile: The Newcastle Thousand Families 1947 Birth Cohort. International Journal of Epidemiology. 2009;38(4):932-7.

142. Inskip HM, Godfrey KM, Robinson SM, Law CM, Barker DJ, Cooper C, et al. Cohort profile: The Southampton Women's Survey. International Journal of Epidemiology. 2006;35(1):42-8.

143. Manchester Asthma and Allergy Study ResearchTeam . Manchester Asthma And Allergy Study <u>http://www.maas.org.uk/default.html</u> [Available from: <u>http://www.maas.org.uk/default.html</u> [Cited 08/12/2016].

144. Wijga AH, Kerkhof M, Gehring U, de Jongste JC, Postma DS, Aalberse RC, et al. Cohort profile: The Prevention and Incidence of Asthma and Mite Allergy (PIAMA) birth cohort. International Journal of Epidemiology. 2014;43(2):527-35.

145. Government DfCaL. Indices of Multiple Deprivation 2015. In: Executive TOotC, editor. <u>https://liverpool.gov.uk/council/key-statistics-and-data/data/deprivation/2016</u>. [Cited 10/01/2017]

146. Public Health England . Child Health Profiles- Liverpool <u>https://fingertips.phe.org.uk</u>: Public Health England; 2016 [Available from: <u>https://fingertips.phe.org.uk/profile/child-health-profiles/data#page/1/gid/1938132948/pat/6/par/E12000002/ati/102/are/E08000012</u>.[Cited 10/01/2017]

147. Delpisheh A, Kelly Y, Brabin BJ. Passive cigarette smoke exposure in primary school children in Liverpool. Public Health. 2006;120(1):65-9.

148. Milligan PJ, Brabin BJ, Kelly YJ, Pearson MG, Mahoney G, Dunne E, et al. Association of spatial distribution of childhood respiratory morbidity with environmental dust pollution. Journal of toxicology and environmental health Part A. 1998;55(3):169-84.

149. Clifford RD, Radford M, Howell JB, Holgate ST. Prevalence of respiratory symptoms among 7 and 11 year old schoolchildren and association with asthma. Arch Dis Child. 1989;64(8):1118-25.

150. Kelly YJ, Brabin BJ, Milligan P, Heaf DP, Reid J, Pearson MG. Maternal asthma, premature birth, and the risk of respiratory morbidity in schoolchildren in Merseyside. Thorax. 1995;50(5):525-30.

151. Rizwan S, Reid J, Kelly Y, Bundred PE, Pearson M, Brabin BJ. Trends in childhood and parental asthma prevalence in Merseyside, 1991-1998. Journal of Public Health. 2004;26(4):337-42.

152. Clague HW, el-Ansary EH, Hopkins CA, Roberts C. Pulmonary infection with opportunist mycobacteria on Merseyside 1974-1983. Postgraduate Medical Journal. 1986;62(727):363-8.

153. Office of National Statistics. Birth Characteristics. In: Statistics OfN, editor. <u>https://www.ons.gov.uk/peoplepopulationandcommunity/birthsdeathsandmarriages/livebirths/dat</u> <u>asets/birthcharacteristicsinenglandandwales2015</u>. [Cited 07/01/2017]

154. Department for Communities and Local Goverment . English Indices of Deprivation 2015. In: Goverment DfCaL, editor. <u>https://www.gov.uk/government/statistics/english-indices-of-deprivation-20152015</u>. [Cited 07/01/2017]

155. Department for Communities and Local Goverment. English indices of deprivation 2015 <u>http://imd-by-postcode.opendatacommunities.org</u>: Department For Communities and Local Goverment; 2015 [08/05/2017]. Available from: <u>http://imd-by-postcode.opendatacommunities.org/</u>. 156. NHS Digital. Statistics on Women's Smoking Status at Time of Delivery, England - Quarter 1, 2014-15

<u>http://content.digital.nhs.uk/searchcatalogue?productid=15384&q=smoking+pregnancy&sort=Most</u> <u>+recent&size=10&page=6#top</u>: NHS; 2014 [cited 2017 08/05/2017]. Available from: <u>http://content.digital.nhs.uk/searchcatalogue?productid=15384&q=smoking+pregnancy&sort=Most</u> <u>+recent&size=10&page=6#top</u>.

157.Offical Labour Market Statisitics. Labour Market Profile - Liverpool City Region
https://www.nomisweb.co.uk/reports/Imp/lep/1925185554/report.aspx#tabrespop2015 [cited 2017
08/05/2017].Cited 2017
from:

https://www.nomisweb.co.uk/reports/Imp/lep/1925185554/report.aspx#tabrespop.

158. Digital N. Maternity Services Monthly Statistics, England – April 2015, Experimental statistics: <u>http://content.digital.nhs.uk/catalogue/PUB19086</u>; 2015 [cited 2017 08/05/2017]. Available from: <u>http://content.digital.nhs.uk/catalogue/PUB19086</u>.

159. Perneger TV. What's wrong with Bonferroni adjustments. BMJ : British Medical Journal. 1998;316(7139):1236-8.

160. Organization TWH. Preterm birth <u>http://www.who.int/mediacentre/factsheets/fs363/en/</u>: WHO; 2016 [updated November 2016; [cited 2017 08/05/2017]. Available from: <u>http://www.who.int/mediacentre/factsheets/fs363/en/</u>.

161. McCormick MC. The contribution of low birth weight to infant mortality and childhood morbidity. The New England journal of medicine. 1985;312(2):82-90.

162. Boyd A, Golding J, Macleod J, Lawlor DA, Fraser A, Henderson J, et al. Cohort Profile: The 'Children of the 90s'—the index offspring of the Avon Longitudinal Study of Parents and Children. International Journal of Epidemiology. 2012.

163. Born In Braford. Born In Bradford <u>http://www.borninbradford.nhs.uk/</u> [updated 20/09/2016. Available from: <u>http://www.borninbradford.nhs.uk/</u>.

164. Raynor P. Born in Bradford, a cohort study of babies born in Bradford, and their parents: Protocol for the recruitment phase. BMC Public Health. 2008;8(1):327.

165. Logan DE, Claar RL, Scharff L. Social desirability response bias and self-report of psychological distress in pediatric chronic pain patients. Pain. 2008;136(3):366-72.

166. Jain R, Jhanjee S, Jain V, Gupta T, Mittal S, Chauhan P, et al. Biochemical Validation of Self-Reported Smokeless Tobacco Abstinence among Smokeless Tobacco Users: Results from a Clinical Trial of Varenicline in India. Journal of psychoactive drugs. 2015;47(4):331-5.

167. Medoua GN, Sajo Nana EC, Ndzana AC, Makamto CS, Etame LS, Rikong HA, et al. Breastfeeding practices of Cameroonian mothers determined by dietary recall since birth and the dose-to-the-mother deuterium-oxide turnover technique. Matern Child Nutr. 2012;8(3):330-9.

168. Wu V, Abo-Sido N, Espinola JA, Tierney CN, Tedesco KT, Sullivan AF, et al. Predictors of successful telephone follow-up in a multicenter study of infants with severe bronchiolitis. Ann Epidemiol. 2017.

169. Power C. A review of child health in the 1958 birth cohort: National Child Development Study. Paediatric and Perinatal Epidemiology. 1992;6(1):81-110.

170. Public Health Institute FoE, Health and Community. Children and Young People Health and Wellbeing Profile: Liverpool City Region. The Public Health Institute, Liverpool John Mores University; 2015.

171. Panico L, Bartley M, Marmot M, Nazroo JY, Sacker A, Kelly YJ. Ethnic variation in childhood asthma and wheezing illnesses: findings from the Millennium Cohort Study. International Journal of Epidemiology. 2007;36(5):1093-102.

172. Kuehni CE, Strippoli MP, Silverman M. Food intolerance and wheezing in young South Asian and white children: prevalence and clinical significance. The Journal of allergy and clinical immunology. 2006;118(2):528-30.

173. Holberg CJ, Wright AL, Martinez FD, Ray CG, Taussig LM, Lebowitz MD. Risk factors for respiratory syncytial virus-associated lower respiratory illnesses in the first year of life. Am J Epidemiol. 1991;133(11):1135-51.

174. Nikiema B, Spencer N, Seguin L. Poverty and chronic illness in early childhood: a comparison between the United kingdom and Quebec. Pediatrics. 2010;125(3):e499-507.

175. Dewey C, Fleming P, Golding J. Does the Supine Sleeping Position Have Any Adverse Effects on the Child? II. Development in the First 18 Months. Pediatrics. 1998;101(1):e5-e.

176. Medical Research Council. Southampton Women's Survey <u>http://www.mrc.soton.ac.uk/sws/</u>: Medical Research Coucnil; [cited 12/01/2017] [Available from: <u>http://www.mrc.soton.ac.uk/sws/</u>.

177. Quigley MA, Kelly YJ, Sacker A. Breastfeeding and hospitalization for diarrheal and respiratory infection in the United Kingdom Millennium Cohort Study. Pediatrics. 2007;119(4):e837-42.

178. Scholtens S, Wijga AH, Brunekreef B, Kerkhof M, Hoekstra MO, Gerritsen J, et al. Breast feeding, parental allergy and asthma in children followed for 8 years. The PIAMA birth cohort study. Thorax. 2009;64(7):604-9.

179. Neuman Å, Hohmann C, Orsini N, Pershagen G, Eller E, Kjaer HF, et al. Maternal Smoking in Pregnancy and Asthma in Preschool Children. American Journal of Respiratory and Critical Care Medicine. 2012;186(10):1037-43.

180. Local Government Association. Smoking Prevalence in adults - current smokers (IHS) in Liverpool <u>http://lginform.local.gov.uk/reports/lgastandard?mod-metric=3527&mod-period=5&mod-area=E08000012&mod-group=AllRegions_England&mod-type=namedComparisonGroup&modify-</u>

report=Apply: Local Government Association; 2017 [cited 2017 08/05/2017]. Available from: http://lginform.local.gov.uk/reports/lgastandard?mod-metric=3527&mod-period=5&mod-

<u>area=E08000012&mod-group=AllRegions</u> England&mod-type=namedComparisonGroup&modifyreport=Apply.

181. Dales RE, White J, Bhumgara C, McMullen E. Parental reporting of childrens' coughing is biased. Eur J Epidemiol. 1997;13(5):541-5.

182. Nicolaou NC, Simpson A, Lowe LA, Murray CS, Woodcock A, Custovic A. Day-care attendance, position in sibship, and early childhood wheezing: A population-based birth cohort study. Journal of Allergy and Clinical Immunology. 2008;122(3):500-6.e5.

183. Melki IS, Beydoun HA, Khogali M, Tamim H, Yunis KA. Household crowding index: a correlate of socioeconomic status and inter-pregnancy spacing in an urban setting. Journal of Epidemiology and Community Health. 2004;58(6):476-80.

184. Koopman LP, Smit HA, Heijnen M-LA, Wijga A, van Strien RT, Kerkhof M, et al. Respiratory Infections in Infants: Interaction of Parental Allergy, Child Care, and Siblings—The PIAMA Study. Pediatrics. 2001;108(4):943-8.

185. Fisk CM, Crozier SR, Inskip HM, Godfrey KM, Cooper C, Roberts GC, et al. Breastfeeding and reported morbidity during infancy: Findings from the Southampton Women's Survey. Maternal and Child Nutrition. 2011;7(1):61-70.

186. Wadsworth M. The origins and innovatory nature of the 1946 British national birth cohort study. Longitudinal and Life Course Studies. 2010;1(2):16.

187. Medical Research Council Unit of Lifelong Health and Aging at UCL. National Survey of Health and Development: NSHD <u>http://www.nshd.mrc.ac.uk/</u> [Available from: <u>http://www.nshd.mrc.ac.uk/</u>.

188. Wadsworth MEJ, Vinall LE, Jones AL, Hardy RJ, Whitehouse DB, Butterworth SL, et al. Alpha1antitrypsin as a risk for infant and adult respiratory outcomes in a national birth cohort. American Journal of Respiratory Cell and Molecular Biology. 2004;31(5):559-64.

189.CentreForLongitudinalStudies.1958NationalChildDevelopmentStudyhttp://www.cls.ioe.ac.uk/page.aspx?&sitesectionid=724&sitesectiontitle=Welcome+to+the+1958+National+Child+Development+Study[Availablefrom:

http://www.cls.ioe.ac.uk/page.aspx?&sitesectionid=724&sitesectiontitle=Welcome+to+the+1958+N ational+Child+Development+Study.

190. Butler NR, Goldstein H, Ross EM. Cigarette smoking in pregnancy: its influence on birth weight and perinatal mortality. Br Med J. 1972;2(5806):127-30.

191. Strachan DP, Butland BK, Anderson HR. Incidence and prognosis of asthma and wheezing illness from early childhood to age 33 in a national British cohort. British Medical Journal. 1996;312(7040):1195-9.

192. Johnston ID, Strachan DP, Anderson HR. Effect of pneumonia and whooping cough in childhood on adult lung function. The New England journal of medicine. 1998;338(9):581-7.

193. Centre Longitudinal Studies. The 1970 British Chohort Study For http://www.cls.ioe.ac.uk/page.aspx?&sitesectionid=795&sitesectiontitle=Welcome+to+the+1970+Br itish+Cohort+Study: Centre For Longitudinal Stdies UCL; [Available from: http://www.cls.ioe.ac.uk/page.aspx?&sitesectionid=795&sitesectiontitle=Welcome+to+the+1970+Br itish+Cohort+Study.

194. Taylor B, Wadsworth J. Maternal smoking during pregnancy and lower respiratory tract illness in early life. Archives of Disease in Childhood. 1987;62(8):786-91.

195. Neuspiel DR, Rush D, Butler NR, Golding J, Bijur PE, Kurzon M. Parental smoking and postinfancy wheezing in children: a prospective cohort study. Am J Public Health. 1989;79(2):168-71.

196. Park ES, Golding J, Carswell F, Stewart-Brown S. Preschool wheezing and prognosis at 10. Arch Dis Child. 1986;61(7):642-6.

197. Shaheen SO, Sterne JA, Montgomery SM, Azima H. Birth weight, body mass index and asthma in young adults. Thorax. 1999;54(5):396-402.

198. Taylor B, Wadsworth J, Golding J, Butler N. Breast feeding, eczema, asthma, and hayfever. Journal of Epidemiology and Community Health. 1983;37(2):95-9.

199. Centre For Longitudinal Studies. Millennium Cohort Study The http://www.cls.ioe.ac.uk/page.aspx?&sitesectionid=851&sitesectiontitle=Welcome+to+the+Millenni <u>um+Cohort+Study</u>: Studies; Centre For Longitudinal [Available from: http://www.cls.ioe.ac.uk/page.aspx?&sitesectionid=851&sitesectiontitle=Welcome+to+the+Millenni um+Cohort+Study.

200. Carson C, Sacker A, Kelly Y, Redshaw M, Kurinczuk JJ, Quigley MA. Asthma in children born after infertility treatment: findings from the UK Millennium Cohort Study. Human reproduction (Oxford, England). 2013;28(2):471-9.

201. Solmi F, Morris S. Association between childhood obesity and use of regular medications in the UK: longitudinal cohort study of children aged 5–11 years. BMJ Open. 2015;5(6).

202. Panico L, Stuart B, Bartley M, Kelly Y. Asthma Trajectories in Early Childhood: Identifying Modifiable Factors. PLOS ONE. 2014;9(11):e111922.

203. Shackleton N, Hale D, Viner RM. Trends and socioeconomic disparities in preadolescent's health in the UK: evidence from two birth cohorts 32 years apart. Journal of Epidemiology and Community Health. 2015.

204. Sciences TIoAH. Aberdeen Maternity and Neonatal Databank <u>http://www.abdn.ac.uk/</u> [Available from: <u>http://www.abdn.ac.uk/iahs/research/obsgynae/amnd/index.php</u>.

205. Aberdeen TUo. Children of the 1950s <u>http://www.abdn.ac.uk/</u> [Available from: <u>http://www.abdn.ac.uk/birth-cohorts/1950s/</u>.

AVON 206. Bristol Uo. Longitudinal Study of Parents and Children http://www.bristol.ac.uk/alspac/ [updated 1/12/2016. Available from: http://www.bristol.ac.uk/alspac/.

207. Golding J. Children of the nineties. A longitudinal study of pregnancy and childhood based on the population of Avon (ALSPAC). West of England medical journal. 1990;105(3):80-2.

208. Hunt L, Fleming P, Golding J. Does the supine sleeping position have any adverse effects on the child? I. Health in the first six months. The ALSPAC Study Team. Pediatrics. 1997;100(1):E11.

209. Fleming PJ, Gilbert R, Azaz Y, Berry PJ, Rudd PT, Stewart A, et al. Interaction between bedding and sleeping position in the sudden infant death syndrome: a population based case-control study. BMJ (Clinical research ed). 1990;301(6743):85-9.

210. Henderson J, Sherriff A, Farrow A, Ayres JG. Household chemicals, persistent wheezing and lung function: effect modification by atopy? European Respiratory Journal. 2008;31(3):547-54.

211. Shaheen SO, Newson RB, Sherriff A, Henderson AJ, Heron JE, Burney PG, et al. Paracetamol use in pregnancy and wheezing in early childhood. Thorax. 2002;57(11):958-63.

212. Cookson H, Granell R, Joinson C, Ben-Shlomo Y, Henderson AJ. Mothers' anxiety during pregnancy is associated with asthma in their children. The Journal of allergy and clinical immunology. 2009;123(4):847-53.e11.

213. Henderson J, Northstone K, Lee SP, Liao H, Zhao Y, Pembrey M, et al. The burden of disease associated with filaggrin mutations: a population-based, longitudinal birth cohort study. The Journal of allergy and clinical immunology. 2008;121(4):872-7.e9.

214. Small N. P1-164 An early report of the rationale, approach and establishment of Born in Bradford a bi-ethnic birth cohort study. Early Human Development.83:S129.

215. Bryant M, Santorelli G, Fairley L, West J, Lawlor DA, Bhopal R, et al. Design and characteristics of a new birth cohort, to study the early origins and ethnic variation of childhood obesity: the BiB1000 study. 2013. 2013;4(2):17.

216. Medicine TUoB-SoSaC. ELSPAC in the Isle of Man <u>http://www.bristol.ac.uk/</u> [Available from: <u>http://www.bristol.ac.uk/social-community-medicine/projects/isleofmanstudy/</u>.

217. Dragonas T, Golding J, Ignatyeva R, Prokhorskas R. Pregnancy in the 90s. European Longitudinal Study of Pregnancy and Childhood. Bristol: Redcliffe Press; 1996.

218. Group PaLER. The Newcastle Thousand Families Study <u>http://research.ncl.ac.uk</u> [Available from: <u>http://research.ncl.ac.uk/plerg/Research/1000F/1000_home.htm</u>.

219. Tennant PW, Gibson GJ, Pearce MS. Lifecourse predictors of adult respiratory function: results from the Newcastle Thousand Families Study. Thorax. 2008;63(9):823-30.

220. Pike KC, Inskip HM, Robinson SM, Cooper C, Godfrey KM, Roberts G, et al. The relationship between maternal adiposity and infant weight gain, and childhood wheeze and atopy. Thorax. 2013;68(4):372-9.

221. van der Gugten AC, Uiterwaal CSPM, van Putte-Katier N, Koopman M, Verheij TJM, van der Ent CK. Reduced neonatal lung function and wheezing illnesses during the first 5 years of life. European Respiratory Journal. 2013;42(1):107-15.

222. Organization TWH. EUROPEAN LONGITUDINAL STUDY OF PREGNANCY AND CHILDHOOD (ELSPAC). <u>http://www.euro.who.int/__data/assets/pdf_file/0004/119803/E68409.pdf;</u> 1999 13-18 September 1999.

223. University IoBaAotM. The European Longitudinal Study of Pregnancy and Childhood (ELSPAC) <u>http://www.elspac.cz/index-en.php?pg=about2017</u> [Available from: <u>http://www.elspac.cz/index-en.php?pg=about</u>.

224. Piler P, Kandrnal V, Kukla L, Andrýsková L, Švancara J, Jarkovský J, et al. Cohort Profile: The European Longitudinal Study of Pregnancy and Childhood (ELSPAC) in the Czech Republic. International journal of epidemiology. 2016.

 225.
 Centre For Longitudinal Studies U. European Longitudinal Study of Pregnancy and Childhood

 <u>http://www.cls.ioe.ac.uk/page.aspx?&sitesectionid=328&sitesectiontitle=European+Longitudinal+St</u>

 <u>udy+of+Pregnancy+and+Childhood</u>:
 nemisys;

 <u>http://www.cls.ioe.ac.uk/page.aspx?&sitesectionid=328&sitesectiontitle=European+Longitudinal+St</u>

 <u>http://www.cls.ioe.ac.uk/page.aspx?&sitesectionid=328&sitesectiontitle=European+Longitudinal+St</u>

 <u>udy+of+Pregnancy+and+Childhood</u>.

226. Custovic A, Simpson A, Woodcock A. Manchester cohort. Pediatric Pulmonology. 2004;37(S26):12-3.

227. Murray CS, Pipis SD, McArdle EC, Lowe LA, Custovic A, Woodcock A. Lung function at one month of age as a risk factor for infant respiratory symptoms in a high risk population. Thorax. 2002;57(5):388-92.

228. Lowe LA, Simpson A, Woodcock A, Morris J, Murray CS, Custovic A. Wheeze phenotypes and lung function in preschool children. Am J Respir Crit Care Med. 2005;171(3):231-7.

229. Simpson A, Custovic A. Allergen avoidance in the primary prevention of asthma. Current Opinion in Allergy and Clinical Immunology. 2004;4(1):45-51.

230. Simpson BM, Custovic A, Simpson A, Hallam CL, Walsh D, Marolia H, et al. NAC Manchester Asthma and Allergy Study (NACMAAS): risk factors for asthma and allergic disorders in adults. Clinical and experimental allergy : journal of the British Society for Allergy and Clinical Immunology. 2001;31(3):391-9.

231. Thomas M, Custovic A, Woodcock A, Morris J, Simpson A, Murray CS. Atopic wheezing and early life antibiotic exposure: a nested case-control study. Pediatric allergy and immunology : official publication of the European Society of Pediatric Allergy and Immunology. 2006;17(3):184-8.

232. Mölter A, Agius RM, de Vocht F, Lindley S, Gerrard W, Lowe L, et al. Long-term exposure to PM10 and NO2 in association with lung volume and airway resistance in the MAAS birth cohort. Environmental Health Perspectives. 2013;121(10):1232-8.

233. Brunekreef B, Smit J, de Jongste J, Neijens H, Gerritsen J, Postma D, et al. The prevention and incidence of asthma and mite allergy (PIAMA) birth cohort study: design and first results. Pediatric allergy and immunology : official publication of the European Society of Pediatric Allergy and Immunology. 2002;13 Suppl 15:55-60.

234. Team TPR. The PIAMA Research Project <u>http://piama.iras.uu.nl/index-en.php2016</u> [updated 16/12/2016. Available from: <u>http://piama.iras.uu.nl/index-en.php</u>.

235. Gehring U, de Jongste JC, Kerkhof M, Oldewening M, Postma D, van Strien RT, et al. The 8year follow-up of the PIAMA intervention study assessing the effect of mite-impermeable mattress covers. Allergy. 2012;67(2):248-56.

236. Willers SM, Wijga AH, Brunekreef B, Kerkhof M, Gerritsen J, Hoekstra MO, et al. Maternal food consumption during pregnancy and the longitudinal development of childhood asthma. Am J Respir Crit Care Med. 2008;178(2):124-31.

237. Roduit C, Scholtens S, de Jongste JC, Wijga AH, Gerritsen J, Postma DS, et al. Asthma at 8 years of age in children born by caesarean section. Thorax. 2009;64(2):107-13.

238. Caudri D, Wijga A, Gehring U, Smit HA, Brunekreef B, Kerkhof M, et al. Respiratory symptoms in the first 7 years of life and birth weight at term: the PIAMA Birth Cohort. Am J Respir Crit Care Med. 2007;175(10):1078-85.

239. Douwes J, van Strien R, Doekes G, Smit J, Kerkhof M, Gerritsen J, et al. Does early indoor microbial exposure reduce the risk of asthma? The Prevention and Incidence of Asthma and Mite Allergy birth cohort study. The Journal of allergy and clinical immunology. 2006;117(5):1067-73.

240. Wijga AH, Smit HA, Kerkhof M, de Jongste JC, Gerritsen J, Neijens HJ, et al. Association of consumption of products containing milk fat with reduced asthma risk in pre-school children: the PIAMA birth cohort study. Thorax. 2003;58(7):567-72.

241. Brauer M, Hoek G, Van Vliet P, Meliefste K, Fischer PH, Wijga A, et al. Air pollution from traffic and the development of respiratory infections and asthmatic and allergic symptoms in children. Am J Respir Crit Care Med. 2002;166(8):1092-8.

242. Gehring U, Wijga AH, Brauer M, Fischer P, de Jongste JC, Kerkhof M, et al. Traffic-related air pollution and the development of asthma and allergies during the first 8 years of life. Am J Respir Crit Care Med. 2010;181(6):596-603.

243. Boudewijn IM, Savenije OE, Koppelman GH, Wijga AH, Smit HA, de Jongste JC, et al. Nocturnal dry cough in the first 7 years of life is associated with asthma at school age. Pediatr Pulmonol. 2015;50(9):848-55.

244. Taussig LM, Wright AL, Morgan WJ, Harrison HR, Ray CG. The Tucson Children's Respiratory Study. I. Design and implementation of a prospective study of acute and chronic respiratory illness in children. Am J Epidemiol. 1989;129(6):1219-31.

245. Ray CG, Minnich LL, Holberg CJ, Shehab ZM, Wright AL, Barton LL, et al. Respiratory syncytial virus-associated lower respiratory illnesses: Possible influence of other agents. Pediatric Infectious Disease Journal. 1993;12(1):15-9.

246. Wright AL, Holberg C, Martinez FD, Taussig LM. Relationship of parental smoking to wheezing and nonwheezing lower respiratory tract illnesses in infancy. Group Health Medical Associates. J Pediatr. 1991;118(2):207-14.

247. Martinez FD, Morgan WJ, Wright AL, Holberg CJ, Taussig LM, Personnel* TGHMA. Diminished Lung Function as a Predisposing Factor for Wheezing Respiratory Illness in Infants. New England Journal of Medicine. 1988;319(17):1112-7.

248. Survey ECRH. European Community Respiratory Health Survey <u>http://www.ecrhs.org/2014</u> [updated 27/03/14. Available from: <u>http://www.ecrhs.org/Default.htm</u>.

249. Burney PGJ, Luczynska C, Chinn S, Jarvis D. The European Community Respiratory Health Survey. European Respiratory Journal. 1994;7(5):954-60.

250. Jarvis D. The European Community Respiratory Health Survey II. European Respiratory Journal. 2002;20(5):1071-9.

251. Svanes C. What has the ECRHS told us about the childhood risks of asthma, allergy and lung function? The Clinical Respiratory Journal. 2008;2:34-44.

252. Verlato G, Calabrese R, De Marco R. Correlation between asthma and climate in the European Community Respiratory Health Survey. Archives of environmental health. 2002;57(1):48-52.

253. Chinn S, Jarvis D, Burney P, Luczynska C, Ackermann-Liebrich U, Antó JM, et al. Increase in diagnosed asthma but not in symptoms in the European Community Respiratory Health Survey. Thorax. 2004;59(8):646-51.

254. Janson C, Anto J, Burney P, Chinn S, de Marco R, Heinrich J, et al. The European Community Respiratory Health Survey: what are the main results so far? European Respiratory Journal. 2001;18(3):598-611.

255. Zock J-P, Jarvis D, Luczynska C, Sunyer J, Burney P. Housing characteristics, reported mold exposure, and asthma in the European Community Respiratory Health Survey. Journal of Allergy and Clinical Immunology. 2002;110(2):285-92.

256. Janson C, Chinn S, Jarvis D, Zock J-P, Torén K, Burney P. Effect of passive smoking on respiratory symptoms, bronchial responsiveness, lung function, and total serum IgE in the European Community Respiratory Health Survey: a cross-sectional study. The Lancet.358(9299):2103-9.

257. Svanes C, Jarvis D, Chinn S, Burney P. Childhood environment and adult atopy: Results from the European Community Respiratory Health Survey. Journal of Allergy and Clinical Immunology. 1999;103(3):415-20.

258. Voll-Aanerud M, Eagan TM, Plana E, Omenaas ER, Bakke PS, Svanes C, et al. Respiratory symptoms in adults are related to impaired quality of life, regardless of asthma and COPD: results from the European community respiratory health survey. Health and Quality of Life Outcomes. 2010;8(1):107.

259. Weiland SK, Björkstén B, Brunekreef B, Cookson WOC, von Mutius E, Strachan DP. Phase II of the International Study of Asthma and Allergies in Childhood (ISAAC II): rationale and methods. European Respiratory Journal. 2004;24(3):406-12.

260. Pearce N, Aït-Khaled N, Beasley R, Mallol J, Keil U, Mitchell E, et al. Worldwide trends in the prevalence of asthma symptoms: phase III of the International Study of Asthma and Allergies in Childhood (ISAAC). Thorax. 2007;62(9):758-66.

261. Strachan D, Sibbald B, Weiland S, Ait-Khaled N, Anabwani G, Anderson HR, et al. Worldwide variations in prevalence of symptoms of allergic rhinoconjunctivitis in children: the International Study of Asthma and Allergies in Childhood (ISAAC). Pediatric Allergy and Immunology. 1997;8(4):161-8.

262. Nagel G, Weinmayr G, Flohr C, Kleiner A, Strachan DP. Association of pertussis and measles infections and immunizations with asthma and allergic sensitization in ISAAC Phase Two. Pediatric allergy and immunology : official publication of the European Society of Pediatric Allergy and Immunology. 2012;23(8):737-46.

263. Weinmayr G, Gehring U, Genuneit J, Buchele G, Kleiner A, Siebers R, et al. Dampness and moulds in relation to respiratory and allergic symptoms in children: results from Phase Two of the International Study of Asthma and Allergies in Childhood (ISAAC Phase Two). Clinical and experimental allergy : journal of the British Society for Allergy and Clinical Immunology. 2013;43(7):762-74.

264. Weinmayr G, Forastiere F, Buchele G, Jaensch A, Strachan DP, Nagel G. Overweight/obesity and respiratory and allergic disease in children: international study of asthma and allergies in childhood (ISAAC) phase two. PLoS One. 2014;9(12):e113996.

265. Nagel G, Buchele G, Weinmayr G, Bjorksten B, Chen YZ, Wang H, et al. Effect of breastfeeding on asthma, lung function and bronchial hyperreactivity in ISAAC Phase II. The European respiratory journal. 2009;33(5):993-1002.

266. Bjorksten B, Clayton T, Ellwood P, Stewart A, Strachan D. Worldwide time trends for symptoms of rhinitis and conjunctivitis: Phase III of the International Study of Asthma and Allergies in Childhood. Pediatric allergy and immunology : official publication of the European Society of Pediatric Allergy and Immunology. 2008;19(2):110-24.

267. Ferrer M, Alonso J, Prieto L, Plaza V, Monso E, Marrades R, et al. Validity and reliability of the St George's Respiratory Questionnaire after adaptation to a different language and culture: the Spanish example. European Respiratory Journal. 1996;9(6):1160.

268. Kuehni CE, Davis A, Brooke AM, Silverman M. Are all wheezing disorders in very young (preschool) children increasing in prevalence? The Lancet.357(9271):1821-5.

269. Kuehni CE, Strippoli MP, Chauliac ES, Silverman M. Snoring in preschool children: prevalence, severity and risk factors. The European respiratory journal. 2008;31(2):326-33.

270. Brooke AM, Lambert PC, Burton PR, Clarke C, Luyt DK, Simpson H. The natural history of respiratory symptoms in preschool children. Am J Respir Crit Care Med. 1995;152(6 Pt 1):1872-8.

271. Strippoli MP, Spycher BD, Pescatore AM, Beardsmore CS, Silverman M, Kuehni CE. Exclusive viral wheeze and allergic wheeze: evidence for discrete phenotypes. The European respiratory journal. 2011;38(2):472-4.

272. Dogaru CM, Strippoli MP, Spycher BD, Frey U, Beardsmore CS, Silverman M, et al. Breastfeeding and lung function at school age: does maternal asthma modify the effect? Am J Respir Crit Care Med. 2012;185(8):874-80.

273. Pescatore AM, Dogaru CM, Duembgen L, Silverman M, Gaillard EA, Spycher BD, et al. A simple asthma prediction tool for preschool children with wheeze or cough. The Journal of allergy and clinical immunology. 2014;133(1):111-8.e1-13.

274. Frank P, Ferry S, Moorhead T, Hannaford P. Use of a postal questionnaire to estimate the likely under-diagnosis of asthma-like illness in adults. British Journal of General Practice. 1996;46(406):295-7.

275. Frank TL, Frank PI, Cropper JA, Hazell ML, Hannaford PC, McNamee RR, et al. Identification of adults with symptoms suggestive of obstructive airways disease: Validation of a postal respiratory questionnaire. BMC Family Practice. 2003;4:5-.

276. Hirsch S, Shapiro JL, Turega MA, Frank TL, Niven RM, Frank PI. Using a Neural Network to Screen a Population for Asthma. Annals of Epidemiology. 2001;11(6):369-76.

277. Frank T, Frank P, McNamee R, Wright T, Hannaford P, Morrison J, et al. Assessment of a simple scoring system applied to a screening questionnaire of asthma in children aged 5-15 yrs. European Respiratory Journal. 1999;14(5):1190-7.

278. Frank PI, Wicks PD, Hazell ML, Linehan MF, Hirsch S, Hannaford PC, et al. Temporal change in the prevalence of respiratory symptoms and obstructive airways disease 1993-2001. The British journal of general practice : the journal of the Royal College of General Practitioners. 2005;55(517):596-602.

279. (IKE) Ifkoem. ABIS (All Babies in Southeast Sweden) <u>https://www.abis-studien.se/hem/english-11100423</u>: Institutionen för klinisk och experimentell medicin (IKE); [Available from: <u>https://www.abis-studien.se/hem/english-11100423</u>.

280. Jaddoe VW, van Duijn CM, Franco OH, van der Heijden AJ, van lizendoorn MH, de Jongste JC, et al. The Generation R Study: design and cohort update 2012. European journal of epidemiology. 2012;27(9):739-56.

281. Lau S, Nickel R, Niggemann B, Grüber C, Sommerfeld C, Illi S, et al. The development of childhood asthma: lessons from the German Multicentre Allergy Study (MAS). Paediatric Respiratory Reviews. 2002;3(3):265-72.

282. Edinburgh Uo. Growing Up in Scotland <u>http://growingupinscotland.org.uk/</u>: Univeristy of Edinburgh; [updated 05/01/2017; cited 2016 09/201/2016]. Available from: <u>http://growingupinscotland.org.uk/</u>.

283. van Eijsden M, Vrijkotte TG, Gemke RJ, van der Wal MF. Cohort Profile: The Amsterdam Born Children and their Development (ABCD) Study. International Journal of Epidemiology. 2011;40(5):1176-86.

284. Fuchs O, Latzin P, Kuehni CE, Frey U. Cohort Profile: The Bern Infant Lung Development Cohort. International Journal of Epidemiology. 2012;41(2):366-76.

285.InstitutSS.DanishNationalBirthCohorthttp://www.ssi.dk/English/RandD/Research%20areas/Epidemiology/DNBC.aspx[updated11/12/2015.Availablefrom:

http://www.ssi.dk/English/RandD/Research%20areas/Epidemiology/DNBC.aspx.

286. Unit DMHaDR. Dunedin Multidisciplinary Health and Development Research Unit <u>http://dunedinstudy.otago.ac.nz/</u>: Univerisy of Otago; [updated 13/12/2016. Available from: <u>http://dunedinstudy.otago.ac.nz/</u>.

287. Welfare NIoHa. Helsinki Birth Cohort Study (HBCS) - Idefix <u>https://www.thl.fi/fi/web/thlfi-en/research-and-expertwork/projects-and-programmes/projects/23572</u>: National Institute ofr Health and Welfare; 2014 [updated 23/11/2014. Available from: <u>https://www.thl.fi/fi/web/thlfi-en/research-and-expertwork/projects-and-programmes/projects/23572</u>.

288. Schooling CM, Hui LL, Ho LM, Lam T-H, Leung GM. Cohort Profile: 'Children of 1997': a Hong Kong Chinese birth cohort. International Journal of Epidemiology. 2011.

289.INMAIyMA.Generaldescriptionhttp://www.proyectoinma.org/presentacion-inma/eninma/en_index.html[updated23/12/2016.Availablefrom:http://www.proyectoinma.org/presentacion-inma/enindex.html.

290. Jedrychowski W, Galas AS, Flak E, Jacek R, Penar A, Spengler J, et al. Increased burden of respiratory disease in the first six months of life due to prenatal environmental tobacco smoke: Krakow birth cohort study. Early Child Development and Care. 2007;177(4):369-81.

291. Najman J, Bor W, O'Callaghan M, Williams G, Aird R, Shuttlewood G. Cohort Profile: The Mater-University of Queensland Study of Pregnancy (MUSP). International Journal of Epidemiology. 2005;34(5):992-7.

292. Kim B-M, Ha M, Park H-S, Lee BE, Kim YJ, Hong Y-C, et al. The Mothers and Children's Environmental Health (MOCEH) study. European journal of epidemiology. 2009;24(9):573-83.

293. Study TNC. THE NATIONAL CHILDREN'S STUDY ARCHIVE- Study Description and Guide. https://www.nichd.nih.gov/research/NCS/Documents/NCS Archive Study Description.pdf; 2016 May 2016.

294.LondonIC.NorthernFinlandBirthCohorts(NFBC)https://www1.imperial.ac.uk/publichealth/departments/ebs/projects/cdel/euroblcs/cohorts/nfbc/:ImperialCollegeLondon;2013[Availablefrom:

https://www1.imperial.ac.uk/publichealth/departments/ebs/projects/cdel/euroblcs/cohorts/nfbc/. 295. Health NIoP. What is the Norwegian Mother and Child Cohort Study? https://www.fhi.no/en/studies/moba/what-is-the-norwegian-mother-and-child-cohort-study:

Norwegian Institute of Public Health; 2016 [updated 27.05.2016 Available from: <u>https://www.fhi.no/en/studies/moba/what-is-the-norwegian-mother-and-child-cohort-study/</u>.

296. Horta BL, Gigante DP, Gonçalves H, dos Santos Motta J, Loret de Mola C, Oliveira IO, et al. Cohort Profile Update: The 1982 Pelotas (Brazil) Birth Cohort Study. International Journal of Epidemiology. 2015;44(2):441-e.

297. Bornehag C-G, Moniruzzaman S, Larsson M, Lindström CB, Hasselgren M, Bodin A, et al. The SELMA Study: A Birth Cohort Study in Sweden Following More Than 2000 Mother–Child Pairs. Paediatric and Perinatal Epidemiology. 2012;26(5):456-67.

298. Lagstrom H, Rautava P, Kaljonen A, Raiha H, Pihlaja P, Korpilahti P, et al. Cohort profile: Steps to the healthy development and well-being of children (the STEPS Study). Int J Epidemiol. 2013;42(5):1273-84.

299. Stenberg S-Å, Vågerö D. Cohort Profile: The Stockholm birth cohort of 1953. International Journal of Epidemiology. 2006;35(3):546-8.

300.BernUo.SwissNationalCohort-Introductionhttp://www.swissnationalcohort.ch/index.php?id=2985:Unviersity of Bern; 2014[Available from:http://www.swissnationalcohort.ch/index.php?id=2985.

301. Sekine M, Yamagami T, Handa K, Saito T, Nanri S, Kawaminami K, et al. A dose–response relationship between short sleeping hours and childhood obesity: results of the Toyama Birth Cohort Study. Child: Care, Health and Development. 2002;28(2):163-70.

302. Patel AS, Siegert RJ, Brignall K, Gordon P, Steer S, Desai SR, et al. The development and validation of the King's Brief Interstitial Lung Disease (K-BILD) health status questionnaire. Thorax. 2012;67(9):804-10.

303. Patel AS, Siegert RJ, Keir GJ, Bajwah S, Barker RD, Maher TM, et al. The minimal important difference of the King's Brief Interstitial Lung Disease Questionnaire (K-BILD) and forced vital capacity in interstitial lung disease. Respiratory Medicine. 2013;107(9):1438-43.

304. Jones PW, Quirk FH, Baveystock CM, Littlejohns P. A Self-complete Measure of Health Status for Chronic Airflow Limitation: The St. George's Respiratory Questionnaire. American Review of Respiratory Disease. 1992;145(6):1321-7.

305. Sanjuás C, Alonso J, Prieto L, Ferrer M, Broquetas JM, Antó JM. Health-related quality of life in asthma: A comparison between the St George's respiratory questionnaire and the asthma quality of life questionnaire. Quality of Life Research. 2002;11(8):729-38.

306. WILSON CB, JONES PW, O'LEARY CJ, COLE PJ, WILSON R. Validation of the St. George's Respiratory Questionnaire in Bronchiectasis. American Journal of Respiratory and Critical Care Medicine. 1997;156(2):536-41.

307. Meguro M, Barley EA, Spencer S, Jones PW. DEvelopment and validation of an improved, copd-specific version of the st. george respiratory questionnaire*. Chest. 2007;132(2):456-63.

308. Yorke J, Jones PW, Swigris JJ. Development and validity testing of an IPF-specific version of the St George's Respiratory Questionnaire. Thorax. 2010;65(10):921-6.

309. Nathan RA, Sorkness CA, Kosinski M, Schatz M, Li JT, Marcus P, et al. Development of the asthma control test: A survey for assessing asthma control. Journal of Allergy and Clinical Immunology. 2004;113(1):59-65.

310. Thomas M, Kay S, Pike J, Williams A, Rosenzweig JRC, Hillyer EV, et al. The Asthma Control Test (ACT) as a predictor of GINA guideline-defined asthma control: analysis of a multinational cross-sectional survey. Prim Care Respir J. 2009;18(1):41-9.

311. Schatz M, Sorkness CA, Li JT, Marcus P, Murray JJ, Nathan RA, et al. Asthma Control Test: Reliability, validity, and responsiveness in patients not previously followed by asthma specialists. Journal of Allergy and Clinical Immunology. 2006;117(3):549-56.

312. Wijkstra PJ, TenVergert EM, Van Altena R, Otten V, Postma DS, Kraan J, et al. Reliability and validity of the chronic respiratory questionnaire (CRQ). Thorax. 1994;49(5):465-7.

313. van der Molen T, Willemse BW, Schokker S, ten Hacken NH, Postma DS, Juniper EF. Development, validity and responsiveness of the Clinical COPD Questionnaire. Health and Quality of Life Outcomes. 2003;1(1):13.

314. Jones PW, Harding G, Berry P, Wiklund I, Chen W-H, Kline Leidy N. Development and first validation of the COPD Assessment Test. European Respiratory Journal. 2009;34(3):648-54.

315. Jones PW, Brusselle G, Dal Negro RW, Ferrer M, Kardos P, Levy ML, et al. Properties of the COPD assessment test in a cross-sectional European study. European Respiratory Journal. 2011;38(1):29-35.

316. Dodd JW, Hogg L, Nolan J, Jefford H, Grant A, Lord VM, et al. The COPD assessment test (CAT): response to pulmonary rehabilitation. A multicentre, prospective study. Thorax. 2011;66(5):425-9.

317. Jones PW, Harding G, Wiklund I, Berry P, Tabberer M, Yu R, et al. TEsts of the responsiveness of the copd assessment test following acute exacerbation and pulmonary rehabilitation. Chest. 2012;142(1):134-40.

318. Burney P, Chinn S. Developing a New Questionnaire for Measuring the Prevalence and Distribution of Asthma. Chest. 1987;91(6, Supplement):79S-83S.

319. Burney PG, Laitinen LA, Perdrizet S, Huckauf H, Tattersfield AE, Chinn S, et al. Validity and repeatability of the IUATLD (1984) Bronchial Symptoms Questionnaire: an international comparison. The European respiratory journal. 1989;2(10):940-5.

320. Birring SS, Prudon B, Carr AJ, Singh SJ, Morgan MDL, Pavord ID. Development of a symptom specific health status measure for patients with chronic cough: Leicester Cough Questionnaire (LCQ). Thorax. 2003;58(4):339-43.

321. Yousaf N, Lee KK, Jayaraman B, Pavord ID, Birring SS. The assessment of quality of life in acute cough with the Leicester Cough Questionnaire (LCQ-acute). Cough. 2011;7(1):4.

Murray MP, Turnbull K, MacQuarrie S, Pentland JL, Hill AT. Validation of the Leicester Cough Questionnaire in non-cystic fibrosis bronchiectasis. European Respiratory Journal. 2009;34(1):125-31.
Berkhof FF, Boom LN, ten Hertog NE, Uil SM, Kerstjens HA, van den Berg JW. The validity and precision of the leicester cough questionnaire in COPD patients with chronic cough. Health and Quality of Life Outcomes. 2012;10(1):4.

324. Stenton C. The MRC breathlessness scale. Occupational Medicine. 2008;58(3):226-7.

325. Bestall JC, Paul EA, Garrod R, Garnham R, Jones PW, Wedzicha JA. Usefulness of the Medical Research Council (MRC) dyspnoea scale as a measure of disability in patients with chronic obstructive pulmonary disease. Thorax. 1999;54(7):581-6.

326. Maille AR, Koning CJ, Zwinderman AH, Willems LN, Dijkman JH, Kaptein AA. The development of the 'Quality-of-life for Respiratory Illness Questionnaire (QOL-RIQ)': a disease-specific quality-of-life questionnaire for patients with mild to moderate chronic non-specific lung disease. Respir Med. 1997;91(5):297-309.

327. van Stel HF, Maillé AR, Colland VT, Everaerd W. Interpretation of change and longitudinal validity of the quality of life for respiratory illness questionnaire (QoLRIQ) in inpatient pulmonary rehabilitation. Quality of Life Research. 2003;12(2):133-45.

328. Jackson GG, Dowling HF, Anderson TO, Riff L, Saporta J, Turck M. Susceptibility and immunity to common upper respiratory viral infections--the common cold. Ann Intern Med. 1960;53:719-38.

329. Barrett B, Locken K, Maberry R, Schwamman J, Brown R, Bobula J, et al. The Wisconsin Upper Respiratory Symptom Survey (WURSS): a new research instrument for assessing the common cold. The Journal of family practice. 2002;51(3):265.

330. Barrett B, Brown R, Mundt M, Safdar N, Dye L, Maberry R, et al. The Wisconsin Upper Respiratory Symptom Survey is responsive, reliable, and valid. Journal of Clinical Epidemiology. 2005;58(6):609-17.

331. Barrett B, Brown R, Voland R, Maberry R, Turner R. Relations among questionnaire and laboratory measures of rhinovirus infection. European Respiratory Journal. 2006;28(2):358-61.